Sustainability and resilience of European farming systems

An integrated assessment



Propositions

- 1. European farming systems operate closely to at least one critical threshold. (this thesis)
- 2. A focus on the economic domain undermines farming system sustainability and resilience. (this thesis)
- 3. Transparent and negotiable trade-offs between quality and workload yield healthy doctors.
- 4. Financialization of agriculture is a risk for global food security.
- Providers of public goods, such as farms, need to "borrow" from society for their resilience.
- 6. Strong governments are necessary for thriving societies.

Propositions belonging to the thesis, entitled

Sustainability and resilience of European farming systems: an integrated assessment

Wim Paas, Wageningen, 24 October 2022

Sustainability and resilience of European farming systems: an integrated assessment

Wim Paas

Thesis committee

Promotors

Prof. Dr M.K. van Ittersum Personal Chair at the Plant Production Systems Group Wageningen University & Research

Prof. Dr M.P.M. Meuwissen Personal Chair at the Business Economics Group Wageningen University & Research

Co-promotor

Dr P. Reidsma Associate professor at the Plant Production Systems Group Wageningen University & Research

Other members

Prof. Dr M. Bakker, Wageningen University & Research Dr J.C.J. Groot, Wageningen University & Research Prof. Dr K Helming, Leibniz Centre for Agricultural Landscape Research (ZALF), Germany Dr O. Therond, Université de Lorraine, France

This research was conducted under the auspices of the Graduate School of Production Ecology & Resource Conservation (PE&RC).

Sustainability and resilience of European farming systems:

an integrated assessment

Wim Paas

Thesis

submitted in fulfilment of the requirements for the degree of doctor at Wageningen University by the authority of the Rector Magnificus, Prof. Dr A.P.J. Mol, in the presence of the Thesis Committee appointed by the Academic Board to be defended in public on Monday 24 October 2022 at 11 a.m. in the Omnia Auditorium of Wageningen University.

Wim Paas

Sustainability and resilience of European farming systems: an integrated assessment, 164 pages.

PhD thesis, Wageningen University, Wageningen, the Netherlands (2022) With references, with summaries in English and Dutch

ISBN: 978-94-6447-347-6 DOI: https://doi.org/10.18174/574656

Abstract

An increasing variety of stresses and shocks provides challenges for European farming systems. As a consequence, the sustainability and resilience of Europe's diverse farming systems is at stake. In particular the possible presence of economic, social or environmental thresholds in farming systems is worrying, as beyond those thresholds permanent and undesired system change may happen.

The aim of this thesis is to operationalize a resilience framework and to assess the sustainability and resilience of current and future European farming systems. Sustainability of a system is in this thesis defined as an adequate performance of all system functions across the environmental, economic and social domains. Sustainability of agricultural systems has been studied extensively, but existing frameworks and tools are not designed to study resilience which is much more about the different capacities of systems to deal with disturbances, i.e. robustness, adaptability and transformability.

The following research questions are central in this thesis: 1) Is there a balance between social, economic and environmental functions in European farming systems in terms of importance and performance? 2) Are European farming systems approaching critical thresholds? 3) What resilience capacities do and should European farming systems have? 4) What strategies enhance sustainability and resilience of European farming systems?

Based on the application of new and (semi-)quantitative methods developed in this thesis, the following conclusions on the sustainability and resilience of European farming systems – and the methods to assess these - can be drawn:

European farming systems are perceived to have low to moderate sustainability and resilience, and operate close to critical thresholds. In the studied farming systems there is an overemphasis on (short-term) economic viability and a lack of attention for (long-term) social variables, while robustness was perceived to prevail over adaptability and transformability. According to stakeholders, main building blocks for current resilience in most case studies were the resilience attributes related to having production coupled with local and natural resources, heterogeneity of farm types, social self-organization, reasonable profitability, and infrastructure for innovation. The latter two were perceived as particularly important for transformability. Past strategies of farming systems were often geared towards making the system more profitable, and to a lesser extent towards the other important building blocks for current resilience. For improving sustainability and resilience, future farming systems need a more balanced attention for economic, social and environmental domains, and an enabling institutional and socio-economic environment. In terms of strategies, technological innovation is often required, provided it is implemented simultaneously with social, agro-ecological and institutional strategies that consider the long-term. To implement such strategies, all involved actors inside and outside the farming system need to collaborate.

Sustainability and resilience of farming systems remains a challenging subject due to its complexity in terms of detail (different domains, many concepts and variables) and dynamics (non-linearity, thresholds, interactions). The research presented in this thesis confirmed the usefulness of the resilience framework in reducing this complexity through a step-wise approach tailored to farming systems. The participatory approaches presented in this thesis contributed mainly to describing and explaining sustainability and resilience of current farming systems. These methods provide, therefore, a good basis for exploring future farming systems. The quantitative approach (presented in Chapter 4) confirmed the impact of weather extremes on economic and environmental farm performance, but was limited in explaining resilience, and raised awareness about the influence researchers have on the results through the selection of response variables. Based on the work and reflections presented in this thesis I see scope for better understanding and assessing farming system sustainability and resilience through system thinking theory and the use of participatory integrated assessments.

Contents

Chapter 1. General introduction1
Chapter 2. Participatory assessment of sustainability and resilience of three specialized farming systems
Chapter 3. How do Stakeholders Perceive the Sustainability and Resilience of European Farming Systems?
Chapter 4. Temporal and inter-farm variability of economic and environmental farm performance: a resilience perspective on potato producing regions in the Netherlands?
Chapter 5. Assessing future sustainability and resilience of farming systems with a participatory method: a case study on extensive sheep farming in Huesca, Spain73
Chapter 6. Participatory assessment of critical thresholds for resilient and sustainable European farming systems
Chapter 7. General discussion 117
References
Summary
Samenvatting
Acknowledgements
About the author
Publications
PE&RC Training and Education Statement

General introduction

1.1 Contemporary agriculture in a global context

In its attempt to produce food and improve the quality of life for an increasing population, humanity is approaching and exceeding the limits of the earth's carrying capacity, i.e. exceeding planetary boundaries (Rockström et al., 2009b, 2009a; Steffen et al., 2015). Beyond those planetary boundaries, there is a high risk of non-linear and drastic changes that have a lasting impact on life on earth. In the context of an increasing demand for food, feed, fibre and fuel, agriculture is a major contributor to the exceedance of two planetary boundaries: biosphere integrity, in particular regarding the loss of genetic diversity, and biogeochemical flows regarding phosphorus and nitrogen (Campbell et al., 2017). For two other planetary boundaries, land-system change and freshwater use, agriculture alone brings humanity in a danger zone, i.e. in a zone with increasing risk of exceeding the earth's limits (Campbell et al., 2017). Without interventions, agricultural activity is expected to contribute to the exceedance of the planetary boundaries for climate change and land-use change in 2050 (Springmann et al., 2018).

The global agricultural system is characterized by specialization, intensification, land expansion and a strong connection to global markets that drive these developments (Giller et al., 2021; Nyström et al., 2019). Globally, expanding and intensive agriculture is a major driver of environmental degradation (Tilman et al., 2002) and collapse of ecosystems (Rocha et al., 2015). Expansion directly contributes to further approaching the land-system change boundary, i.e. nature area is replaced by farmland area. Intensification coerces agricultural systems to high food production levels (Rist et al., 2014), but simultaneously decouples them to a certain extent from the underlying ecosystem and its natural regulating and supporting processes (Nyström et al., 2019; Rist et al., 2014; Therond et al., 2017). This reduces the capability of actors in agricultural systems to notice feedback signals and develop abilities to adapt to changing conditions (Nyström et al., 2019). Continued intensification and expansion is expected in particular in the developing world (Giller et al., 2021; Koning and van Ittersum, 2009), implying, with current production practices, a greater pressure on the environment, thus reducing the ecological base. This reduces the resilience of agricultural systems (Cabell and Oelofse, 2012). At the same time, the main resources that currently make agricultural intensification possible, i.e. fossil fuel and rock phosphate, are finite.

Although a wide variety of farms exists in terms of area, degree of technology use and management structure (family farm/corporate structure), almost all farms are susceptible to low and declining market prices determined by global markets (Giller et al., 2021; Koning and van Ittersum, 2009). The globalized setting for agriculture further reduces the feedback signals, or signals are simply ignored as production and consumption are spatially distanced (Nyström et al., 2019; Sundkvist et al., 2005). One quarter of all production is traded internationally (Nyström et al., 2019). Producing more locally could improve feedback signals (Ericksen, 2008; Sundkvist et al., 2005) and thus improve resilience (Biggs et al., 2012; Cabell and Oelofse, 2012). However, a large proportion of the human population is dependent on importing food (Fader et al., 2013; Kinnunen et al., 2020). In the, by now, seemingly necessary, global markets, overexploitation of natural resources can go unnoticed and support regions to displace pressure on the environment to somewhere else in the world (Nyström et al., 2019).

While humanity is approaching global environmental limits, the most important feedback signals that are received by agricultural systems are the increased occurrence of mostly climate-change induced shocks in food production (Cottrell et al., 2019). Towards the future, the risk of co-occurring yield failures in bread basket regions is expected to increase, which will affect global food security (Gaupp et al., 2019). An important feedback signal from society to agriculture and the broader food system (including production and consumption) is the large proportion of the global population that is either obese or undernourished. This signal suggests that the current food system is also approaching or exceeding socio-economic limits related to human health and affordability of food. An important feedback signal in the economic domain is the low income of farmers around the world. Small-holder farmers, in particular in developing countries, often have an income below the poverty line (van de Ven et al., 2021; van Vliet et al., 2021). In western countries, farmers have (partly subsidized) incomes that are substantially lower than the national average (Giller et al., 2021; van Vliet et al., 2015). The social and economic feedback signals from agriculture to

society are important and therefore need to be considered in local and global projects or frameworks for improved sustainability, i.e. the Sustainable Development Goals (sdgs.un.org).

1.2 European agriculture

1.2.1 Structure and diversity

In this thesis the focus is on European agriculture. The structure of the agricultural system in Europe resembles largely the global agricultural system with regard to intensification, specialization and being connected to global markets (Giller et al., 2021). In Europe, the bulk of the food is produced by medium to highly intensive, specialized farms (DG-Agri, 2017), which are often run by families (European Commission, 2018; Eurostat, 2018) and, to a lesser, but increasing extent, by large-scale agricultural corporations (Giller et al., 2021). Medium to highly intensive, specialized farms occupy roughly two-thirds of Europe's agricultural land (Andersen et al., 2007; European Commission, 2018). The other third of the agricultural land is mostly under extensive farming and often located in mountainous and/or remote areas with natural production constraints, e.g. low soil fertility (European Commission, 2018). Europe is self-sufficient regarding cereals and vegetables, depends on imports for tropical products (e.g. coffee, cane sugar, palm oil) and animal feed, and mainly exports dairy products, processed foods (European Environment Agency, 2020) and meat (Chatellier, 2021).

Through its regional diversity in climate, environment and cultures, European agriculture comprises many different types of farms regarding specialization, size, intensity and land use (Andersen, 2017; Andersen et al., 2007). Within a farm type, further variation can be found (e.g. small-scale farms; Guarín et al., 2020). In addition, within a relative homogeneous farm type regarding specialization, size and intensity, different farm types can be distinguished regarding their orientation (e.g. Mandryk et al., 2012). Apart from the farmers in one particular region, there is a wide diversity of actors that influence farmers (Meuwissen et al., 2019; Urguhart et al., 2019), for instance national and regional governments, social and environmental NGOs and value chain actors. The combination of actors across agricultural systems can vary widely, which is most likely related to the diversity of agricultural products in Europe; the degree of technology required for cultivation, whether the produce is fresh or can be stored, the degree to which produce needs to be processed and whether it is intended for export or national/local consumption. The high degrees of technology use and long value chains in most European food systems increase the risk for low added values for farmers, power concentration of supplying and processing actors and a lock-in of interest of all system actors (see e.g. Balmann et al., 1996; Plumecocq et al., 2018 on this topic). This limits the capacity of European agricultural systems to make necessary adaptations in case of low sustainability and resilience.

1.2.2. Performance

In Europe, agricultural food production has increased substantially since 1945. Cereal yields per hectare, for instance, have increased with over 150% since 1961 (Giller et al., 2021). From an economic perspective, however, contemporary European agriculture is associated with low labour productivity and low income compared to other economic sectors (DG-AGRI, 2017; European Commission, 2018), and decreasing numbers of farms and job opportunities (European Commission, 2013). Even with direct-income support from the European Union, around 70% of farmers in Europe earn less than an average wage (DG-AGRI, 2017). Across years, farmers experience a large variation in income (DG-AGRI, 2017). With continued exposure to global markets in combination with increased weather variability due to climate change, the prices of agricultural commodities are expected to become more volatile (European Environment Agency, 2020) and thus farmers' incomes as well.

Farmers in Europe are (to a certain extent) acknowledged for their important role in maintaining rural landscapes through which they provide public goods to society. However, farm practices may also result in trade-offs with the environment. In Europe, different agro-environmental sustainability indicators show distinct dynamics. For instance, the populations of farmland birds and grassland butterflies have declined by 30% in the EU-28 since 1990 and continue to decline, even with the current measures in place (European Court of Auditors, 2020), while pressure on the environment through nitrogen and phosphorus

surpluses remains at more or less stable levels after roughly 2010. The environmental cost (water pollution, loss of diversity, climate change) of nitrogen surplus in Europe has been estimated to be between 70 and 320 billion Euro per year (Sutton et al., 2011). For a comparison: the total budget of the current Common Agricultural Policy is 365 billion Euro for seven years (2021-2027). Greenhouse gas emissions of EU-28 agriculture, accounting for roughly 10% of emissions in these countries, have declined over the period 2000-2016, but the decline has slowed down and has reversed in some countries (European Commission, 2018).

From a social perspective, rural areas in Europe are associated with an aging population and lower (selfreported) well-being and education levels compared to urban regions (Eurofound, 2019). These social aspects are reflected in agriculture. For instance, more than half of the farmers in the EU is over 55 years old (European Commission, 2018); agriculture knows long working hours and increased health risks compared to other sectors in the economy (Eurofound, 2017); and, except for a few countries, over half of the farmers in EU member states have not received any formal education related to their profession (European Commission, 2018). Farm numbers in the EU have declined by about 25% from 2005 till 2016, while the utilized agricultural area has remained stable. The remaining farms have become larger in terms of area (Eurostat, 2018) and economic production (European Commission, 2013), requiring a higher investment for potential farm successors. Generational renewal of Europe's (family) farms is needed, but increasingly problematic due to the low financial and social perspectives for potential farm workers and successors (Coopmans et al., 2021; Suess-Reyes and Fuetsch, 2016).

A diverse set of policies are put in place to improve the performance of Europe's agricultural systems. However, their success varies and often negative externalities occur. Income or production subsidies, for instance, are known to primarily support farms with relatively high environmental foot prints (FAO UNDP and UNEP, 2021). Another example is the observation that current EU policies for stimulating efficient water resource management in agriculture actually seem to stimulate increased water usage (European Court of Auditors, 2021). Also the European Commission's new Farm to Fork strategy may result in negative externalities as it may imply a greater dependence on import, resulting in environmental pressure elsewhere (Fuchs et al., 2020). In addition, this makes food supply in the Europe more vulnerable as at least 40% of imports is vulnerable to drought (Ercin et al., 2021). It should be noted however that new public policies also have the potential to avoid supply risks by reducing the demand and import of vulnerable products regarding their availability and supply. This potential of public policies is actually seen as an opportunity for reducing products with a large environmental footprint (European Environment Agency, 2020). It should also be noted that any policy change to address the above mentioned issues regarding agricultural performance may be experienced as a severe challenge by actors in farming systems (Spiegel et al., 2020), even when it actually could lead to a win-win situation. This applies in particular to environmental sustainability as policies aimed at this often limit the current basket of options of farmers, while benefits may only be experienced in the long-term. The creation, implementation, monitoring and evaluation of new policies should therefore be well-considered.

1.3 Sustainability and resilience

What becomes clear from the previous sections is that the sustainability of agricultural systems is generally at stake and their resilience is decreasing while their environmental, social and economic context is becoming more instable. Sustainability and resilience needs, therefore, to be assessed, monitored and improved.

Sustainability of a system is in this thesis defined as an adequate performance of all system functions across the environmental, economic and social domains (see e.g. Morris et al. 2011, König et al. 2013). In line with the Brundtland report (United Nations, 1987), an adequate performance is realized when functionality can be preserved for future generations. Sustainability of agricultural systems has been studied extensively (Godfray, 2015; Pretty, 2008), but existing frameworks and tools are not designed to study resilience, i.e. dealing with disturbance. Moreover, in agricultural sustainability assessments, the social aspects are often least integrated, compared to the economic and environmental aspects (Helfenstein et al., 2022). Of particular importance for improving environmental, social and economic

sustainability and resilience are participatory assessments that are designed to come up with adaptation options and action-oriented approaches together with relevant stakeholders (Ridder and Pahl-Wostl, 2005; Toth, 2001).

Resilience has been defined in different ways, dependent on the discipline and context in which it was studied (Brand and Jax, 2007), Ecological resilience can be defined as the capacity to resist change without changing its feedback system and functionality, i.e. robustness, while acknowledging the possibility of alternative stable states (Gunderson and Holling, 2002). The acknowledgement of possible stable alternative states of a system has led scholars to argue that ecological resilience thinking should encompass, besides robustness, the system's capacity to adapt or organize structural and functional change, too (Anderies et al., 2013; Folke et al., 2010). The notion of ecological resilience finds its origin in extensive studies on socio-ecological systems with a strong natural component and relatively clearly delineated system boundaries, e.g. a regional ecosystem, such as a lake. However, agricultural systems also have a strong technological component (Blomkvist and Larsson, 2013) and their boundaries are difficult to delineate (see e.g. Giller, 2013). A definition of resilience geared towards technology, i.e. engineering resilience, is the capacity to return to its stable state after a perturbation, i.e. robustness without considering the possibility of an alternative state (Pimm, 1984). For agricultural systems, the choice for applying an ecological or technological definition of resilience is likely to be influenced by disciplinary background (Brand and Jax, 2007) and the degree of control over a system that is assumed (Hoekstra et al., 2018).

In this thesis, the ecological resilience thinking is taken as point of departure. This allows for evaluating the seemingly much-needed change of European agricultural systems to improve their sustainability and resilience simultaneously. In fact, sustainability and resilience should be considered as complementary concepts (Marchese et al., 2018; Meuwissen et al., 2020, 2019; Westley et al., 2011), as sustainability is aimed at preserving system functioning in the long-term, while resilience realizes a continued functioning in the meantime while being faced with disturbances (Tendall et al., 2015). As agriculture, by definition, supposes some form of control over the natural environment, notions of engineering resilience are also included in this thesis.

Resilience of agricultural systems has been studied at a conceptual level (e.g. Callo-Concha and Ewert, 2014; Ge et al., 2016; Prosperi et al., 2016; Tendall et al., 2015) and resilience indicators have been proposed (e.g. Peterson et al., 2018; Prosperi et al., 2016). The actual operationalization has taken place only to a limited extent. In particular, quantitative assessments of agricultural system and farm resilience are lacking in literature (Dardonville et al., 2021; Thomas Slijper et al., 2021). Lack of good data may be one of the reasons for this. Much resilience work on agricultural systems is therefore qualitative in nature, for instance involving assessments based on system parameters that supposedly bring resilience to the system (Cabell and Oelofse, 2012). The examples in which the resilience concept of agricultural systems is operationalized (e.g. Ashkenazy et al., 2018; Kinzig et al., 2006; van Apeldoorn et al., 2011) are not guided by an integrated resilience framework that was applied to many different agricultural systems. It can be concluded that, in order to address resilience of agricultural systems, approaches and methods need to be borrowed from different research fields. Agricultural systems can be defined as socio-ecological systems in which large technological systems interact with common pool resources, such as water reserves (Blomkvist and Larsson, 2013). Such a definition implies that, besides resilience thinking, notions from different disciplines need to be considered: 1) as in most agricultural systems multiple farm(er)s and other actors have different kinds of agency (Mathijs and Wauters, 2020), collaboration with all actors is needed to collectively manage the common pool of resources (Ostrom, 1990). This implies that individual and collective decision making processes need to be understood, e.g. in relation to risk management and adaptive capacity (Slijper et al., 2020), learning (Slijper et al., 2022; Urguhart et al., 2019) and farm succession (Bertolozzi-Caredio et al., 2020; Coopmans et al., 2021); 2) the technological aspect of agricultural systems makes that the boundaries of the system are diffuse regarding the influence of sociotechnical regimes and the introduction of innovations (Geels, 2011). This implies that understanding of the potential (technological) transition process of the system needs to be understood as well (e.g. Geels, 2011; Termeer and Dewulf, 2019); 3) combining the first two points, a notion of agency (capacity to decide) and structure (that what determines/limits decision making) is necessary in terms of identifying the

boundary of the system and in terms of which actors inside and outside the farming system can and need to take action.

Finally, it should be noted that the notion of planetary boundaries is firmly grounded in common pool resource management research and resilience theory (Biermann and Kim, 2020), i.e. through the perception of a sustainable safe operating space for the global commons and simultaneously considering environmental limits after which non-linear changes are expected. However, to be useful for the diverse set of multi-dimensional agricultural systems, socio-economic boundaries need to be included and boundaries should be assessed more locally (Biermann and Kim, 2020). Dearing et al., (2014) argue more specifically that the planetary boundaries concept is best discussed at regional level because governance systems are more developed at that level, compared to the planetary level. Although global orchestration is needed to deal with, for instance, climate change, some other environmental problems are much more local or regional than global, e.g. nitrogen pollution (Schulte-Uebbing and de Vries, 2021). Finally, critical boundaries, whether global or more regional, should be assessed by a diversity of (local) actors, rather than a selected group of scientists (Biermann and Kim, 2020).

1.4 Framework to assess resilience and sustainability of farming systems

The work presented in this thesis was part of the EU Horizon 2020 project SURE-Farm to assess the sustainability and resilience of European farming systems. SURE-Farm focused in particular on risk management, farm demographics, policies, public & private goods and an enabling environment in 11 case studies (see below). The work presented in this thesis was part of SURE-Farm's work package 5 on the integrated impact assessment of resilience-enhancing strategies on the provisioning of public and private goods. Within SURE-Farm, a framework was developed that considers the need for combining approaches, having a local focus and stimulating participation of relevant (local) actors when assessing the sustainability and resilience of farming systems.

The SURE-Farm framework proposes five steps to assess farming system sustainability and resilience (Meuwissen et al., 2019; Figure 1.1). From step 1 till 5 specific resilience is addressed, i.e. resilience in case the type of shock is known (Walker and Salt, 2012). Steps 1 to 3 relate to the questions "of what?", "to what?" and "for what purpose?". Step 4 addresses the farming system specific resilience capacities that need to be developed. Based on the previous steps, system characteristics are identified that convey general resilience to the system, regardless the type or shock (resilience attributes; Step 5).

1.4.1 Step 1. Resilience of what? (Farming system)

Combining the reflections on sustainability and resilience approaches in the previous section 1.3., it can be concluded that a system delineation is needed that includes interacting stakeholders, the potential for including local knowledge and solutions and interactions with an enabling environment.

In this study we take the farming system as the focus level. In the context of resilience, the farming system relates to the question "Resilience of what?" (Meuwissen et al., 2019). The social delineation of farming systems includes farms producing the main products of interest in a regional context with relatively homogeneous conditions with regard to soils, climate, demographics and institutions. Farming system actors included in the farming systems are the producers of main products and other actors that mutually influence one another (Meuwissen et al., 2019). The geographical delineation of the system follows more or less the common pool resource of interest, e.g. an ecosystem, watershed, or the highest governance level that encompasses the common pool resource. This definition provides flexibility with regard to determining the geographical size of the farming system. In more abstract terms, this implies that a farming system should be delineated at a level where emergent properties are expected anywhere above farm level (K. E. Giller, 2013) and below the governance level that encompasses the common pool resource of system for the system for pool resource of interest.



Figure 1.1. Resilience framework used in this thesis. Source: Meuwissen et al. (2019)

1.4.2 Step 2. Resilience to what? (Challenges)

The agricultural production process of farming systems is faced with environmental challenges such as weather extremes and disease pressure. Because of their open boundaries, most European farming systems are also facing challenges in the institutional, economic and social domain. Institutional challenges may be the result of social or environmental challenges that are primarily observed by actors outside the farming system. For instance institutional challenges arising from regulations on crop protection products in the context of safeguarding public health and reducing environmental pollution.

In case of low sustainability, for instance in the social domain, challenges may be experienced as stemming from within the farming system, i.e. dynamics in systems caused by low sustainability may push the system further towards critical thresholds. For instance, rural depopulation may lead to lack of farm labour, causing farm exit, which contributes to further rural depopulation.

1.4.3 Step 3. Resilience for what purpose? (Functions)

The output is an important element of a system. The output of a system can be seen as the reason for the system to be maintained, which, in the case of this thesis, is to provide goods to society. In the resilience framework, four private and four public goods (or 'functions') across the economic, social and environmental domains are considered (Table 1.1). Farming systems can provide multiple of these functions simultaneously. Farming systems will differ in the emphasis put on different functions. For instance, intensive agricultural systems are generally emphasizing food production rather than maintaining (local) biodiversity & habitat. In this thesis, the focus is mostly on the categorization of functions according to domains: for society, being the ultimate benefiter of the provided goods, all functions are important.

Function	Description	Domain
	Private goods	
Food production	Deliver healthy and affordable food products	Economic
Bio-based resources	Deliver other bio-based resources for the processing sector	Economic
Economic viability	Ensure economic viability (viable farms help to strengthen the economy and contribute to balanced territorial development)	Economic
Quality of life	Improve quality of life in farming areas by providing employment and offering decent working conditions. <i>Public goods</i>	Social
Natural resources	Maintain natural resources in good condition (water, soil, air)	Environmental
Biodiversity & habitat	Protect biodiversity of habitats, genes, and species	Environmental
Attractiveness of the area	Ensure that rural areas are attractive places for residence and tourism (countryside, social structures)	Social
Animal health & welfare	Ensure animal health & welfare	Environmental

Table 1.1. System functions considered in the resilience framework of Meuwissen et al. (2019).

1.4.4 Step 4. Which resilience capacities?

Withstanding shocks, adapting to global change and improving sustainability at the same time requires at least two, and probably three resilience capacities: robustness, adaptability and transformability (Anderies et al., 2013; Folke et al., 2010; Meuwissen et al., 2019; Walker et al., 2004). For farming systems, Meuwissen et al. (2019) define robustness as the capacity to resist to and endure shocks and stresses; adaptability as the capacity to actively respond to shock and stresses without changing farming system structures and feedback mechanisms; and transformability as the capacity of a system to reorganize its structure and feedback mechanisms in response to shocks and stresses (Figure 1.2).



Figure 1.2. Illustration of robustness, adaptability and transformability. Source: Meuwissen et al. (2019)

Adaptability may be employed to make a system more robust, while reducing its transformability. For instance, investment in irrigation infrastructure may increase the robustness of a farming system with regard to drought, but reduces the (economic) resources to keep other adaptation options viable that could lead to a transformation. Robustness and transformability are each other's inverse with regard to approaching critical thresholds: being closer to a critical threshold implies a higher likelihood that a system cannot resist to and endure shocks and stresses, i.e. lower robustness, which leads to a forced reorganization of its structure and feedback mechanism, i.e. a transformation. Transformability can also be expressed more gradually, where transformation is the result of small incremental changes (Meuwissen et al., 2019; Termeer et al., 2017; Figure 1.2). These incremental changes may be the result of deliberately planned adaptations in anticipation of transformations (Enfors-Kautsky et al., 2018). Hence, the distinction between adaptability and transformability is in some cases a result of the time-horizon that is used.

1.4.5 Step 5. What enhances resilience? (Resilience attributes)

After addressing resilience in Steps 1 to 4, it is useful to identify the system characteristics that have realized robustness, adaptability and/or transformability. Based on these characteristics, a system's general resilience can be assessed, even in case an unknown shock or stress would occur (Meuwissen et al., 2021). General resilience helps a system in 1) responding quickly, 2) having (access to) reserves in

times of need, and 3) keeping options open (Walker and Salt, 2012). From a methodological point of view, resilience attributes can be used as a time-efficient way to assess general resilience (Allen et al., 2018).

The initial and basic idea behind general resilience is that the underlying structure of a system, i.e. the slow variables/processes in the system, should allow for resilience. This idea is derived from early resilience work on slow and fast variables in lakes where phosphorus levels in lake sediment (slow variable) are more determinant for resilience than fast variables such as nutrient fluxes. These studies and others (e.g. Bennett and Peterson 2005), while focusing on ecological resilience, propose a system modelling approach to identify quantifiable proxies for general resilience. However, not all important variables in socioecological systems (SES) can be quantified and modelled. Biggs et al. (2012) consulted resilience literature and experienced resilience researchers. Their study yields seven principles of which three relate to the actual socio-ecological system and four to the governance system in place to guide the SES. Walker and Salt (2012) propose six resilience principles that convey resilience, regardless the type of shock or stress: modularity, openness, diversity, tightness of feedbacks, system reserves and high levels of all sorts of capital. Meuwissen et al. (2019) include the resilience principles of (Walker and Salt (2012) and propose that "system reserves" include the different levels of capital. Considering the importance of governance for system resilience (Biggs et al., 2012), these capitals include, for instance, learning capacities (e.g. Spiegel et al., 2020; Urguhart et al., 2019) and policy arrangements to enable farming system's resilience (e.g. Buitenhuis et al., 2020).

The resilience principles enable to compare different socio-ecological systems, but are generally too abstract to be used for specific systems. More concrete characteristics, specified for a farming system, are needed. These concrete characteristics are called "resilience attributes" in this thesis. Nemec et al. (2014), for instance, propose a list of nine resilience attributes for studying watersheds. Cabell and Oelofse (2012) developed resilience attributes for agricultural systems that are widely used (e.g. Tittonell, 2020). These resilience attributes consider agro-ecological aspects such as ecological self-regulation, socio-economic aspects such as being reasonable profitable, and social self-organization, but lack attention for the role of innovation as was put forward by Gunderson and Holling (2002), the role of policies (e.g. Biesbroek et al., 2017; Buitenhuis et al., 2020) and connections with actors outside the farming system (e.g. Mathijs and Wauters, 2020). Hence, adaptations need to be made to tailor the resilience assessment for contemporary European farming systems. Moreover, so far, the resilience attributes have not been formally (and quantitatively) evaluated/tested by researchers and local farming system stakeholders. Because of their origin in ecological research, it is questionable whether the resilience attributes are actually contributing to building resilience of a farming system. Dardonville et al. (2020), for instance, show that conclusions based on quantitative empirical research on the role of diversity of resilience are diverse and context dependent. Finally, in case resilience attributes do contribute to resilience, their specific contribution to robustness, adaptability and/or transformability needs to be considered and evaluated.

1.5 Objectives/Aim

The aim of this thesis is to operationalize the above introduced resilience framework with new and (semi-)quantitative methods and to assess the sustainability and resilience of current and future EU farming systems (Figure 1.3). The following research questions are central in this thesis:

- Is there a balance between social, economic and environmental functions in European farming systems in terms of importance and performance? (Chapters 2 & 3)
- Are European farming systems approaching critical thresholds? (Chapters 5 & 6)
- What resilience capacities do and should European farming systems have? (All chapters)
- What strategies enhance sustainability and resilience of European farming systems? (All chapters)

The methods applied in this thesis follow the steps of the resilience framework as much as possible. Specific attention is given to current (Chapters 2, 3 & 4) and future (Chapters 5 & 6) resilience (Figure 1.3). From a methodological point of view, the methods aim to operationalise the framework by using locally adapted indicators and different sources and types of data. An overview of the concepts used in this thesis is presented in Table A1.1 in the Appendix of this Chapter.

1.6 Case studies

Eleven European farming systems are studied in this thesis: large-scale arable farming in Northeast Bulgaria (BG-Arable), intensive arable farming in the Veenkoloniën, the Netherlands (NL-Arable), arable farming in East of England, United Kingdom (UK-Arable), large-scale corporate arable farming with additional livestock activities in Altmark, Germany (DE-Arable&Mixed), small-scale mixed farming in Nord-Est Romania (RO-Mixed), intensive dairy farming in Flanders, Belgium (BE-Dairy), extensive beef cattle systems in the Massif Central, France (FR-Beef), extensive sheep farming in Huesca, Spain (ES-Sheep), high-value egg and broiler systems in southern Sweden (SE-Poultry), small-scale hazelnut production in Lazio, Italy (IT-Hazelnut), and fruit and vegetable farming in the Mazovian region, Poland (PL-Horticulture). These case studies were selected based on the difference in produce and the challenges that they are facing, and the presence of participants in the consortium of SURE-Farm. These case studies are, obviously, not exhaustive for the diversity of farming systems across Europe. Nevertheless, using the resilience framework as a reference, and having a relatively large number of case studies, allow for comparison between the farming systems regarding challenges, functions, resilience capacities and resilience attributes and strategies.

1.7 Outline

This thesis includes an introductory chapter (Chapter 1), three research chapters on past and/or current sustainability and resilience (Chapters 2, 3 & 4), two chapters on future sustainability and resilience (Chapters 4 & 5) and a general discussion chapter (Chapter 7) (Figure 1.3).

Chapter 2 presents a participatory approach for an integrated assessment of current sustainability and resilience of European farming systems. As the approach has been developed in tandem with the resilience framework of Meuwissen et al. (2019), all steps of the resilience framework are included. The approach in Chapter 2 is illustrated with an application to three specialized farming systems: BE-Dairy, IT-Hazelnut and NL-Arable. Chapter 3 is based on the same participatory approach (Chapter 2) and presents a synthesis of perceived current sustainability and resilience in all 11 European farming systems of the project.



Figure 1.3. Outline of the thesis. Arrows indicate information flows among chapters.

Chapter 4 presents a multi-variate statistical approach in which longitudinal data on weather, economic and farm-level conditions are studied in relation to economic and environmental farm performance. The aim in this work is to detect farm level indicators that may help to improve economic and environmental sustainability and resilience at farm level. We apply the approach to three potato producing regions in the Netherlands: ware potato production in the Dutch provinces Flevoland and Zeeland and starch potato production in the Dutch agricultural region Veenkoloniën (part of NL-Arable). Weather and market conditions are considered as challenges (Step 2 RF). Economic and environmental performance are considered as representative indicators of functions (Step 3 RF). The farm level indicators selected for the analysis are linked to resilience attributes (Step 5). The resilience capacities robustness and adaptability (Step 4 RF) are deduced based on results from the statistical analyses.

Chapter 5 presents a participatory approach for assessing future sustainability and resilience. The chapter commences with an elaboration on the importance of critical thresholds in ecological and agricultural systems. The first half of the approach in Chapter 5 concerns the assessment of critical thresholds of important farming system variables. The second half of the approach concerns the identification of alternative systems and strategies to improve future sustainability and resilience. The approach in Chapter 5 addresses Steps 2-5 of the RF and is illustrated with an application to ES-Sheep. Chapter 6 is based on the same participatory approach as presented in Chapter 5 and makes a synthesis of the outcomes of 11 European farming systems regarding the assessment of critical thresholds.

Chapter 7 provides a synthesis of the research presented in this thesis. Based on the synthesis, policy recommendations are made. Subsequently, I evaluate the framework and methods used in this thesis. After that, I further reflect on the relevance of my work and on ways to evaluate and monitor sustainability and resilience of farming systems in Europe.

Appendix

Table A1.1. Overview of concepts with their explanations and main references as used in this thesis. The table continues on the next page.

Concept	Explanation	References
Sustainability	An adequate performance of all system functions across the environmental, economic and social domain. Obviously adequate is normative and depends on environmental thresholds and societal constraints and objectives.	See e.g. König et al. (2013); Morris et al. (2011)
Resilience capacities	Robustness, adaptability and transformability potential of systems in the face of shocks and stresses. The explanation of the resilience capacities follows below and is influenced by the mentioned sources.	Anderies et al. (2013); Folke et al. (2010); Meuwissen et al. (2019); Walker et al. (2004)
Robustness	Robustness is the capacity to resist to and endure shocks and stresses.	· · · ·
Adaptability	Adaptability is the capacity to actively respond to shock and stresses without changing farming system structures and feedback mechanisms	
Transformability	Transformability is the capacity of a system to reorganize its structure and feedback mechanisms in response to shocks and stresses.	
Specific resilience	Resilience specified with regard to answering the questions "resilience of what, to what and for what purpose?"	Carpenter et al. (2001); Quinlan et al. (2016)
General resilience	General resilience is related to a system's robustness, adaptability and transformability, regardless the type of challenge or shock, including the unknown, uncertainty and surprise.	Resilience Alliance (2010), Walker and Salt (2012), Meuwissen et al. (2019)
Farming system	The basis of a farming systems consists of farms producing the main products of interest in a regional context. Farming system actors included in the farming systems are the producers of main products and other actors that mutually influence one another. In the context of resilience, the farming system relates to the question "Resilience of what?"	Meuwissen et al. (2019)
Challenges	Shocks or stresses that constrain farming system functioning. In the context of resilience, challenges relate to the question "Resilience to what?".	Meuwissen et al. (2019)
Functions	Delivery of public and private goods from the farming system to society (categorized according to the domain they belong to): production of food (economic), bio-based resources (economic), economic viability (economic), quality of life (social), maintenance of natural resources (environmental), biodiversity & habitat (environmental), attractiveness of the area (social), and animal health & welfare (environmental).	Meuwissen et al. (2019)
Function indicators	Indicators that represent farming system functions in the absence of a unique metric for these functions. Indicators with high allocated importance are assumed to represent the identity of the farming system.	Meuwissen et al. (2019)

Concept	Fxplanation	References
concept	Explanation	
Resilience attributes	Specific system characteristics that are supposedly	Cabell & Oelofse
	contributing to general resilience of farming systems. For	(2012), Meuwissen
	the <i>resilience attributes</i> that are treated in this study, see also Table 2.5.	et al. (2019)
Resilience principles	Generic system characteristics that are associated with general resilience: diversity, modularity, openness, tightness of feedbacks, system reserves. The explanation of the principles follows below and is, apart from the mentioned sources, influenced by the work of Biggs et al. (2012).	Resilience Alliance (2010), Walker and Salt (2012), Meuwissen et al. (2019)
Diversity	Diversity in the system with regard to functioning of sub- components and their response to shocks and stresses.	
Modularity	The degree of independence of connected sub- components in the system.	
Openness	Connectivity within the farming system and with systems beyond the farming system.	
Tightness of feedbacks	The degree into which the farming system and its sub- components and processes can create signals and interact in reaction to these internal signals as well as external signals from other (overarching) systems. Included are signals from slow variables and feedbacks.	
System reserves	Natural, economic and social capital that the farming system can access to use as a buffer to compensate for losses or changes in the system during and after a disturbance.	
Critical thresholds Levels at which <i>function indicators, resilience attributes</i> or <i>challenges</i> are expected to cause large and permanent system change.		Adapted from Kinzig et al. (2006) and Biggs et al. (2018).
Enabling conditions	Conditions around the farming system that enable the maintenance of the current system or the realization of alternative systems in the future	This study
Interacting thresholds	Critical thresholds, when exceeded, leading to the exceedance of another critical threshold.	Kinzig et al. (2006)
Current strategies	Strategies implemented to counteract impact of current shocks and stresses on the farming system (indicators).	Meuwissen et al. (2019)
Future strategies	Strategies to maintain the current system in the future or to realize alternative systems in the future.	

Participatory assessment of sustainability and resilience of three specialized farming systems

This chapter has been published as:

Paas, W., Coopmans, I., Severini, S., van Ittersum, M.K., Meuwissen, M.P.M., Reidsma, P., 2021. Participatory assessment of sustainability and resilience of three specialized farming systems. Ecology and Society 26 (2): 2. <u>https://doi.org/https://doi.org/10.5751/ES-12200-260202</u>

Abstract

There is a need for participatory methods that simultaneously assess agricultural sustainability and resilience at farming system level, as resilience is needed to deal with shocks and stresses on the pathways to more sustainable systems. We present the Framework of Participatory Impact Assessment for Sustainable and Resilient FARMing systems (FoPIA-SURE-Farm), FoPIA-SURE-Farm investigates farming system functioning, dynamics of main indicators, and specifies resilience for different resilience capacities, i.e., robustness, adaptability, and transformability. Three case studies with specialized farming systems serve as an example for the used methodology: starch potato production in Veenkoloniën, The Netherlands; dairy production in Flanders, Belgium; and hazelnut production in Lazio, Italy. In all three farming systems, functions that related to food production, economic viability, and maintaining natural resources were perceived as most important. Perceived overall performance of system functions suggest moderate sustainability of the studied farming systems. In the studied systems, robustness was perceived to be stronger than adaptability and transformability. This indicates that finding pathways to higher sustainability, which requires adaptability and transformability, will be a challenging process. General characteristics of farming systems that supposedly convey general resilience, the so-called resilience attributes, were indeed perceived to contribute positively to resilience. Profitability, having production coupled with local and natural resources, heterogeneity of farm types, social self-organization, and infrastructure for innovation were assessed as being important resilience attributes. The relative importance of some resilience attributes in the studied systems differed from case to case, e.g., heterogeneity of farm types. This indicates that the local context in general, and stakeholder perspectives in particular, are important when evaluating general resilience and policy options based on resilience attributes. Overall, FoPIA-SURE-Farm results seem a good starting point for raising awareness, further assessments, and eventually for developing a shared vision and action plan for improving sustainability and resilience of farming systems.

2.1 Introduction

2.1.1 Assessing sustainability and resilience of farming systems

Sustainability and sustainable intensification of agriculture is well studied (Pretty 2008, Godfray 2015) and multiple frameworks and tools for sustainability assessments are available (Alkan Olsson et al., 2009; Arodudu et al., 2017; Sieber et al., 2018). Several approaches are specifically designed for assessing multidimensional sustainability and sustainable development with participatory approaches (Delmotte et al., 2013; König et al., 2013; Morris et al., 2011; Vaidva and Maver, 2014). Yet, these have not been designed to study resilience of agriculture, i.e. the robustness, adaptability and transformability of agricultural systems in the context of shocks and challenges (Meuwissen et al., 2020). Many studies have contributed to the search for operationalization of resilience research in agricultural and food systems (e.g. Callo-Concha and Ewert, 2014; Ge et al., 2016; Peterson et al., 2018; Prosperi et al., 2016). Only few have reflected on addressing agricultural sustainability and resilience simultaneously (e.g. Meuwissen et al., 2019: Tendall et al., 2015). We see sustainability and resilience as two separate, but often complementary concepts that may, or may not influence each other, depending on the situation (Marchese et al., 2018). Assessing both sustainability and resilience is necessary to move beyond static sustainability assessments and to explain dynamics. Addressing both is also needed to identify unsustainable, but resilient systems or vice versa. Resilience is needed to deal with known and novel shocks and stresses, in order to keep track of pathways to more sustainable systems. Other researchers have studied resilience and sustainability of farms (e.g. Darnhofer, 2014, 2010), but did not address the farming system as a whole in which multiple actors beyond the farm also play a prominent role. Ashkenazy et al. (2018) address resilience at farm and regional level, but lack the perspective of a well-defined resilience assessment framework and the attention for the role of different actors. The latter can be evaluated with a participatory assessment which is necessary to adequately address perspectives and issues of multiple actors and issues. Currently, multiple resilience assessment frameworks are available (Douxchamps et al., 2017; Quinlan et al., 2016). The "wayfinder guide" of the Stockholm Resilience Centre (https://wayfinder.earth/) provides an extensive framework for the integration of multiple, iterative sustainability and resilience assessments for social-ecological systems, including notions on persistence, adaptability and transformability. However, to the best of our knowledge, there is no single framework, designed for a one-day workshop, that combined the assessment of multi-dimensional sustainability and resilience (specified and general) of farming systems which include notions of all three resilience capacities, i.e. robustness, adaptability and transformability.

This paper aims to present and test a framework for using a participatory integrated assessment (PIA; Toth 2001, Ridder and Pahl-Wostl 2005) to address perceived sustainability and resilience of farming systems. PIA, in combination with other methods, can contribute to a project cycle where the following steps can be distinguished: data gathering and analysis, planning, implementation, and monitoring and evaluation (Ridder and Pahl-Wostl, 2005). In the PIA presented in this paper, the emphasis is on data gathering and analysis at the start of a project cycle. Participatory input allows to identify the most important indicators of the system from the stakeholders' point of view. The identification of these indicators is a first step towards determining the identity of the system, i.e. mapping its most important structure and feedback mechanisms (Cumming and Peterson, 2017). Participatory input also allows to assess variables that are not readily measurable, e.g. social variables such as satisfaction or pride of being a farmer. Variables from the social domain are potentially important for system functioning, but often neglected in studies for instance related to land use (Gliessman, 2015; Winkler et al., 2018). Participatory research also reveals differences in perceptions of goals and functioning of a system, which is important to take into account when assessing sustainability and resilience (Robards et al., 2011).

2.1.2 Intensive specialized agriculture in Europe

The framework was applied to three intensive, specialized farming systems in the EU through three workshops. Having three farming systems as case studies in different regions allows to demonstrate how the framework works in different conditions. A common denominator, which in our case studies is the intensive and specialized nature of the farming systems, also allows for evaluating the usage of the framework for comparison between farming systems. Applying the proposed framework to intensive,

specialized systems in this study is relevant and interesting from different perspectives. Regarding representativeness, the bulk of the food produced within the European Union is provided by medium to highly intensive, specialized farming systems (Andersen et al., 2007). Regarding sustainability, more intensive agricultural systems are associated with higher yields and revenues through economy of scales (Peterson et al., 2018), but also with higher pressure on the environment, certainly per unit of area (Pretty and Bharucha, 2014; Tilman et al., 2002). As to resilience, specialized and intensive agricultural systems are often optimized for production under stable socio-economic and biophysical environments (Urruty et al., 2016). However, many external influences, such as increased occurrence of weather extremes and volatile markets, create a more variable production environment than most specialized agricultural systems are designed for. Farms in such systems often have a relatively high share of financial capital invested in production equipment that cannot be reinvested without making substantial losses (sunk costs). Sunk costs create path-dependency and lock-in of individual farmers, which makes it difficult for the whole agricultural system to adapt and transform if new challenges arrive (Balmann et al., 2006). In addition, the professional network required for providing inputs to and processing outputs from intensive, specialized systems can create a lock-in of stakeholders' interests inside and outside the agricultural system. Examples may refer to businesses that need to fulfil shareholder expectations regarding economic profit (Westley et al., 2011), or intellectual property rights in technology intensive agriculture (Plumecocg et al., 2018).

2.1.3 Concepts used in this study

In this study we define a farming system as a geographical region with relatively homogenous agroecological and social conditions. In the farming system, we start with one farming sector and its farms as the focal point of attention and then include all actors in the farming system who influence the focal farmers and who are themselves influenced by focal farmers (Meuwissen et al., 2019). Our working definition for sustainability of farming systems is an adequate performance of all system functions across the environmental, economic and social domain (see e.g. Morris et al. 2011, König et al. 2013). Obviously 'adequate' is normative and depends on environmental thresholds and economic and societal constraints and objectives. For resilience, we distinguish three resilience capacities; robustness, adaptability and transformability. Robustness is the capacity to resist to and endure shocks and stresses; adaptability is the capacity to actively respond to shocks and stresses without changing farming system structures and feedback mechanisms; and transformability is the capacity of a system to reorganize its structure and feedback mechanisms (Meuwissen et al., 2019). We consider along with Walker and Salt (2012) that to assess resilience, we need to distinguish between specified and general resilience. Assessing specified resilience relates to the questions of resilience "of what?", "to what?" and "for what purpose?" (Carpenter et al., 2001; Ouinlan et al., 2016). General resilience relates to system's robustness, adaptability and transformability, regardless the type of challenge or shock. General resilience is mainly assessed by looking at system principles that are presumably conveying resilience. The five principles for general resilience that we use in this paper are tightness of feedbacks, modularity, diversity, openness, and system reserves (Resilience Alliance, 2010). Based on these principles, more concrete resilience attributes or indicators are needed in order to assess the general resilience of a specific system. Cabell and Oelofse, (2012), for instance, present 13 resilience indicators for agro-ecosystems that are considered as conveying resilience to the system. Some of these indicators are somewhat linked to either robustness, adaptability or transformability, although not in a consistent manner, as we will elaborate further in this study by addressing potential contributions of resilience attributes to the three defined resilience capacities. Table A1.1 in the Appendix of Chapter 1 provides an overview of all important concepts used in this study.

2.2 Methods

To study the perceptions of farming systems' current sustainability and resilience, we designed the Framework of Participatory Impact Assessment for Sustainable and Resilient EU Farming systems (FoPIA-SURE-Farm: Paas et al., 2019: Reidsma et al., 2019), FoPIA-SURE-Farm includes elements from five different existing sources. First, for assessing perceived sustainability, it is inspired by the semiquantitative approach of the Framework for Participatory Impact Assessment (FoPIA; Morris et al. 2011, König et al. 2013), Second, for assessing perceived resilience, elements from the Resilience Assessment Framework are implemented (RAF; Resilience Alliance, 2010). Third, for analyzing dynamics of sustainability indicators, we included participatory techniques used for system dynamics modelling by Herrera (2017). Fourth, general resilience was assessed based on a list of resilience attributes as proposed by Cabell and Oelofse (2012), which was tailored and complemented for the assessment of farming systems, Last, FoPIA-SURE-Farm builds on the framework developed by Meuwissen et al. (2019) in the context of the SURE-Farm project (https://surefarmproject.eu). Meuwissen et al. (2019) propose to investigate farming system resilience by answering the questions a) Resilience of what? (defining the farming system), b) Resilience to what? (identifying challenges), c) Resilience for what purpose? (identifying main goods and services delivered by farming systems to society), d) What resilience capacities? (assessing robustness, adaptability as well as transformability), e) What resilience attributes? (identifying system characteristics that convey resilience to the system). These questions facilitate the framing of general and specific research topics related to resilience for which qualitative as well as quantitative research methods can be applied.

2.2.1 Farming systems

Compared to studies at farm level, a study at the farming system level allows to take into account challenges that operate at similar levels of integration (Peterson et al., 2018), such as the decrease of farm numbers and specific climate changes in a region. Also, at the farming system level, processes and actors that can influence system dynamics in the face of challenges, such as stakeholder interaction with the environment (Urruty et al., 2016) and self-organization (Cabell and Oelofse, 2012), can be included. Consequently, at this level the system dynamics are not entirely due to external challenges, but include strategies by different actors to adapt and transform (Cumming et al., 2017). At the same time, the farming system is a level above the farm (K. E. Giller, 2013) at which individual stakeholders can still be heard (Cabell and Oelofse, 2012), allowing for multiple stakeholder input in participatory settings.

2.2.2 Case studies

Workshops were held in three specialized farming system case studies (CS): starch potato production in Veenkoloniën, the Netherlands (NL-Starch potato), dairy production in Flanders, Belgium (BE-Dairy) and hazelnut production in Lazio, Italy (IT-Hazelnut).

NL-Starch potato is a capital and input-intensive system with relatively low economic productivity per unit of input (Table 2.1). Predominant soils are sandy with a high amount of inactive organic matter. The most economically productive crop, starch potato, is typically grown with cereals and sugar beets in a narrow 1:2 or 1:3 rotation. On some farms also onions, carrots or tulips are cultivated. Relative to cereals and sugar beets, most crop protection products are applied to starch potatoes. Farmers are organized in a cooperative that processes the starch potatoes, which are often grown on a contract basis. Sugar beets are also grown on contract and sold to a cooperative. Main challenges in NL-Starch potato are low economic productivity, plant parasitic nematodes in the soil, and changing policies and legislation. In NL-Starch potato, the number of farmers is decreasing and the prices of agricultural land are increasing.

BE-Dairy is also a capital intensive system (Table 2.1). Livestock diets contain mainly grass (silage), supplemented with maize silage and feed concentrates. Farmers are organized in cooperatives that collect, process, and market the milk and its derived products. Important challenges are competition on export markets and fluctuating prices of milk and feed. Nitrogen surpluses put pressure on the environment. Other debated subjects in this system are the production of greenhouse gasses and the use of antibiotics. Future farm succession is a concern in this farming system due to the capital intensity (Table 2.1), a decreased

interest into farming of the younger generation (relating to a competition with other occupations) and the challenge of administrative and legislative demands.

IT-Hazelnut is the least capital intensive system (Table 2.1). Use of crop protection products and abstraction of ground-water for irrigation have been claimed to put a pressure on the environment, especially on surface waters causing public concern (Liberti, 2019). Most of the production is sold to processing facilities outside the farming system. In the region some cooperatives collect the raw product and perform only the first processing step (i.e. shelling) and provide storage services. Main challenges for this system are price instability and competition on the world market, mainly with Turkey. Recent modernization of harvesting through auto-propelled machines has increased labour productivity and instigated a demand for more land for hazelnut cultivation. As a result, land that is less suitable for hazelnut cultivation will be taken into production in the coming years. Results from FoPIA-SURE-Farm in IT-Hazelnut are presented in detail in Nera et al. (2020). Compared to Nera et al. (2020), this study puts more emphasis on presenting and evaluating the methodology when applied to three different farming systems. Nera et al. (2020) evaluate sustainability and resilience in the case study in more detail.

Indicator	Unit	BE-Dairy	IT-Hazelnut	NL-Starch potato
NUTS-areas CS		BE02	ITI41	parts of NL111,NL112, NL131, NL132 $^{+}$
Number of farms of interest in sector of CS	#	2756 ¹	5,640 ²	727 ³
People working in agriculture in CS [‡]	AWU [§]	4841	11,226.5 ²	1209
Total area CS	1000 ha	1,352	361	140
Total agricultural area CS	1000 ha	623	242	84
NUTS-areas harmonized public FADN-dataset ¹⁴		BE02	ITI4 ¹	NL01 [#]
Area per farm	ha	46.6	10.0	61.9
Labour input per farm	AWU	1.76	1.13	1.66
Economic size of farms	European Size Unit (1,200 gross margin)	256	55	203
Intensity (Input per area)	€/ha	4097	2419	4669
Intensity (Output per area)	€ / ha	4858	5298	5184
Crop output	€ output from crops / € total output *100	6.5	97.4	79.8
Livestock output	€ output from livestock / € total output *100	91.3	0.2	0.3
Other output	€ other output / € total output *100	2.2	2.4	19.8
Rentability	€ output / € input	1.19	2.17	1.11
Total subsidies per farm- excluding subsidies on investments	1,000€	21	3	28
Family farm income	1000 € / family work unit	33.8	33.9	46.5
Leverage ratio per farm	€ total liabilities / € total assets	0.20	0.00	0.22
Fraction of sunk costs per farm	€ fixed assets / € total assets	0.83	0.65	0.89
Cost of crop protection	€ / ha	83	138	445
Cost for chemical fertilizer	€/ha	158	269	223

Table 2.1. Average farming system ch	aracteristics related to size,	economic performance,	specialization and	intensity.
--------------------------------------	--------------------------------	-----------------------	--------------------	------------

[†]The CS area does not follow the contours of administrative boundaries, as a result, data at the scale of this CS is based on data from two encompassing, and hence larger, nationally defined areas: "Westerwolde and Groninger Veenkoloniën" and "Drentse Veenkoloniën en Hondsrug", [‡]People working in agriculture is calculated as "number of farms of interest of sector in CS" * "Labour input per farm", [§]AWU: Annual Work Unit, equaling 1800 working hours, ¹Below this point in the table, data is derived from the public data base of the Farm Accountancy Data Network of the European Commission, [§]ITI4: Lazio region, encompassing the CS, [‡]NL01: The Netherlands, encompassing the CS.

¹Departement Landbouw en Visserij (2019), ²Italian National Institute of Statistics (2020), ³ Centraal Bureau voor de Statistiek (2019), ⁴Farm Accountancy Data Network of the European Commission (2019).

2.2.3 Stakeholder participation

Multiple actors influence the dynamics of a farming system. The heterogeneity of actors included in the systems has hence been accounted for in the workshops. Participants were invited via existing stakeholder networks in the case study areas. Attendance to the workshop was based on participants' own initiative and therefore not necessarily balanced across stakeholder groups. Participants mainly consisted of farmers and representatives from the government, NGOs, research institutes and the processing industry (Table 2.2). Participant numbers varied across case studies (from one to eight per stakeholder group, and 12 to 21 in total).

Country	Workshop dates	Partici- pants	Farmers	Industry	Govern- ment	NGO	Research /consultancy	Miscellaneous
BE-Dairy	27-11-2018	16	5	5	2	1	1	2
IT-Hazelnut	21-01-2019	21	8	3	3	3	4	-
NL-Starch potato	11-12-2018	12	4	1	3	1	3	-

Table 2.2. Overview workshop dates and number of participants.

To bring all participants to the same level of analysis, the research team, which differed in each case study, started the workshop by presenting the social delineation of the farming system (Table 2.3), showing farming system actors and (in)direct influencers of the farming system. Participants were given the opportunity to react to the farming system representation. Updates were made if necessary.

2.2.4 Assessing sustainability of farming system functions

Eight functions of the farming system, along with their representative indicators, all identified by the research team, were presented to the participants and discussed in a plenary session (Table 2.3). If necessary, changes were made to the list of representative indicators (Table 2.4). In BE-Dairy, the function "Bio-based resources" was interpreted broadly as all edible products from the system other than milk. In IT-Hazelnut, "Animal health and welfare" was not assessed because animals are not part of this farming system. Participants were invited to individually assess the importance of the different functions (see Table 2.3 for details). Similarly, they were asked to assess all indicators regarding their degree of representativeness for the function they were to represent. The outcomes for the degree of representativeness were transformed to relative importance in order to compare importance of indicators across functions (see also SM2.1). Furthermore, the performance of each indicator was assessed by each participant. Function performance was calculated per participant as the sum of scores of indicators per function times the average indicator representativeness according to the stakeholder group to which the participant belonged to (see also SM2.1). Importance and performance of functions and indicators was directly fed back and discussed with the participants during the workshop. Perceived indicator and function performance was interpreted as being indicative for perceived sustainability levels of the system (see e.g. Morris et al. 2011, König et al. 2013 who applied this method for assessing policy impact on sustainable development). Perceived importance and performance levels of functions between farmers and nonfarmers were tested for significant differences, using a Kruskal Wallis test in R (R Core Team, 2015). Perceived function importance and performance across functions were tested for significant differences using a Kruskal Wallis test and a post-hoc Conover Iman test with Bonferroni correction using the Rpackage "conover.test" (Dinno, 2017).

Assessment of:	Activity	Format	Scoring
Farming system	Identifying the actors and boundaries of	Plenary	-
delineation	the farming system	discussion	
System	Feedback on list with representative	Plenary	-
sustainability	indicators for system functions	discussion	
	Assessing function importance	Filling in a form	Divide 100 points over the
		individually	eight functions
	Assessing indicator representativeness	Filling in a form	Divide 100 points over the
	per function	individually	indicators per function
	Assessing indicator importance	Calculation	Equation A2.1
	Assessing indicator performance	Filling in a form individually	Score from 1-5; where 1: very poor performance, 2: poor performance, 3: moderate performance, 4: good performance, 5: perfect performance.
	Assessing function performance	Calculation	Equation A2.2
	Discussion indicator and function	Plenary	-
	importance and performance	discussion	
	Selecting indicators for further analysis	Plenary	-
Custom and iline as	Fundamentian of unburstiments of deuterbility.	discussion	
System resilience	and transformability	Presentation	-
	Sketching dynamics of selected indicators	Discussion in small groups	-
	Identifying major challenges and strategies	Discussion in small groups	-
	Assessing strategy implementation	Filling in a form individually	Score from 1-5 for implementation; where 1: not to very poor, 2: poor, 3: moderate, 4: good, 5: perfect implementation.
	Assessing the contribution of strategies to robustness, adaptability and transformability	Filling in a form individually	Score from -3 to +3 for contribution; where 0: no, 1: weak, 2: moderate, 3: strong contribution, and -: negative, +: positive contribution
	Assessing presence of resilience attributes	Filling in a form individually	Score from 1-5 for presence; where 1: not to very poor, 2: poor, 3: moderate, 4: good , 5: perfect presence.
	Assessing the contribution of resilience attributes to robustness, adaptability and transformability	Filling in a form individually	Score from -3 to +3 for contribution; where 0: no, 1: weak, 2: moderate, 3: strong contribution, and -: negative, +: positive contribution

Table 2.3. FoPIA-SURE-Farm workshop design. For details see Reidsma et al. (2019).

Farming system function	Indicators NL-Starch notato	Indicators BE-Dairy	Indicators IT-Hazelnut
Deliver healthy and	Starch potato production	Total milk production	Hazelnut production
affordable food products	(t/ha)	Flanders	•
(Food production)	Sugar beet production	Real milk price for	Hazelnut quality
	(t/ha)	consumers [†]	
	Cereal production (t/ha)	-	-
Deliver other bio-based resources for the	Diversity of industrial potato products	Tons of meat produced	Shell production for heating
processing sector (Bio- based resources)	Straw production (t/ha)	Tons of crops produced	Production of pruning waste for energy generation
	-	Total number of farms with bio-gas systems	-
Ensure economic viability (viable farms help to	Profit (Euro/ha)	Share of total farm income from milk	Gross Margin per hectare
and contribute to balanced	Income from agricultural activities (%)	Labour income	Public support to agriculture (CAP and RDP)
territorial development) (Economic viability)	Land prices	Gross margin per liter of milk	Margin from in situ processing activities
Improve quality of life in farming areas by providing employment and offering	Working hours per year per farmer	Average amount of working hours per farmer per day	Number of people in the area employed in the farming system
decent working conditions. (Quality of life)	Employment related to agriculture	Number of fully employed workers per farm	Percentage of women among the people employed in the system
	Satisfaction of being a farmer	Pride of profession	Health of agricultural workers
	Women working in agriculture (%)	-	-
Maintain natural resources in good condition (water,	Greenhouse gas emissions	Soil quality	Groundwater availability
soil, air) (Natural	Soil quality	Water quality	Water quality in the area
resources)	Regional water availability Responsible use of	-	-
Protect biodiversity of	Responsible use of crop	Genetic diversity of	Diversification in land use
species (Biodiversity & habitat)	Number of bird species	Share of ecologically valuable grassland	Number of organic farms
	Surface of land with nature friendly management	Responsible use of crop protection	-
Ensure that rural areas are attractive places for residence and tourism (countryside, social	Unhealthy stress under farmers	Extent to which farms are involved in public activities such as; education, tourism, healthcare.	Touristic flow
structures) (Attractiveness	Farms with broadened	Share of farms with outside	Retention of young people
of the area)	activities	grazing	in the area
	Villages with a minimum of one school and	Income from farm tourism	-
Ensure animal health & welfare (Animal health &	Farms with certificates for animal welfare	Longevity	-
welfare)	Responsible use of antibiotics	Amount of antibiotics per cow	-

Table 2.4. Overview of farming system functions and case study specific indicators representing those functions. The first four functions are private goods, and the last four are public goods.

[†] Participants in the Belgian case study insisted on "Real price for consumers", because in their point of view it relates to affordability from the perspective of consumers. This pronounced preference was not ignored because of the participatory setting and to stimulate participants to keep giving input.
2.2.5 Assessing resilience

Based on indicator importance and performance, participants decided in a plenary session which 3 to 4 indicators were most interesting to assess in further detail with regard to farming system resilience. Participants were invited to sketch the yearly dynamics of the selected indicators over the timespan 2000-2018, and to identify challenges that induced the sketched dynamics. Also they were invited to identify strategies that have been applied by farmers and other farming system actors to deal with the identified challenges. Identified strategies were assessed for their level of implementation and for their contribution to resilience capacities. Results were directly fed back to participants in a plenary setting. In the evaluation phase, strategies were linked to resilience attributes in order to visualize the connection between specified and general resilience and to allow for comparability of strategies between case studies.

To assess general resilience, a list with resilience attributes was constructed (Table 2.5) based on Cabell and Oelofse (2012) and Meuwissen et al. (2019) to serve the purpose of this study within the context of the SURE-Farm project. This implied that details on the farming system were added in the resilience attribute description. We also split up certain attributes as provided by Cabell and Oelofse (2012). Finally, we arrived at a list with 22 resilience attributes from which we selected 13 to study in the workshop to not overtask the participants. This selection was based on the 1) SURE-Farm research focus, 2) paying equal attention to the different resilience principles, and 3) avoiding overlap between attributes. For instance, "Reflective and shared learning" is partly dependent on "Social self-organization", which is why we selected the latter as the overarching attribute. We use the term "resilience attribute", which refers to a higher hierarchical level compared to the term "resilience indicator" as was originally used in Cabell and Oelofse (2012). This distinction also helps to avoid confusion with the concept of 'function indicators' as used in this study. Resilience attributes were assessed for their level of presence and for their contribution to resilience capacities. Due to time limitations, results could only be fed back into a limited extent during the workshop. Perceived contribution of resilience attributes to resilience capacities across attributes was tested for significant differences using a Kruskal Wallis test and a post-hoc Conover Iman test with Bonferroni correction in R using the "conover.test" package (Dinno, 2017).

Table 2.5. Resilience attributes used in FoPIA-SURE-Farm based on Cabell & Oelofse (2012) and Meuwissen et al. (2019). The resilience attribute name and the explanation statement were presented to stakeholders to assess presence and contribution to resilience capacities. The table continues on the next page.

Resilience attribute	Definition	Implications	Explanation statement	Link with
				resilience
				principle ⁺
Reasonably	Persons and	Being reasonably	Farmers and farm workers	Systems
profitable	organizations in the	profitable allows	earn a livable wage while	reserves
	farming system are	participants in the	not depending heavily on	(economic
	able to make a	system to invest in the	subsidies	capital)
	livelihood and save	future; this adds		
	money without	buffering capacity,		
	relying on subsidies	nexibility, and builds		
	employment	tapped into following		
	employment	release		
Production [‡] coupled	The system functions	Responsible use of local	Soil fertility, water resources	Systems
with local and	as much as possible	resources encourages a	and existing nature are	reserves
natural capital	within the means of	system to live within its	maintained well	(natural
	the bio-regionally	means; this creates an		capital),
	available natural	agroecosystem that		tightness of
	resource base and	recycles waste, relies on		feedbacks
	ecosystem services	healthy soil, and		
From a big on a la alis or a site of	Franking al discusion in	conserves water	These is a high seriety of	Disconsites
Functional diversitys	the variety of	Diversity Duriers against	inputs outputs income	Diversity
	(ecosystem) services	(insurance) and provides	sources and markets	
	that components	seeds of renewal	sources and markets	
	provide to the	following disturbance		
	system	-		
Response diversity§	Response diversity is	Diversity buffers against	There is a high diversity of	Diversity
	the range of	perturbations	risk management strategies,	
	responses of these	(insurance) and provides	e.g. different pest controls,	
	components to	seeds of renewal	weather insurance, flexible	
	environmental	following disturbance	payment arrangements [*]	
Eveneed to	change	Cuch frequent amall	The amount of year to year	Ononnoos
Exposed to	ovposed to discrete	Such frequent, small-	according any ironmontal	Openness
uistui bance	low-level events that	increase system	social or institutional	
	cause disruptions	resilience and	disturbance is small (well	
	without pushing the	adaptability in the long	dosed) in order to timely	
	system beyond a	term by promoting	adapt to a changing	
	critical threshold	natural selection and	environment	
		novel configurations		
		during the phase of		
		renewal; described as		
		"creative destruction"		
Spatial and temporal	Patchiness across the	Like diversity, spatial	There is a high diversity of	Modularity,
form tupost	landscape and	neterogeneity provides	farm types with regard to	diversity
lann types	time	following disturbance	orientation and degree of	
	une	Tonowing disturbance	specialization [‡]	
Optimally redundant§	Critical components	Redundancy may	Farmers can stop without	Modularity
farms [‡]	and relationships	decrease a system's	endangering continuation of	(localancy
	within the system are	efficiency, but it gives	the farming system and new	
	duplicated in case of	the system multiple	farmers can enter the	
	failure§	back-ups, increases	farming system easily [‡]	
		buffering capacity, and		
		provides seeds of		
		renewal following		
		disturbance [§]		

Resilience attribute	Definition	Implications	Explanation statement	Link with resilience principle [†]
Supports rural life ¹	The activities in the farming system attract and maintain a healthy and adequate workforce, including young, intermediate and older people.	A healthy workforce that includes multiple generations will ensure continuation of activities and facilities in the area, and the timely transfer of knowledge.	Rural life is supported by the presence of people from all generations, and also supported by enough facilities in the nearby area (e.g. supermarkets, hospital, shops)	Systems reserves (social capital)
Socially self- organized	The social components of the agroecosystem are able to form their own configuration based on their needs and desires	Systems that exhibit greater level of self- organization need fewer feedbacks introduced by managers and have greater intrinsic adaptive capacity	Farmers are able to organize themselves into networks and institutions such as co- ops, community associations, advisory networks and clusters with the processing industry [‡]	Tightness of feedbacks, system reserves (social capital)
Appropriately connected with actors outside the farming system [†]	The social components of the agroecosystem are able to form ties with actors outside their farming system [‡]	In case self-organization fails, signals can be send to actors that indirectly influence the farming system [‡]	Farmers and other actors in the farming system are able to reach out to policy makers, suppliers and markets that operate at the national and EU level [‡]	Tightness of feedbacks
Legislation [‡] coupled with local and natural capital	Regulations are developed to let ⁺ the system function as much as possible within the means of the bio-regionally available natural resource base and ecosystem services	Responsible use of local resources encourages a system to live within its means; this creates an agroecosystem that recycles waste, relies on healthy soil, and conserves water	Norms, legislation and regulatory frameworks are well adapted to the local conditions ⁺	Systems reserves (social capital)
Infrastructure for innovation	Existing infrastructure facilitates diffusion of knowledge and adoption of cutting- edge technologies (e.g. digital)	Through timely adoption of new knowledge and technologies, a farming system can better navigate in a changing environment	Existing infrastructure facilitates knowledge and adoption of cutting-edge technologies (e.g. digital)	Openness, system reserves
Diverse policies	Various policy instruments stimulate different mechanisms that improve different resilience capacities.	Policies addressing all three resilience capacities avoid situations in which farming systems are permanently locked in a robust but unsustainable situation. Or situations in which adapting and transforming systems are increasingly vulnerable	Policies stimulate all three capacities of resilience, i.e. robustness, adaptability, transformability	Diversity

[†]Link of resilience attributes with resilience principles, as perceived by the authors, [†]Deviating from Cabell and Oelofse (2012) for the purpose of this study, [§]Only part of the original resilience attribute of Cabell & Oelofse is presented, ^INew resilience attributes for the purpose of this study.

2.3 Results

2.3.1 Farming system actors

Participants provided feedback on the social delineation of all three farming systems. For BE-Dairy and NL-Starch potato, many farming system actors and influencers were identified, while these were much fewer in IT-Hazelnut (Figure 2.1). In NL-Starch potato, the cooperative for processing starch potatoes was seen as part of the farming system. In BE-Dairy, the cooperative for processing and distributing milk was moved inside the farming system, after feedback from participants. In IT-Hazelnut, cooperatives exist within the farming system, but main processors of hazelnut were considered to be outside the farming system, because they operate on the international market and are not directly affected by changes within the considered farming system. Local NGOs were mentioned in BE-Dairy and NL-Starch potato, but not in IT-Hazelnut.



Figure 2.1. Farming system visualization after feedback from participants.

2.3.2 System sustainability

In all three case studies, "Food production" and "Economic viability" were considered to be among the most important functions (Figure 2.2). In NL-Starch potato and BE-Dairy, "Maintaining natural resources" was also considered as important. In NL-Starch potato, the function "Food production" was evenly represented by the three main crops: starch potato, sugar beet and wheat (Table 2.4). In IT-Hazelnut, this function was represented by hazelnut quantity and quality. In BE-Dairy, this function was represented by milk production and the price consumers pay for milk in the supermarket. Representative indicators for "Economic viability" related to farm income and profit per hectare. Representative indicators for "Maintaining natural resources" related to soil and water quality.



Figure 2.2. Relative importance (size of symbols and accompanying number) and performance (y-axis) of farming system functions. Relative importance was assessed individually by dividing 100 points over the eight functions (Table 2.4). Performance was assessed on a scale from 1 to 5 (Table 2.3). "Animal health & welfare" was not assessed in IT-Hazelnut.

Performance of "Food production" and "Economic viability" was considered high in IT-HazeInut and moderate in BE-Dairy (Figure 2.2). For "Natural resources", participants in BE-Dairy perceived a moderate to high performance, while participants in IT-HazeInut and NL-Starch potato perceived a moderate performance. Participants in NL starch potato indicated that they found it challenging to assess individual functions, because they perceived functions as interacting with each other. Participants in IT-HazeInut indicated that they perceived that the recent modernization and enlargement of the sector increased pressure on the environment, while neglecting the importance of the function of "Natural resources".

In all case studies, it seemed that farmers perceived "Economic viability" more important and "Natural resources" less important compared to other stakeholders. However, only for "Natural resources" in BE-Dairy, a significant difference was detected. Farmers generally allocated less importance to functions that deliver public goods. At the same time, in BE-Dairy and NL-Starch potato, farmers tended to assess performance of "Natural resources" on average higher than other stakeholders. In BE-Dairy, participants indicated that these results would have been different if more people from nature organizations had participated in the workshop. In NL-Starch potato, farmers also assessed the performance of "Biodiversity & habitat" higher than other stakeholders (SM2.2).

2.3.3 Dynamics of sustainability indicators

In all case studies, participants indicated that they had little knowledge on year-to-year fluctuations of selected indicators. However, they were able to indicate trends and important years with regard to changes in trends, lows and peaks. For each selected indicator, participants identified main challenges and strategies applied to maintain or improve the indicator performance (Table 2.6). According to participants,

important underlying causes for dynamics in NL-Starch potato were nematode pressure and increased costs per hectare that were amongst others counterbalanced by improved potato varieties that resist nematodes, reducing costs and increasing efficiency. The change from production-based to area-based subsidies in 2013 was seen as a big challenge in NL-Starch potato. As strategy to deal with this challenge, the cooperative decided to abandon less lucrative lower quality starch markets and invest in product innovation. In BE-Dairy, participants indicated that main indicators were particularly affected by the abolishment of the milk quota and events that affected the international market. Identified strategies to cope with these challenges were mainly related to improved efficiency, but also included strategies related to risk management and diversification at farm and farming system level. In IT-Hazelnut, dynamics of indicators were perceived to be mainly influenced by the development of new machinery that enhanced labour productivity. Apart from competition at the international market with Turkey, no major challenges were reported. Strategies to maintain or improve main indicators in IT-Hazelnut were related to mechanization, cooperatives, producer groups and using funds of the Rural Development Program of the European Commission (RDP funds).

In all three case studies, indicators related to food production and economic viability were assessed to have improved over time, except in BE-Dairy, where farm income was perceived to be on average stable, but with increased yearly variation. The perspective in all three case studies is more negative for indicators representing the environmental domain. In NL-Starch potato, soil quality was perceived to decline, in BE-Dairy, emissions of carbon were perceived to go up again. In IT-Hazelnut, the indicator "Area of organic hazelnut production" was seen as positively related to the function "Biodiversity & habitat" and was perceived to be increasing. However, participants indicated that they perceived that biodiversity in the farming system was generally decreasing because of the limited habitat provision in the expanding area with hazelnut monocultures. In the expansion areas, hazelnut cultivation also requires more ground water extraction compared to the cultivations of chestnut and olives that it usually replaces

			Main trend in last		
Case study	Function	Indicator	20 years	Main causes of dynamics	Strategies
NL-Starch	Food production	Starch potato	Remained stable	Cultivated area decreased and	Exchanging land with dairy farmers, reduce
potato		production		production per hectare increased;	costs
				decoupled payments reducing subsidies	
	Economic viability	Profit per hectare	Increased, but a	Increased but also variable prices;	Extending knowledge on soils and varieties,
			plateau seems to	higher yields; increased input prices;	scale enlargement, increase value of starch
			be reached	extreme events; decoupled payments	potato products, better varieties, reduce costs,
				reducing subsidies; more alternative	have land available outside contract farming,
				crops	precision agriculture
	Natural resources	Soil quality	Decreased	Nematode pressure, nutrients not being	Improved varieties against nematodes, raising
				replenished, lack of awareness	awareness, replenish soil minerals, avoid
					artificial fertilizer
BE-Dairy	Food production	Total milk production	Increased	Increased efficiency, abolishment of	Increase efficiency, expansion of business
				quota	(own rearing), expansion of business (buying
					cattle), broaden business, futures exchange
	Food production	Real milk price for	Decreased,	Banking crises, Arabic spring, stop on	Open up to international markets, exceptional
		consumers	currently stable	export to Russia	financial support from sectoral federation,
					creating milk powder stocks
	Economic viability	Farm income	Average is stable,	Financial crises, globalization,	Intensification and scale enlargement, cyclic
			but more	decreasing demand from China, export	investing, investments of cooperatives,
			fluctuations in the	ban to Russia	maintain diversity of dairy farms
			last decade		
	Natural resources	Total carbon foot	Gradual decrease	Increased efficiency, abolishment of	More efficient feeding, manure
		print	in 00's, increase in	quota	recycling/circular agriculture, production of
			10's		green energy, increase longevity of cows,
					genetic improvement
IT-Hazelnut	Food production	Gross saleable	Increase, with a	Machinery development, frost in Turkey	Mechanization, cooperatives
		production	peak in 2012-2014		
	Economic viability	Gross margin per	Increase, with a	Machinery development, frost in Turkey	Mechanization, producer organizations
		hectare	peak in 2012-2014		
	Biodiversity &	Organic cultivated	Increase	Launch of tenders for organic	Applying for rural development program funds
	habitat	area		production	
	Attractiveness of	Retention of young	Stable	Hazelnut value chain generating job	Mechanization, value chain activities
	the area	people in the area		opportunities, attracting young people	
				that would have out-migrated otherwise	

Participatory assessment of current sustainability and resilience - Method and application

2.3.4 Resilience strategies for the farming system

Overall, perceptions on implementation levels of strategies was scored most positive in IT-Hazelnut (Figure 2.3A-C). In IT-Hazelnut, the strategies of establishing cooperatives and starting new value chain activities were perceived to be least well implemented. In BE-Dairy and NL-Starch potato implementation levels of different strategies were scored poor to good. In BE-Dairy, strategies related to the carbon footprint were less well implemented than for the indicators 'real milk price' and 'labour income'. In NL-Starch potato, strategies related to soil quality were less well implemented than for profit per hectare. (Figure 2.3A-C)

Strategies (Table 2.6) could mostly be related to the resilience attributes "Reasonably profitable", "Infrastructure for innovation", "Production coupled with local and natural capital", "Socially self-organized" and "Functional diversity". Strategies linked to "Socially self-organized" were perceived to be well implemented in all case studies. Strategies related to innovation were perceived to be very well implemented in IT-Hazelnut, and moderately in the other case studies. Only in BE-Dairy and NL-Starch potato several strategies could be evaluated as contributing to functional diversity and the coupling of production with local and natural capital. In the case studies there were no strategies identified that could be linked to redundancy of farms or to policies (SM2.3).

In general, perceived contribution of strategies to robustness was moderate. Contribution to adaptability was generally equal or lower, with a few exceptions. Contribution to transformability was considered positive as well as negative. When positive, contribution was equal or lower than moderate, with a few exceptions. When negative, contribution was considered weak and sometimes moderate. (Figure 2.3D-F)

In IT-Hazelnut, the perceived contribution of RDP funds was controversial, where participants scored negative as well as positive. Participants indicated that the RDP funds were de facto used as subsidies by farmers, without changing farming practices in the long-term. Dependent on the indicator that was considered, mechanization was sometimes seen as being negative for transformability, but overall positive. Cooperatives, producer organizations and value chain activities were all perceived to contribute to farming system resilience. In BE-Dairy, all strategies were perceived to contribute positively to robustness, and to a lesser extent to adaptability and transformability. Moreover, interventions from outside this farming system, such as exceptional financial support from the sectoral federation (Fedis-support), genetic improvement and creation of milk powder stocks, as well as strategies that require on-farm investments, were perceived to negatively affect transformability. In NL-Starch potato, many strategies were perceived to contribute positively to robustness and adaptability. There were four strategies that were evaluated to weakly affect transformability in a negative way: scaling (of area and hence the production), increase value of starch products, have land available outside contract farming, apply precision agriculture. In NL-Starch potato, strategies related to soil quality and potato production were perceived to be good for transformability. With regard to strategies related to profit per hectare, only cost reduction, better varieties and improved knowledge on soil and varieties were perceived to be good for transformability. However, strategies that require investments from mainly within the farming system, such as scaling, increased value of starch products and adopting precision agriculture were regarded as negatively affecting transformability.





Figure 2.3. Perceived level of implementation of resilience enhancing strategies (A-C) and their contribution to the resilience capacities (D-F) per CS per studied indicator. Level of implementation was assessed on a scale from 1 to 5 and contribution to the resilience capacities was assessed on scale from -3 to 3 (Table 2.3).

2.3.5 Resilience attributes

As shown in Figure 2.4A, perceptions about presence of most resilience attributes followed similar patterns in the three case studies. Presence of resilience attributes was perceived to be low to moderate, with some exceptions, especially in IT-Hazelnut. For example, BE-Dairy and NL-Starch potato scored "Reasonably profitable" as low, while this attribute scored high in IT-Hazelnut, providing it with financial capital reserves. "Socially self-organized" scored high in IT-Hazelnut and moderate in other case studies, which has an effect on the social capital reserves. In IT-Hazelnut, the coupling of production to local and natural capital was perceived low, while it was perceived to score moderately in the two other case studies. Perceptions on legislation being coupled to local and natural capital were low in all three case studies, especially in NL-Starch potato. The studied farming systems were perceived to have a poor to moderate degree of openness. Especially "Exposed to disturbance" was evaluated lower, as participants perceived that disturbances are threatening system functioning instead of making a system more resilient. Diversity was evaluated to be poorly present in all case studies, with the exception of "Spatial and temporal heterogeneity of farm types", which was assessed to be moderately present. Modularity in the farming system was perceived to be moderately present in IT-Hazelnut and weakly present in BE-Dairy and NL-Starch potato.

Figure 2.4B reveals that the potential contribution of resilience attributes to robustness was perceived to be very weakly to moderately positive in all three case studies. High scoring attributes in all three case studies related to the profitability of the system and its production being coupled with local and natural resources. Being exposed to disturbance was evaluated negatively as well as positively by stakeholders in BE-Dairy and NL-Starch potato, respectively, explaining the low overall score. Scores for robustness were specifically low for some resilience attributes in NL-Starch potato, namely "Exposed to disturbance", "Optimally redundant (farms)", "Supports rural life" and "Legislation coupled with local and natural capital".

Scores for contribution of resilience attributes to adaptability were similar or lower compared to contributions to robustness (Figures 2.4B and 2.4C). An exception was "Infrastructure for innovation", which received similar scores for adaptability and for robustness in all three case studies. Other resilience attributes scoring relatively high in all three case studies related to the profitability, production being coupled with local and natural capital, response diversity and diverse policies.

Finally, contribution of resilience attributes to transformability was assessed to be very weak to moderate (Figure 2.4D). Scoring patterns for transformability deviated from the patterns as observed for robustness and adaptability. In BE-Dairy, expectations were lower for resilience attributes contributing to transformability than to adaptability. In all three case studies, "Infrastructure for innovation" got relative high scores for contributing to transformability compared to other resilience attributes.



Figure 2.4. Perceived level of presence of resilience attributes (A) and their perceived contributions to robustness (B), adaptability (C) and transformability (D). Level of presence was assessed on a scale from 1 to 5 and contribution to robustness, adaptability and transformability was assessed on a scale from -3 to 3 (Table 2.3).

2.4 Discussion

2.4.1 Farming system sustainability assessment

The studied intensive, specialized farming systems are perceived to primarily provide economic viability, to provide food and, in NL-Starch potato and BE-Dairy, to maintain natural resources. Farmers (participating in the workshops) put more emphasis on economic viability compared to other participants, who divided importance more equally over farming system functions. This difference in perception indicates that no optimal solution exists across all stakeholder groups and that a balance between differing views needs to be found (in line with Robards et al. 2011). Detecting the difference between stakeholder groups was possible because participants assessed the same set of functions. The top-down approach of our method, dividing 100 points over eight functions, forced participants to make choices between economic, environmental and social functions and indicators. In our case studies, this revealed a lack of attention for social functions such as guality of life and attractiveness of the area, which might result in missing out on important feedbacks from the social domain. The identification of this knowledge gap is an important result of our participatory method. Mosse (1994) points out that the identification of the boundaries of local knowledge is an important, but often overlooked, goal of participatory research. Performance of functions was generally perceived to be moderate, with a few low and well performing functions in each case study. Contrary to perceptions on function importance, perceptions on function performance were similar between the stakeholder groups in most case studies. The remarkable variety in allocated importance of functions, together with an only moderate performance of more important functions, suggests the presence of interactions and trade-offs between functions. Further indications of trade-offs were found in the studied farming systems (SM2.4). Existence of trade-offs may influence stakeholders perceptions, which emphasizes the importance to have both information on perceptions of stakeholders as well as observational data.

2.4.2 Farming system resilience assessment

Based on perceived presence and potential contribution of resilience attributes the case studies were perceived to show more robustness than transformability, which is typical for specialized systems aiming to control external factors as much as possible (Hoekstra et al., 2018). Strategies applied in the past 20 years, mainly with regard to economic functions and food production, show that the studied farming systems mainly use their adaptability to increase robustness, e.g. increasing farm size in BE-Dairy and NL-Starch potato to better cope with small margins. It should be noted that regarding the strategies, the method is biased towards the interest of stakeholders for specific functions. This interest may also have resonated within the stakeholders' mind when reflecting on resilience attributes.

All farming systems were perceived to have a relative low presence of functional and response diversity. and contribution of diversity to resilience is perceived low. According to Hoekstra et al. (2018), the lack of diversity is an indication that production systems are operating more under a control rationale rather than a resilience rationale. Hoekstra et al. (2018) pose that for systems to optimally perform, a balance needs to be found between the control and resilience rationales. In the case studies, this balance between rationales might be partly found in the spatial heterogeneity of farm types, which relates to diversity (see also Reidsma and Ewert 2008). Heterogeneity of farm types is assessed to have relatively high presence, and especially in BE-Dairy and IT-Hazelnut this resilience attribute was perceived to contribute to resilience. Common building blocks for resilience in all case studies were profitability, production coupled with local and natural capital, social self-organization and infrastructure for innovation. Profitability was perceived as having a large potential for improving robustness and adaptability in all case studies, but was currently perceived to be low in BE-Dairy and NL-Starch potato. Higher profitability was perceived to mainly increase robustness and adaptability in BE-Dairy. To attain higher profitability in BE-Dairy, many strategies in the past required large investments, which can explain the perceived negative contribution of these strategies to transformability. Production being coupled with local and natural capital is also assessed to have a large contribution to resilience in all case studies, but was currently considered to be low in IT-Hazelnut and moderate in BE-Dairy and NL-Starch potato. Loss of natural capital such as loss of ecosystem quality might be more visible in the quickly intensifying and expanding IT-Hazelnut (Biasi and Botti, 2010). This could explain why the other case studies, which actually have more intensive systems, scored higher. Selforganization is commonly accepted as enhancing resilience (Cabell and Oelofse, 2012) However, too much connections between actors in a system can increase the risk for co-dependency and reduce modularity. This could be the case in NL-Starch potato where the transformative capacity of the local cooperative has provided a pathway towards higher profitability. However, to stay on this pathway, the cooperative maintains a high demand of starch potatoes, resulting in a very narrow rotation and an increased pressure from nematodes in the soil. Our study indicates that infrastructure for innovation is an important resilience attribute for specialized farming systems, especially for adaptability and transformability, and should receive more emphasis. Although Gunderson and Holling (2002) emphasized the importance of innovation for resilience, resilience literature often lays more emphasis on social and ecological aspects (e.g. Cabell and Oelofse, 2012).

Assessing perceived sustainability and resilience allows to reflect on both concepts simultaneously. In the studied farming systems, function indicators relating to sustainability were on average perceived to perform moderately. This suggests that adaptations or even transformations need to be realized. Without those, sustainability might further decline, especially social sustainability that currently seems to receive relatively little attention compared to economic and environmental sustainability. On the one hand further decline could lead to undesired transformational change. On the other hand current perceived lower levels of adaptability and transformability do not seem to allow orchestrated transformations by stakeholders in the farming system: apart from infrastructure for innovation, no other attribute was assessed to support transformability well in any of the three farming systems. A promising resilience attribute for transformability is related to an enabling environment for shared learning and experimentation and should be included in further assessments (SM2.5).

2.4.3 Methodology

The FoPIA-SURE-Farm framework employed in this research captures essential steps to assess farming system resilience (see Meuwissen et al. 2019). The framework encompasses multiple dimensions and perspectives by including lists with many farming system functions and resilience attributes. These lists form a base for knowledge exchange between researchers and participants. Researchers exposed participants to the notion that sustainability and resilience need to be addressed in a structured, integrated approach. Based on this approach, participants could add local meaning to the still abstract system functions and resilience attributes. Adding local meaning enabled further discussions about sustainability and resilience. For instance, for both, researchers and participants, it was informative that most important system parameters, according to participants' perceptions, could be identified and directly discussed. This allowed for identification of important system dynamics and also strategies that were aimed to maintain or enhance system functions. Selection of important system parameters can also support further analyses on systems dynamics. For instance, by zooming in on a specific set of parameters, Kinzig et al. (2006) were able to study critical thresholds of agro-ecological systems. It should be noted, however, that by first identifying main sustainability indicators, a path-dependency is created, which in the application to the three case studies resulted in an emphasis on economic and production indicators in later steps of the workshop. In general, the selection of main system parameters is also a further simplification of reality. This increases the risk of not being able to understand the dynamics of the studied system (Ouinlan et al., 2016). Related to adequately understanding farming system behaviour, it should be noted that other levels than the farming system should be taken into account as well when studying multi-level concepts such as sustainability (e.g. Van Passel and Meul 2012, Delmotte et al. 2017) and resilience (e.g. Peterson et al. 2018, Meuwissen et al. 2019). In that sense FoPIA-SURE-Farm needs to be complemented with analyses at farm level (Spiegel et al., 2019) and the level beyond the farming system (Feindt et al., 2019). The level beyond the farming system is not well defined in agricultural literature as there are multiple possibilities, e.g. entire value chains, food systems, and the political or socio-technical environment.

The selected case studies were different in geographical size and had different positions in governmental hierarchy. On the one hand this shows the wide applicability of FoPIA-SURE-Farm to handle various farming systems with different geographical and political boundaries. On the other hand it made the comparison between the three case studies in this paper challenging, i.e. differences observed could be confounded with size and hierarchical level of the case study. Differences could also be confounded with the influence

of the individual research teams in each case study. Also culturally defined inclinations, e.g. towards optimism or pessimism, could play a role. To deal with these challenges in this paper, we did not compare scores between case studies directly, e.g. stating that a function was performing better in one case study compared to another, or by performing statistical tests on differences between case studies. Instead, we treated scores as being relative to other scores within the same case study. This revealed certain patterns such as the relative absence of attention for social functions, the emphasis on robustness and the relative importance of the resilience attributes related to profitability, social self-organization and infrastructure for innovation in all case studies.

Summarizing weighted indicator scores into sustainability indices is a common practice in sustainability science (Mayer, 2008), and for instance applied in the original FoPIA-approach (König et al., 2013; Morris et al., 2011). However, letting participants divide 100 points over functions and indicators is less commonly applied. We argue that this method helps to raise awareness of trade-offs between functions, in case such trade-offs are until then only implicit or part of a subconscious process. Dividing 100 points over functions and indicators is less commonly apricipants are forced to allocate scores to a number of items, while following a predefined distribution in which extreme values are more rare than moderate values. In the approach taken in this study, the participants themselves effectively determine their own distribution. This makes that an imbalance between function or indicator importance can be interpreted as an outcome of the study rather than a design input as is the case for the Q-methodology.

Using Likert items and scales poses another challenge for interpreting the results from FoPIA-SURE-Farm. In this paper, the performance of functions can be seen as Likert scales, where the representative indicators are weighted Likert items, allowing for presenting the mean as summarizing statistic (Guerra et al., 2016). Regarding the scoring of strategies and resilience attributes, we chose to present the mean as well, which is not entirely correct according to some, but acceptable to others (e.g. Norman 2010) and more intuitive compared to using the median an quartiles in communicating results (Guerra et al., 2016). This is especially true for the strategies where the number of observations is low. SM2.2 provides means as well as individual observations, which shows that means and medians, basically the observations in the middle of the data, do not differ much. Moreover, for testing significant differences, non-parametric tests were used that correspond with the ordinal nature of the data. It can be argued that, although participants may have different points of reference regarding whether a function for instance is performing poorly or perfectly, perceptions of performance between stakeholders and across indicators can be compared (e.g. Morris et al. 2011, König et al. 2013). The possibility that participants have avoided extreme values on the provided scoring scales, might be reflected in in the moderate scorings that many functions and resilience attributes received. This is an additional reason to look at differences between scores and consequently focus on the patterns of higher and lower scoring items, as is done in this study. This is somewhat similar to the analyses on the outcomes of Q-methodology, where patterns of extreme values for a set of specific items can be interpreted as being expressions of mental models of stakeholder in a system (McKeown and Thomas, 2013). A final point of attention relates to the notion that negative and positive values are for various reasons not true opposites of each other, possibly leading to a method bias (Alexandrov, 2010). In our study, all Likert-type items were phrased in a positive way, thus reducing the impact of a possible method effect (Alexandrov, 2010). Practically, this reduces the likelihood that for instance the positive scoring of resilience attributes is partly a methodological artefact.

Another point of discussion is that participants showed signs of fatigue towards the end of the FoPIA-SURE-Farm workshop. This coincided with the intellectually challenging scoring exercise on presence of resilience attributes and their contribution to robustness, adaptability and transformability. Still, this exercise was completed correctly in six out of eight other European case studies (Paas et al. 2019). Dependent on the research question, scoring on presence and contribution could be combined into a score that summarizes the overall importance of the resilience attribute for the farming system. Especially filling out forms was experienced as tedious. Moreover this method was top-down, which is advised to be avoided in participatory approaches that deal with the topic of resilience (Callo-Concha and Ewert, 2014). In FoPIA-SURE-Farm we could not avoid top-down questions in order to save time of participants, to enhance comparability between case studies, and to identify knowledge gaps. However, we stimulated participants to influence the content of the workshop, e.g. by providing feedback during the plenary and small group discussions that alternated the individual exercises. To further compensate for top-down questions, and to make sure that the right issues are addressed, stakeholders should be consulted again when main indicators of the farming system as identified in this study are used for further analyses. For instance when moving to the planning phase of a project cycle (Ridder and Pahl-Wostl, 2005) where concrete strategies for improved sustainability and resilience have been identified (Chapter 5; Paas et al. 2020).

With the FoPIA-SURE-Farm framework, the underlying system mechanisms that bring current resilience were only revealed occasionally in plenary discussions, rather than being a fundamental part of the framework. For instance, interactions between resilience attributes through competition for resources or co-dependence was not addressed. To complete the resilience assessment and understand underlying mechanisms, further research is necessary that includes the impact of (new) shocks, adaptation measures and future scenarios (Walker et al., 2002). For that reason we continued within the SURE-Farm project with participatory integrated assessments in which we assess performance, interactions and thresholds of important system parameters in different possible futures (Chapters 5 & 6; Paas et al. 2020).

2.5 Conclusion

The framework presented in this paper is based on existing sustainability and resilience frameworks. It provides a method to identify main indicators of a farming system and to obtain a qualitative assessment of its perceived sustainability and resilience, based on opinions of stakeholders from that system. This reveals stakeholder perspectives on importance and performance of functions and resilience attributes accounting for the complex nature of farming systems. Perspectives on importance were sometimes imbalanced, i.e. too little importance was allocated to social and environmental functions. Also attention for resilience attributes was imbalanced. The identification of imbalance is an important outcome of the method, as it indicates the boundaries of local perspectives and knowledge. Perspectives on performance were sometimes ever sometimes deviating from findings presented in literature, which emphasizes the need to have input from quantitative analytical sources as well.

Assessing perceived sustainability and resilience simultaneously allows to reflect on pathways to higher sustainability. Taking the case of specialized systems in the EU, workshop outcomes suggest that function performance relating to sustainability was perceived to be moderate, while presence of resilience attributes was perceived to be low to moderate and contribution of these attributes to resilience was perceived to be weak to moderate. In the studied systems, robustness was perceived to be stronger than adaptability and transformability. This indicates that finding pathways to more sustainability, which requires adaptability and transformability, will be a challenging process.

Strategies to maintain performance of indicators of the studied systems were mainly related to keeping the system economically viable, partly through innovations in the system. Across case studies, profitability, production coupled with local and natural capital, infrastructure for innovation and self-organization were perceived as important resilience attributes. Based on workshop results, we conclude that an additional resilience attribute related to an enabling environment for experimentation and learning is necessary. The relative importance and contribution of some resilience attributes in the studied systems differed from case to case, e.g. heterogeneity of farm types. This indicates that the local context in general, and stakeholder perspectives in particular, are important when evaluating general resilience and policy options based on resilience attributes.

Overall, despite some methodological limitations, the case study specific results seem a good starting point for raising awareness, further assessments, and eventually for developing a shared vision and action plan for improving sustainability and resilience of a farming system.

Supplementary Materials

Supplementary materials 2.1: <u>https://www.ecologyandsociety.org/vol26/iss2/art2/appendix2.pdf</u>
Supplementary materials 2.2: <u>https://www.ecologyandsociety.org/vol26/iss2/art2/appendix3.pdf</u>
Supplementary materials 2.3: <u>https://www.ecologyandsociety.org/vol26/iss2/art2/appendix4.pdf</u>
Supplementary materials 2.4: <u>https://www.ecologyandsociety.org/vol26/iss2/art2/appendix5.pdf</u>
Supplementary materials 2.5: <u>https://www.ecologyandsociety.org/vol26/iss2/art2/appendix6.pdf</u>

How do Stakeholders Perceive the Sustainability and Resilience of European Farming Systems?

This chapter has been published as:

Reidsma, P., Meuwissen, M.P.M., Accatino, F., Appel, F., Bardaji, I., Coopmans, I., Gavrilescu, C., Heinrich, F., Krupin, V., Manevska-Tasevska, G., Peneva, M., Rommel, J., Severini, S., Soriano, B., Urquhart, J., Zawalinska, K., Paas, W., 2020. How do stakeholders perceive the sustainability and resilience of EU farming systems? Eurochoices 19: 18–27. <u>https://doi.org/10.1111/1746-692X.12280</u>

Abstract

An increasing variety of stresses and shocks provides challenges and opportunities for European farming systems. This article presents findings of a participatory assessment on the sustainability and resilience of eleven European farming systems, to inform the design of adequate and relevant strategies and policies. According to stakeholders that participated in workshops, the main functions of farming systems are related to food production, economic viability and maintenance of natural resources. Performance of farming systems assessed with regard to these and five other functions was perceived to be moderate. Past strategies were often geared towards making the system more profitable, and to a lesser extent towards coupling production with local and natural resources, social self-organisation, enhancing functional diversity, and facilitating infrastructure for innovation. Overall, the resilience of the studied farming systems was perceived as low to moderate, with robustness and adaptability often dominant over transformability. To allow for transformability, being reasonably profitable and having access to infrastructure for innovation were viewed as essential. To improve sustainability and resilience of European farming systems, responses to short-term processes should better consider long-term processes. Technological innovation is required, but it should be accompanied with structural, social, agro-ecological and institutional changes.

3.1 Sustainability and resilience of European farming systems

With further liberalization of markets, a changing policy context and climate change, agriculture in Europe is increasingly subject to a variety of stresses and shocks. These disturbances provide challenges and opportunities for farming systems and affect their ability to deliver private and public goods. The recent COVID-19 outbreak provides an additional challenge. Farming systems in Europe vary widely in terms of characteristics, production, actors involved and challenges faced. Dependent on the context, they function differently and show different degrees of sustainability and resilience, two complementary concepts (see SM 1.1, for definitions and concepts). Sustainability can be defined as an adequate performance of all system functions across the environmental, economic and social domains (Morris et al., 2011). We define resilience of a farming system as its ability to ensure the provision of the system functions in the face of increasingly complex and accumulating economic, social, environmental and institutional shocks and stresses, through capacities of robustness, adaptability and transformability (Meuwissen et al., 2019). A proper understanding of the local context and underlying mechanisms of resilience is essential for designing adequate and relevant strategies and policies (Biesbroek et al., 2017). In the case of European farming systems, these strategies and policies should help to improve the system functions to 1) deliver healthy and affordable food products, 2) deliver other bio-based resources for the processing sector, 3) ensure a reasonable livelihood for people involved in farming, 4) improve quality of life in farming areas by providing employment and decent working conditions, 5) maintain natural resources in good condition, 6) protect biodiversity of habitats, genes and species, 7) ensure that rural areas are attractive places for residence and tourism with a balanced social structure, and 8) ensure animal health and welfare (Figure 3.1.). Not every farming system needs high performance levels on all functions and attributes that support those functions. Stakeholders can provide insights in requirements of particular farming systems and indicate where adjustments in systems and policy incentives are needed.

Hence, in this article, we assess stakeholder perceptions regarding sustainability and resilience across European farming systems (see Section 1.6. for case studies and Chapter 2 for methods), focusing on:

- the importance and performance of the farming system functions, which we interpreted as aspects that determine sustainability levels,
- resilience-enhancing strategies based on historical dynamics, and perceived contribution to robustness, adaptability, and transformability,
- presence of attributes that enhance resilience and their perceived contribution to robustness, adaptability, and transformability.

This leads to conclusions regarding overall perceived resilience of the farming systems and policy implications.



44



Figure 3.1. Framework to assess resilience of farming systems (adapted from Meuwissen et al., 2019), with the three boxes under step 3, 4 and 5 indicating assessments presented in this article. The complete description of the functions is provided in the first section of the text, in the same order. Explanation statements of resilience attributes are provided in Table 2.5.

3.2 Farming system functions

According to stakeholders, the main functions of the studied farming systems related to food production, economic viability and the maintenance of natural resources (Figure 3.2). Most studied farming systems were perceived to perform moderately for most functions, indicating moderate levels of economic, social and environmental sustainability. Often there was cause for concern for at least one function with low performance. For example, the attractiveness of the area scored relatively low in terms of performance across case studies.



Figure 3.2. Relative importance (size of bubbles; 100 points could be distributed among 8 functions) and performance of functions of all case studies. Performance scores are from 1 to 5, where 1 is very low, 2 is low, 3 is moderate, 4 is good and 5 is very good performance. Performance is presented on both axes to allow comparison of case studies and/or functions.

Clearly, there were differences in perceived function performance among farming systems. The level of food production was considered moderate to high in all case studies, except ES-Sheep and PL-Horticulture (see Section 1.6. for country codes). In the latter countries, also the performance of farming systems with regard to other functions was perceived as low. This related to recent policy changes (decoupled payments in ES-Sheep and accession to the EU in PL-Horticulture) affecting economic viability (less net subsidies, as farmers have to rent land to keep the payment rights in ES-Sheep and lower product prices in PL-Horticulture) and consequently other functions. Performance of private functions including economic viability was particularly high in IT-Hazelnut and RO-Mixed. The hazelnut production system in IT-Hazelnut was very profitable and expanding, while in RO-Mixed the presence of EU subsidies, of various selling channels and a large agricultural employment drove the positive assessment.

In RO-Mixed the high performance of private functions was accompanied by a high performance of public functions. While there had been some decline in environmental sustainability in the past, the accession of

Romania to the EU and cross-compliance policies have increased awareness to maintain natural resources and biodiversity. In IT-Hazelnut on the other hand, the expansion of monoculture hazelnut production caused environmental concerns (Nera et al., 2020). The increased awareness for environmental sustainability observed in RO-Mixed after accession to the EU was less visible and not explicitly mentioned in BG-Arable, where larger farms dominate. Perceived performance of public functions was on average moderate, but perceptions differed per stakeholder type. Specifically in more intensive or intensifying systems (e.g., NL-Arable, BE-Dairy, BG-Arable), farmers perceived the performance as better compared to other stakeholders. Perceived performance of public functions was particularly high in FR-Beef, and generally scored higher for livestock systems compared to arable, horticultural and perennial systems.

Interestingly, functions that represent the social domain (quality of life, attractiveness of the area) were not given much importance. These functions were considered to perform low to moderately in most systems. For the studied farming systems, we found an imbalance in the importance given to the economic, environmental and social domains. This imbalance could be caused by more or less conscious trade-offs encountered by farming system actors, who, facing direct, immediate challenges in the economic and environmental domains, might pay less attention to the social domain.

3.3 Resilience-enhancing strategies

Strategies applied in the past 20 years suggest that farming systems were generally resilient, but mainly in terms of robustness. Still, participants in workshops often perceived positive contributions of past strategies to adaptability and transformability.

In all case studies, the most frequently mentioned strategies related to reducing costs, technology implementation, and increasing farm size, in order to increase production and/or cost efficiency and make the farming system "reasonably profitable" (Table 3.1). These strategies were emphasized in BE-Dairy, BG-Arable and ES-Sheep. In BE-Dairy, these strategies were seen as enhancing robustness while constraining transformability. This was explained by the relative high investment costs which cause a lock-in on the pathway to higher efficiency. In ES-Sheep and BG-Arable, these strategies were considered to enhance adaptability, but also transformability. Specifically, in a more extensive system like ES-Sheep increasing efficiency could also improve transformability, as increased efficiency was needed for the reorganization of the system.

Strategies that have coupled production with local and natural capital (e.g., manure recycling in BE-Dairy) were mentioned quite frequently, but less than half as often as the ones above. Stakeholders perceived the contribution to robustness, but less to adaptability and transformability. As an example, agrienvironmental schemes in UK-Arable tend to tie farmers into fixed approaches: new schemes need to be flexible to allow farmers to react to external stresses.

Only in a few case studies stakeholders mentioned that strategies related to diversification (at different levels) had been applied in the past (BE-Dairy, ES-Sheep, UK-Arable, PL-Horticulture). Such strategies were perceived to have high and (relatively) balanced contributions to all three resilience capacities, although in PL-Horticulture participants saw a negative effect in the short-term.

			Contrib	ution of	linked		Contri	ibution o	f
			strateg	ies to*			attrib	ute to ^l	
Resilience attribute	number of strategies linked to attribute [†]	Implementation level of strategies [‡]	robustness	adaptability	transformability	Presence [§]	robustness	adaptability	transformability
Reasonably profitable	54	3.4	1.5	1.5	0.3	2.2	1.8	1.5	0.9
Coupled with local and natural capital (production)	22	2.5	1.4	1.0	0.7	2.9	1.8	1.6	1.1
Functional diversity	15	2.6	1.7	1.7	1.2	2.2	1.4	1.3	1.2
Response diversity	8	3.1	2.1	1.5	0.7	2.4	1.3	1.4	1.0
Exposed to disturbance	4	3.8	2.6	1.2	0.3	2.5	0.6	0.7	0.3
Spatial and temporal heterogeneity (farm types)	3	2.2	1.8	1.9	1.9	3.1	1.5	1.4	1.2
Optimally redundant (farms)	1	n.a.	n.a.	n.a.	n.a.	2.5	1.0	1.1	0.9
Supports rural life	12	4.5	2.3	1.0	0.5	2.6	1.2	1.1	0.8
Socially self-organized	21	2.81	1.9	1.9	1.0	3.0	1.6	1.6	1.2
Appropriately connected with actors outside the farming system	5	4.2	1.7	1.8	1.3	2.3	1.1	1.0	0.8
Coupled with local and natural capital (legislation)	12	3.9	1.1	1.5	1.2	2.8	0.7	0.6	0.4
Infrastructure for innovation	13	3.5	1.7	1.8	0.3	2.1	1.6	1.6	1.7
Diverse policies	3	2.6	1.0	1.2	1.2	2.1	1.2	1.1	0.9

Table 3.1. Resilience attributes and results from the assessment.

[†] Past strategies to cope with challenges are linked to attributes (one strategy can be linked to multiple attributes). Attributes with more linkages are coloured in darker blue.

⁺ Implementation level of strategies was scored from 1-5 and averaged across case studies, with 2-2.5 in light blue (poor), 2.5-3 in blue (towards moderate), 3-3.5 in dark blue (above moderate), 3.5-4 in darker blue (towards good) and 4-5 in darkest blue (good to very good).

* The perceived contribution of strategies to resilience capacities was averaged across strategies, with yellow representing a score between 0-1 (very weak to weak), light green 1-1.5 (weak), green 1.5-2 (towards moderate), and dark green 2-3 (moderate to strong). In FR-Beef, strategies were not scored.

[§] The perceived presence of attributes was averaged across case studies, with 2-2.5 in light blue (poor), 2.5-3 in blue (towards moderate) and 3-3.5 in dark blue (above moderate).

¹ The average assessed contribution of attributes to capacities, with a similar colour scheme as for strategies. In FR-Beef and ES-Sheep, attributes were not scored.

In seven out of 11 case studies stakeholders indicated that strategies related to the organizational forms of farming system actors (social self-organization; e.g., cooperatives) were applied; mainly to improve the production and economic functions. These strategies were perceived to enhance robustness and adaptability, while the contribution to transformability differed depending on the strategy and case study. For example, cooperatives and producer organizations in IT-Hazelnut were perceived to have a moderate to strong positive contribution to transformability, while vertical cooperation was perceived to have a strongly negative contribution in PL-Horticulture.

Six case studies emphasized strategies that focus on infrastructure for innovation (e.g., mechanization, improved varieties). Similar to strategies that aimed to improve profitability, these were often seen to enhance robustness and adaptability, but when investment costs were large, to constrain transformability. One of the few transformability-enhancing strategies was to extend knowledge on soil and varieties in NL-Arable, i.e. a strategy that also addressed public functions instead of only food production and economic viability.

3.4 Resilience attributes

In general, resilience attributes were perceived to be weakly to moderately present in the case studies (Table 3.1; see Paas et al. (2019), for details). IT-Hazelnut and SE-Poultry were an exception with multiple resilience attributes that were assessed to have a moderate to good presence. PL-Horticulture often received the lowest scores. "Diverse policies" was scored low in all case studies. Except for IT-Hazelnut, "reasonably profitable" was also assessed to have a low to very low presence. Surprisingly, the mixed systems RO-Mixed and DE-Arable&Mixed did not score higher for perceived presence of "functional diversity" compared to other case studies. However, for "response diversity", RO-Mixed and DE-Arable&Mixed were among the higher scoring case studies. "Spatial and temporal heterogeneity of farms" (similar to previous attributes related to diversity) scored relatively high in all case studies, compared to other attributes. The related attribute "optimal redundancy of farms", however, scored lower. The low score for NL-Arable was mainly related to the difficulties for potential farm successors to take over the farm. In ES-Sheep, participants emphasised that each farmer dropping out created a problem for the system, indicating that redundancy was low.

Although scores for the contribution of resilience attributes to resilience capacities were generally positive (Table 3.1) – which is in line with previous research – they are generally low. The contribution to robustness was generally considered higher than to adaptability and transformability. Sometimes trade-offs between resilience capacities were observed. For example, in some case studies the attribute "reasonably profitable" was perceived to be the most important for robustness, while it was seen as a negative contributor to transformability. The reasoning was that as long as economic returns were above a certain level, other incentives needed to be very convincing before change would be considered. However, in some cases the contribution to both robustness and transformability scored high, as profitability was seen as essential for building up economic reserves that could help to support system transformations.

On average, "production coupled to local and natural capital", "reasonably profitable", "socially selforganized" and "infrastructure for innovation" were seen as contributing mostly to robustness and adaptability, while "infrastructure for innovation" was seen as specifically important for transformability. Hence, while strategies implemented in the past related to "infrastructure for innovation" (e.g. mechanization in IT-Hazelnut and investment in buildings and technology in SE-Poultry) were seen as constraining transformability, stakeholders perceive the role of infrastructure that facilitates diffusion of knowledge and adoption of cutting-edge technologies as important to allow for transformability.

Participants concluded that "diverse policies" that equally aim at robustness, adaptability as well as transformability would contribute weakly to all three capacities. As strategies in the past mainly contributed to robustness, specific policies will be needed to improve adaptability and transformability.

3.5 Overall resilience

Taking into account the assessed contribution of resilience attributes to each capacity, resilience was considered to be low. The arable systems and the horticulture system were among the lower-scoring case studies regarding resilience attributes. SE-Poultry and IT-Hazelnut scored higher. In most case studies, robustness was perceived to be higher than adaptability and transformability. In case studies with immediate challenges, however, the capacity to adapt and transform was more similar to the capacity to remain robust. In UK-Arable, it was specifically argued that adaptability and transformability of the farming system were likely to be essential in the future, driven by new agricultural policies (focusing on environmental land management) and future trade deals, following the UK's exit from the EU.

The relatively high presence of robustness was also observed when assessing historical dynamics of main indicators and strategies applied to deal with challenges (Table 3.1). Despite fluctuations and some declining functions, most functions were seen as still being viable. However, when evaluating results from a variety of methods used in SURE-Farm (Reidsma et al., 2019b), it was concluded that in all case studies adaptation, and in some transformation, is required. For many of the intensive or intensifying production systems (e.g., NL-Arable, UK-Arable, BG-Arable, SE-Poultry, BE-Dairy), strategies to increase efficiency are losing their positive impact on private functions, while having accumulated negative impacts on public functions. This implies that alternative strategies are needed to improve sustainability and resilience. On the other hand, more extensive small-scale systems (e.g., RO-Mixed, ES-Sheep) may still benefit from increased efficiency to improve both private and public functions. In IT-Hazelnut and PL-Horticulture there is still room for growth, mainly because of expanding markets for hazelnuts and fruits, but also here the delivery of public goods is a concern.

3.6 Policy implications

The observed preference for functions related to food production and economic viability resulted in a tradeoff with functions related to the environment and society. Many stakeholders were primarily concerned with the more immediate stress signals from the faster processes in the farming system (e.g. year-to-year variation of income or production levels) compared to slower processes (e.g. development of soil quality and social well-being of the population in the farming system). This preference induces myopia among farming system actors, at least as long as performance in the environmental and social domain is considered to be acceptable. In other words, when biodiversity decline is not clearly visible and social infrastructure is still available, little action is taken.

Options to shift focus towards slower processes consist of policies that reduce stress signals from faster processes (e.g. through insurance), while improving noticeability of feedback signals from the slower processes (e.g. monitoring adapted to the scale of the process, of for instance biodiversity). Improving noticeability of these signals will help to assess and communicate long-term impacts of developments in farming systems.

Moreover, policies should be designed to safeguard the presence of resilience-enhancing attributes, especially in the light of ongoing trends of intensification and scale-enlargement that could diminish these attributes in European farming systems. Currently, attributes that mostly contribute to all three resilience capacities relate to having appropriate infrastructure for innovation, self-organization of actors in the farming system, the coupling of agricultural production with local and natural capital, and different aspects of diversity. Concluding, technological innovation is required to enhance sustainability and resilience, but should be accompanied with structural, social, agro-ecological and institutional changes (see also Mann 2019). Farmers can change, but they cannot do it alone.

Supplementary Materials

The dataset used in this study can be downloaded at https://zenodo.org/record/5005175#.Ym-q8dpByUk

Temporal and inter-farm variability of economic and environmental farm performance: a resilience perspective on potato producing regions in the Netherlands?

This chapter will be submitted as:

Paas, W., Meuwissen, M.P.M., van Ittersum, M.K., Reidsma, P. Temporal and inter-farm variability of economic and environmental farm performance: a resilience perspective on potato producing regions in the Netherlands?

Abstract

In the context of resilience and sustainability of farming systems it is important to study the trade-offs and synergies between economic and environmental variables. In this study, we selected food production, economic and environmental performance indicators of farms in three potato producing regions in the Netherlands: Flevoland, Zeeland and Veenkoloniën. We studied the period 2006 to 2019 using farm accountancy data. We used threshold regressions to determine gradual development and year-to-year variation of those indicators. Subsequently we applied a sparse Partial Least Square (sPLS) regression to study the response of performance, gradual development and year-to-year variation under different conditions regarding weather, market and farm structure. sPLS-model performance was at best moderate. Models could explain most of variability of the data in Veenkoloniën, a region with relatively little interfarm variability and relatively stable economic prices. Model results were very sensitive for the selection of response variables. We found that food production, economic and environmental performance levels and gradual developments were primarily determined by intensity levels. How these levels are determined by intensity, i.e. positively or negatively, differed per case study. Year-to-year variability was determined by average yearly weather conditions and weather extremes. Overall, we conclude that results do only provide insights that confirm existing knowledge at case study level, sPLS can be seen as a filter and projector of high-dimensional data that accentuates patterns in the data. In the context of resilience of farms, while using a relatively small dataset, our methodology seems limited to a rather homogeneous farm population in a stable economic environment. Researchers intending to apply this method to (arable) farming systems should be well aware of the influence they can have on the results through their selection of response variables.

4.1 Introduction

In an increasingly variable climatic and socio-economic context, a sustainable and resilient performance of farming systems is challenging (Meuwissen et al., 2019). Sustainable performance is important regarding the provision of system functions in the long-term, while resilient performance is important to maintain function performance in the face of disturbances in the short-term (Tendall et al., 2015). Sustainability and resilience of a farming system is dependent on a balanced performance regarding social, economic and environmental functions (Walker and Salt 2012, Chapter 5). However, trade-offs between those functions are common in farming systems (Kanter et al., 2018; Klapwijk et al., 2014), thus destabilizing the base for sustainability and resilience. One can imagine that these trade-offs only become sharper when faced with disturbances that require an immediate response.

Resilience and sustainability of farming systems are complementary concepts that need to be studied simultaneously in integrated assessments (Meuwissen et al. 2019, Chapters 2 & 5). Many theoretical and qualitative studies have suggested attributes that increase resilience and sustainability (e.g. Cabell and Oelofse, 2012; Resilience Alliance, 2010). For example, diversity is often suggested to increase both. However, few studies have quantitatively studied resilience of farming systems (Dardonville et al., 2021), and even fewer address both sustainability *and* resilience indicators. Hence, assessing the performance in terms of both types of indicators, quantitatively, is the focus of this paper.

Existing agro-econometric methods often use production functions to assess resource allocation efficiency and thus assess sustainability. The general notion behind these methods is that increased agronomic and economic efficiency, and thus sustainability, of farming systems can be achieved by increased output efficiency of individual farms. Silva et al. (2021), for instance use a yield gap analyses based on an approach that combines concepts from econometrics and production-ecology (van Dijk et al., 2017; Van Ittersum and Rabbinge, 1997). Production functions require specific input regarding the shape of functions and are primarily developed for evaluating a single good, e.g. (food) production or economic output. Reidsma et al. (2009) include trans-log distance functions and, interestingly, consider multiple response variables simultaneously. Other, purely econometric methods are geared towards assessing the potential for increasing production or economic performance, such as the Just-Pope production function (Just and Pope, 1978) and damage abatement functions (Hall and Norgaard, 1973). These methods do not include environmental response variables, which makes these less useful for an integrated sustainability study. In addition, these methods usually employed datasets with a limited number of years.

As to resilience, a concept relating to the dynamics of the system, the element of time becomes more important, requiring longitudinal data. There are few studies on the quantitative analysis of resilience in general and in particular studies using longitudinal data are rare (Dardonville et al., 2021). In absence of longitudinal data, cross-sectional data may be used in which the performance or resilience of regions is evaluated relative to one another (e.g. Abson et al., 2013; Reidsma et al., 2007; Van Passel et al., 2017). Also model-based methods including future scenarios may be used (e.g. Herrera et al., 2022). Longitudinal data can be used in different ways to study resilience. Reidsma et al. (2009a, 2009b) studied yield variability in relation to weather conditions and farm characteristics. More recently, Sneessens et al. (2019) proposed a framework that includes multiple variables that also include resilience concepts such as the recovery time after a shock. In another recent study, Slipper et al. (2021) used longitudinal data to study the resilience capacities in terms of robustness, adaptability and transformability for agricultural regions in 11 European countries.

In a review on quantitative resilience studies, Dardonville et al. (2021) note that environmental indicators are hardly included as response variables. Martin et al. (2017) propose a framework that allows to explore covariation of multiple explanatory and response variables over time without the need to pre-define a production function. This provides opportunities to evaluate economic as well as environmental response variables for which no production function can be defined. The framework of Martin et al. (2017) has been applied to livestock systems (Bouttes et al., 2018; Martin et al., 2017), but to the best of our knowledge not to arable systems. In specialized livestock systems, intermediate activities, such as the growth of grass, are ultimately used to produce one or two outputs, e.g. milk (Bouttes et al., 2018) and/or meat. In arable farms, the cultivation of multiple crops are parallel activities with parallel outputs, i.e. the output is

usually not concentrated in one or two outputs. As a consequence, variability of output at farm level may play out differently than at crop level (Mandryk et al., 2017).

In this paper we aim to study economic and environmental sustainability and resilience simultaneously and quantitatively. We apply and evaluate the framework of Martin et al. (2017) to arable systems for this purpose. We selected three different potato growing regions in the Netherlands as case studies. Employing the method we aim to identify resilience attributes at farm level, i.e. farm characteristics, that support sustainability and resilience regarding market and climatic conditions.

4.2 Methods

4.2.1 Case studies

In this study, three potato growing regions in the Netherlands are compared. Veenkoloniën (VK) is a region in the North-East of the Netherlands with sandy and peaty soils. In this region it is common to find a crop rotation with starch potato up to once in two years in combination with mainly sugar beet and cereals. Since about ten years, onion is increasingly cultivated in VK. Zeeland (ZE) in the South-West and Flevoland (FL) in the middle of the Netherlands on the contrary, have clayey soils with somewhat wider crop rotations including mainly ware potatoes, sugar beet, cereals and onions (once in three to five years is common). Common additions to crop rotations are carrots in ZE and carrots, vegetables and tulips in FL. Arable farming in VK is less profitable compared to ZE and FL and more prone to the impacts of weather variability and climate change (Diogo et al., 2017). However, due to the cooperative structure of starch potato cultivation and processing in the area, cultivated area and farm gate prices of starch potato are relatively stable compared to the ware potato FL.

4.2.2 Data

We used farm accountancy data collected by Wageningen Economic Research (WEcR) for the Farm Accountancy Data Network (FADN; Poppe, 2004; Veen et al., 2014). This data is mainly collected to study economic and environmental performance at farm level, while economic data at crop level is also available. Because of privacy regulations, individual farm data cannot be presented in this study. The data in this study include time series for the period 2006 to 2019 of seven to 14 subsequent years of potato growing arable farms from the three case study regions. The final number of individual farms per region included in the analysis was 15 (FL), 19 (VK) and 17 (ZE). (See also Table A1.1 and A1.2 in SM4.1). Weather data was retrieved from the data platform Agri4Cast (JRC of the European Commission, 2021). Market data was retrieved from different online sources (Kadaster, 2021; OECD, 2021; WEcR, 2020).

4.2.3 Variable selection

Overview

The variable selection in this paper is guided by the resilience framework of Meuwissen et al. (2019) (Figure 4.1). Meuwissen et al. (2019) propose five steps to assess the resilience of farming systems: identification of 1) the farming system, 2) challenges, 3) functions, 4) resilience capacities and 5) resilience attributes. The farming systems are described in the case study section above (Step 1). Explanatory variables related to weather and market conditions represent challenges that are hypothesized to affect the response variables (Step 2). The response variables are related to functions of farms (Step 3), e.g. food production. Resilience capacities (e.g. adaptability) are deduced based on the outcomes of this study (Step 4). Explanatory variables related to farm characteristics, e.g. farm area and crop diversity, represent resilience attributes that possibly affect response variables directly, but possibly can also moderate the impact of challenges (Step 5; Figure 4.1). Variable selection and links to the framework of Meuwissen et al. (2019) are elaborated below. All variables and abbreviations of these variables are presented in Table A2.1 in SM4.2.



Figure 4.1. Overview of variables included in the analyses (blue blocks) and their link to the steps in the resilience framework of Meuwissen et al. (2019) (grey blocks). Green blocks indicate the different analyses that are performed on the data. Orange arrows indicate the type of patterns that are studied in the sPLS regression. PCA: Principle Component Analysis, sPLS: sparse Partial Least Squares.

Response variables characterising system functions

For an integrated analysis, we included response variables that cover production, economic and environmental functions at crop, crop rotation and farm level. For the production at crop level we used the average yield of potato (tons/hectare). For the production at crop rotation level we calculated the consumable energy produced (kJ/ha; Silva et al., 2017). Energy content per crop was based on Meul et al. (2007). Energy produced was only calculated for the main crops (potatoes, sugar beets, wheat, barley, onions), resulting in a value that was on average representing more than 85, 91 and 80% of the farm area in FL, ZE and VK. Average operating profit of crops (€/ha) is taken as economic indicator at crop rotation level. At farm level, all monetary output (revenues; excluding off-farm income) per monetary input (all fixed and variable costs) represents the output efficiency of the farm (€ output/€ input). Having an indicator that expresses efficiency may help to explore possible trade-offs between efficiency and variability that are hypothesized in resilience literature, i.e. more efficient systems are more vulnerable to disturbance (e.g. Cumming and Peterson, 2017; Hoekstra et al., 2018; Resilience Alliance, 2010). The nitrogen surplus at farm level is used as an environmental indicator. Nitrogen surplus contributes to expulsion of greenhouse gases, acidification of nearby nature areas and leaching or runoff of N leading to eutrophication of water bodies. N-surplus is calculated based on a nutrient balance at farm level that includes all nitrogen inputs, outputs and stock changes (Eurostat, 2013; Lamkowsky et al., 2021; Poppe, 2004).

Table A 1	Overview	of response	variables
TUDIE 4.1.	Overview	<i>oj response</i>	vuriubies.

Type of variable	Sub-category	Abbreviation	Unit
Response variables	Output efficiency	OutputEff	€ output / €
			input
		OutputEff_resi	€ output / €
			input
		OutputEff_slope	€/€/year
	Potato yield	Potatoyield	ton / ha
		Potatoyield_resi	ton / ha
		Potatoyield_slope	ton / ha / year
	Crop rotation energy yield	Energyperha	kJ / ha
		Energyperha_resi	kJ / ha
		Energyperha_slope	kJ / ha / year
	Profit crops	ProfitCropsperha	€ / ha
		ProfitCropsperha_resi	€ / ha
		ProfitCropsperha_slope	€/ha/year
	Nitrogen surplus	Nsurplus	kg / ha
		Nsurplus_resi	kg / ha
		Nsurplus_slope	kg / ha / year

High observed levels for potato yield, crop rotation energy yield, profit of crops, output efficiency and low observed levels for N-surplus were seen as positive for sustainability. Slopes and residuals of trendlines were used as additional variables that describe the resilience of farms (Figure 4.2; Bouttes et al. (2018) and Martin et al. (2017)). Positive slopes for potato yield, crop rotation energy yield, profit of crops and output efficiency, and negative slopes for nitrogen surplus were seen as signs of adaptation towards more sustainability. Small residuals were seen as indicative for farm stability and therefore positive for farm robustness. See also Table A2.1 in SM4.2 for the response variable abbreviations.

Trendlines were fitted using one and two segmented linear regression analyses allowing for an evaluation of structural change in the observed values over time (Zeileis et al., 2002; details in SM4.3). We argue that, besides the trends themselves, structural changes in trends leading to positive or negative developments can also be seen as indicators for the presence or absence of farm adaptability.



Figure 4.2. Fitted trend lines for two imaginary farms regarding a unitless efficiency indicator. For the farm with observed values in blue, a two-segmented trendline gives a significant better fit than a single trendline. Three types of Y-variables are eventually included in the analyses: the observed values, residuals and the slope of the trend lines.

Explanatory variables linking to resilience attributes

Explanatory variables related to farm characteristics are divided into the following sub-categories; land use, input intensity, assets and management (Table 4.2; Figure 4.1). These sub-categories can be linked to resilience attributes, which are system characteristics that convey general resilience to a farming system (Chapters 1 & 2). Land use indicators can be used as a proxy for crop diversity, for instance the share of cereals or potatoes in the rotation. Diversity is generally seen as buffer against perturbations and is also considered as a source of renewal after a perturbation (Cabell and Oelofse, 2012). In the case of crop diversity, we see specialization as the inverse of crop diversity, i.e. a large cultivation area dedicated to main crops. Regarding diversity we took the fraction of cereals, the fraction of the main crops (potato, sugar beet and cereals) and the effective number of crops (also known as true diversity index). The indicators under input intensity could be seen as proxies for the degree to which the system is coupled with local and natural capital. For instance, a low input of crop protection products may suggest a better coupling of farm practices with the environment. From a resilience perspective, high dependence on external inputs (e.g. mineral fertilizer) for a high and stable production in the face of environmental fluctuations (e.g. weather, pests & diseases), may imply a lower degree of autonomy (Cabell and Oelofse, 2012). Certain European crop-livestock systems, for instance, may not be robust enough to withstand a situation in which the import of mineral nitrogen fertilizers is halted (Pinsard et al., 2021). Asset indicators relate most to system reserves that can be used in difficult times. Modernity of machines and buildings (actual value/value when new) is linked to the availability of infrastructure for innovation. However, modernity of machines and buildings could also be related to the absence of adaptability and transformability due to sunk costs, i.e. money invested that cannot easily be re-invested (Westley et al.,

2011). Management indicators, such as the number of full time equivalent (fte) managers per hectare, relate to the degree of experience and attention that is available for agricultural practices. This relates to the potential for learning from past experiences and building human capital, both being important for general resilience (Cabell and Oelofse, 2012).

Explanatory variables related to external influences are classified into the following sub-categories: market prices, average weather conditions and extreme weather events. Average market indicators per year include: oil price (\leq / 100 literWEcR, 2020), fertilizer price (\leq /kg; NPK12:10:18; WEcR, 2020), land prices (\leq /ha; (Kadaster, 2021), interest rates (%; OECD, 2021) (Table A2.2 in SM4.2). Average weather conditions included in the analysis are average temperature (degree Celsius), average daily precipitation (mm/day) and average daily precipitation deficit (mm/day) for the whole year, spring (April-June) and summer (July-September)¹ (See SM4.2 for more details). Based on the average daily precipitation and temperature, extreme weather events for potato production were calculated using the AgroClimateCalendar (ACC; Schaap et al., 2011, 2013; Table A2.1 in SM4.2). Included weather extremes were wet (and warm) conditions, heatwaves, late frosts, warm winters and drought (Table A2.3 and A2.4 in SM4.2). Descriptions of the effect of weather extremes are described in Table A2.3 in SM4.2. Throughout the observation period, weather extremes were observed in all three case studies.

 $^{^1}$ To cover the whole growing season of potatoes from planting (April) till harvesting (September), we have deviated from the meteorological definition of spring (1 March – 31 May) and summer (1 June – 31 August)

Sub-category	Sub-category	Abbreviation	Unit
(1st)	(2nd)		0.110
Farm	Land use	AreaCereals	ha
characteristics	Lund use	Alcucercuis	na
characteristics		AreaMainCrons	ha
		TrueDiversity	#
	Innut intensity	Monetary input intensity [†]	″ €/cultivated ha
	input intensity	Labour	
		Crop protection products (CPP)	£/ cultivated ha
		Nitrogen	€/ cultivated ha
		Phosphate	€/ cultivated ha
		Energy	€/ cultivated ha
		Total Costonorba [‡]	€/ cultivated ha
	Managamont	FormManagement	# fto managors /
	Management	Fairmanagement	# Ite Indiagers /
		AccEstmor	Veara
		Agei aimei OthorPoyonuo	f cultivated by
	Accoto	Area	
	Assels	Area	iid
		OwnCapital	€ OWN / € LOLAI
		MadamituDuildinga	# (0,100)
		ModernityBuildings	# (0-100)
		ModernityMachines	# (0-100)
Maathan	A		€ / cultivated ha
weather	Average	Temperature_Spring	degree Ceisius
conditions		Due sinitation Couries	and the second second
		Precipitation_Spring	mm / day
		Temperature_Summer	degree Celsius
		Precipitation_Summer	mm / day
		Temperature	degree Celsius
		Precipitation	Mm / day
	Extremes*	ExtPrec45_1 (extreme precipitation in 1	#
		day)	
		ExtPrec60_3 (extreme precipitation in 3	#
		days)	
		HWave (heat waves)	#
		Frost	#
		WarmWinter	#
		WarmWet	#
		D_Spring (drought in spring)	#
		D_Summer (drought in summer)	#
		WetHumPlant (wet and humid at planting)	#
		WetHumGrow (wet and humid in growing	#
		phase)	
		WetHumHarv (wet and humid at harvesting)	#
Market indicators§		OilPrice	€/ 100 L
		FertilizerPrice	€/ 100 ka
		LandPrice	€/ha
		Interest rate	%

Table 4.2. Overview of explanatory variables included in the analysis.

[†]Monetary input at farm level, i.e. all fixed and variable costs. [‡]Cultivation costs, i.e. variable costs for crop cultivation. *See Table A2.3 in SM4.2 for more information. [§]See Table A2.2 in SM4.2 for more information.

4.2.4 Detecting the underlying data structure

Principal component analyses

To obtain insight in the underlying data structure, correlation plots were created for response and explanatory variables. In addition, principal component analyses (PCA) were performed, separately for the response and explanatory farm variables for each case study area. Strongly correlated response variables were removed from further analyses as they can distort the results. We illustrate this potential for distortion by presenting additional model runs with highly correlated response variables (for details see SM4.3).

sparse Partial Least Squares regression

We used sparse Partial Least Squares regression (sPLS) with year and farms as random effects to study the impact of explanatory variables on the response variables. In sPLS-regressions, explanatory variables (X-variables) are projected on latent variables in such a way that the projected variables can explain as much variation of the response variables (Y-variables) that are also projected on latent variables. Latent variables represent the most dominant patterns in the data. Leave-one-out cross-validations were conducted to determine the performance of the sPLS-model. Resulting Q2 -scores were used to determine the number of latent variables (components) in the sPLS-model. Q2-scores express the marginal contribution of components to increase the covariation between the original X- and Y-Variables. A component with a Q2-score larger than 0.095 is considered to have a significant contribution. (Lê Cao et al., 2008). We performed multiple sPLS analyses in which we varied the number of Y-Variables (2-5) and X-Variables (2-9) per component. The best model was selected based on the aggregated Q2-score over all components. We compared the correlation matrix of projected values of continuous explanatory variables and response variables of the sPLS-model with the correlation matrix of the original data (for details see SM4.3).

PCA and (s)PLS were performed with the software package "mixOmics" (Cao et al., 2017) in the software environment R (R Core Team, 2015). "mixOmics" does not facilitate the inclusion of interaction terms.

4.3 Results

4.3.1 Response variables

General observations

Observed levels of potato yield and profit of crops were highest in FL (Figure 4.3). The inter-farm variability of potato yields and profit of crops within ZE and FL were much higher than for VK (Figure 4.3). Observed levels of crop rotation energy yield and nitrogen surplus were lowest in ZE. On average, observed levels of output efficiency were lowest in VK ($1.05 \in / \in$) and ZE ($1.02 \in / \in$) and highest in FL ($1.12 \in / \in$) (Figure 4.3). In ZE and VK there were multiple outliers regarding N-surpluses of more than 200 kg/ha. In the context of earlier work (Silva et al., 2021b) these values were however not surprising.

The pattern of output efficiency levels from 2006 till 2012 was similar in ZE and FL, with relatively high levels in 2006, 2010 and 2012 (Figure 4.3). In VK, output efficiency was highest in 2012, which coincided with a high potato yield. Potato yield in all regions was relatively low in the dry year of 2018. Interestingly, output efficiency and profit of crops in FL were relatively high in 2018. Nitrogen surplus levels seemed stable in all case studies. Based on a visual inspection of Figure 4.3, there were no particular years in which nitrogen surplus was deviating substantially, except for 2018 in VK when it was high, probably because of low yields due to drought.

Structural change

Breaks in trends were mostly detected in VK for the output efficiency between 2011 and 2013 (15 farms) and crop profit in the years 2011 and 2012 (12 farms) (Table A4.1 in SM4.4). This corresponds with the increase in output efficiency and profit of crops until 2012 in VK that can be observed in Figure 4.3. In FL and ZE, breaks in trend were observed for a few farms, mostly in 2012, for potato yield (FL), crop rotation yield (FL, ZE), output efficiency (ZE) and nitrogen surplus (FL, ZE) (Table A4.1 in SM4.4).


Figure 4.3. Observed levels of potato yields, crop rotation energy yield, profits from crops, output efficiency and nitrogen surplus for three regions (FL = Flevoland, ZE = Zeeland, VK = Veenkoloniën).

Yield, profit and output efficiency

Explained variance of the PCA on response variables (levels, residuals and slopes of potato yield, crop rotation yield, profit from crops, output efficiency and nitrogen surplus) per region was between 50-57% over the first three components. Important variables for the first component in all three regions (accounting for 19-29% of variation) were levels of potato yield and crop rotation yield, often accompanied with their residuals, indicating larger absolute variation at higher crop yield levels (Figure A5.1, A5.3, and A5.5 in SM4.5). In VK, profit of crops and output efficiency were positively associated with higher crop yields (Figure A5.5 in SM4.5). Potato is the largest crop in VK in terms of area and volume, partly explaining the positive relation with energy yield and profit. The positive relation between yield and profit could also be attributed to the local cooperative structure. With relatively inelastic prices for starch potato products, the cooperative benefits from larger volumes to be able to pay a good farm gate price to farmers (Herrera et al., 2022) as long as prices of processed products stay relatively inelastic. In FL, residuals of output efficiency and profit of crops and the slope of profit of crops were negatively associated with potato yield and crop rotation energy yield (Figure A5.1 in SM4.5). This suggests that farmers in FL somehow can benefit from relatively high prices when yields are relatively low. By contrast, in ZE, level of profit of crops and residuals of profit of crops and output efficiency (second component), had no or very little association with crop vield levels (Figure A5.3 in SM4.5).

Synergies and trade-offs with nitrogen surplus

In FL, the second component (17% of variation) was mostly correlated with observed levels and residuals of nitrogen surplus, associated negatively with residuals and level of profit of crops and residuals of output efficiency (Figure A5.1 in SM4.5). Overall, this suggested that years with (relatively) high nitrogen surplus coincided with (relatively) low profit of crops and low output efficiency, and vice versa. Farms associated with high levels of nitrogen surplus also showed declining profit of crops (3rd component; 14% of variation; Figure A5.2 in SM4.5).

On the first component of ZE and VK, higher crop yield levels and residuals were negatively associated with residuals of nitrogen surplus (Figure A5.3, A5.5 in SM4.5). Moreover, in ZE, on the third component (14% of variation), increasing potato yields were associated with farms that had low and decreasing nitrogen surpluses (Figure A5.4 in SM4.5). In VK, on the second component (18% of variation), decreasing nitrogen surplus was mostly correlated with increasing output efficiency, potato yield and energy production (Figure A5.5 in SM4.5). Residuals of nitrogen surplus were mostly negatively correlated with residuals of profits of crops and output efficiency (third component; 10% of variation; Figure A5.6 in SM4.5).

Pre-selection of response variables

Based on the high correlation found between response variables in the PCA and additional correlation analyses (SM4.5), we continued our analyses with crop rotation energy yield, profit of crops and nitrogen surplus (SM4.6). We performed additional analyses with all five response variables and with a different selection of three response variables (i.e. with output efficiency instead of profit of crops). These additional analyses were used to assess the impact of selecting different sets of response variables (SM4.7).

4.3.2 Explanatory variables

Farm characteristics

Explained variance of the PCA on explanatory farm variables per region was between 45-50% over the first three components (Figure A5.9-A5.14 in SM4.5). In the PCA's for the three regions, years appeared to be clustered, indicating that years explained part of the variation.

In all case studies, most of the variation (1st component; 22-27% of variation) could be related to many correlated indicators on input intensity in terms of fixed and variable costs (Figure A5.9, A5.11, A5.13 in SM4.5). Values of variables related to input intensity did increase over the years. In particular, cultivation costs increased (Table A2.1 in SM4.2). In FL and ZE, the second component was related to the area of main crops and area of cereals (Figure A5.9, A5.11 in SM4.5). Area of main crops and cereals seemed to have decreased in the observation period, suggesting decreased specialisation (Figure A5.9, A5.11, A5.13 in SM4.5; Figure 4.4). In FL more specialized farms were associated with less modern machinery, i.e. depreciated machinery (Figure A5.9 in SM4.5). In ZE, more specialized farms were associated with larger nitrogen inputs (Figure A5.11 in SM4.5). In FL and ZE, the third component was associated with larger farm sizes, lower shares of land owned, lower number of farm managers per hectare and lower labour input intensity (FL; Figure A5.10 in SM4.5) or modernity of buildings (ZE; Figure A5.12 in SM4.5).

In VK, the second largest part of the variation was captured by labour input intensity (second component; 12% of variation; Figure A5.13 in SM4.5). A third part of the variation could be explained by the number of full time equivalent managers per hectare and the age of the farmer (third component; 10%; Figure A5.14 in SM4.5). These indicators seemed unrelated with the indicators describing the degree of intensity.

Market indicators

Land prices increased from 2006 till 2008, after which prices stabilized at just above 50,000 ϵ /ha until 2013. From 2013 onwards, land prices increased till over 70,000 ϵ /ha in 2019. Interest rates went up from 3.8% in 2006 to 4.3% in 2007, after which interest rates steadily decreased to negative values in 2019. Interest rates often dropped more than 0.5% per year. Oil prices varied from 64 in 2006 to over 100 ϵ /100 L in 2019 and fluctuated over time with a peak in 2013 and 2014. Fertilizer prices increased from 27.75 ϵ / 100 kg fertilizer in 2007 to 61.50 ϵ /100 kg in 2009 after which they fluctuated between 41 and 47 ϵ /100 kg. (Figure 4.5; Table A2.2 in SM4.2 for absolute values)



Figure 4.4. Observed levels of important farm characteristics (explanatory variables) for Flevoland, Zeeland and Veenkoloniën.



Figure 4.5. Development of relative values of market indicators over time. Absolute values can be found in Table A2.2 in SM4.2. Absolute values in 2006 were 44506 \notin /ha (land price), 3.8 % (interest rate), 64 \notin /100L (oil price), and 27.35 \notin /100 kg (fertilizer price).

Weather conditions

The three case studies were similar in terms of average temperatures per year and per season (spring, summer). With regard to the precipitation deficit, the three case studies had similar values for spring (1.5 + -0.4 - 0.5 mm/day precipitation deficit; Figure 4.6), but for summer, FL had a lower average deficit (0.3 + -1.0 mm/day) than VK (0.6 + -0.9 mm/day) and ZE (0.8 + -1.0 mm/day), which was probably related to the higher precipitation in FL (2.6 + -0.8 mm/day) than in VK and ZE (both 2.3 + -0.7 mm/day). No significant trends in temperature, precipitation and precipitation deficit could be detected over the measured period (2006-2019; Table A2.1 in SM4.2). Weather extremes occurred regularly (Table A2.4 in SM4.2).



Figure 4.6. Precipitation deficit, precipitation and temperature in spring in the three case study areas.

4.3.3 Sparse Partial Least Squares regressions

Model performance

On average, across the three case studies, best performing sPLS models included the response variables related to profit of crops, nitrogen surplus and crop rotation energy yield. In all case studies, the predictive power of selected components was at most moderate. The variation in X-Variables explained by the X-components was low (Table 4.3). The explained variation in the Y-Variables was higher but still moderate. The best model in VK had three components with a varying number of response variables across components (Table 4.3). In ZE, sPLS-models performed better when including output efficiency instead of profit of crops (SM4.7). Because the interpretation of sPLS models including either profit of crops or output efficiency is almost identical, we proceed with sPLS models including profit of crops.

above the selection threshold of 0.095 (see section 4.2.4). Number of variables Variation explained kept Region Component Q2-score R2-score Х-Y-space X-space Y-space space FL 1 0.113 0.185 6 4 0.154 0.274 2 0.127 2 0.133 9 0.078 0.210 0.227 VK 0.351 9 3 1 0.147 0.298 2 0.304 0.234 9 2 0.299 0.096

9

2

4

5

5

2

0.098

0.175

0.081

0.184

0.193

0.164

0.093

0.129

0.186

Table 4.3. Number of variables and performance per component of selected sPLS-models with the response variables crop rotation energy yield, nitrogen surplus and profit of crops. All X-variables included. Bold font indicates Q2-scores that are above the selection threshold of 0.095 (see section 4.2.4).

Interpretation

ZE

3

1

2

0.170

0.073

0.064

In all case studies, there was one component associated with weather conditions that covaried with the residuals of nitrogen surplus (FL, VK), profit of crops (FL,ZE) and/or crop rotation energy yield (VK). In FL, nitrogen surplus was affected mostly by drought in spring, or wet conditions later on in the growing period, while profit of crops was positively affected by heatwaves, and generally high temperatures in summer (see 2nd component in Figure 4.7 and Figure 4.8; Table A6.3 in SM4.6). Interestingly, in contrast to droughts in spring, precipitation deficiency in spring seemed to somewhat improve profits and reduce nitrogen surplus. In ZE, profit of crops was affected negatively by high temperatures, specifically in spring, which was also related to precipitation deficit in that season (SM4.6). Interestingly, farms in ZE seem to benefit from warm winters. The availability of water (precipitation, absence of drought) was correlated to high crop rotation energy yields in VK.

In all case studies, the other component was associated with at least one indicator related to input intensity, the most important being monetary input intensity (all fixed + variables costs of a farm expressed per ha)(ZE, FL; Figure 4.7; SM4.6), total costs per hectare (ZE,VK), depreciation (FL) and labour (VK). These were negatively correlated with phosphate (VK) and nitrogen (VK, FL) and true diversity (FL). The high intensity in FL in combination with low nitrogen inputs and low diversity, resulted in high and increasing profits, low crop rotation energy yields and a declining nitrogen surplus (Figure 4.7 and Figure 4.8). For "expensive" crops, relatively little money is spent on nutrients. The high intensity in ZE was associated with declining profits from crops, low crop rotation energy yields, and to a lesser extent with low, but increasing nitrogen surplus were positively linked with phosphate input and low intensity. However, for VK, additional indicators related to economic conditions were associated with lower profit of crops (oil prices, interest rates, land prices), while labour input, farm management and other revenues seemed to compensate for this to a small extent.

In VK, a third component was associated with input indicators of which nitrogen, energy and crop protection products were the most important. These were negatively associated with the share of cereals in the crop rotation. Higher levels of nitrogen and energy input were associated with higher crop energy rotation yield, higher, but over time decreasing, nitrogen surplus and higher and increasing profit of crops. Vice versa, a higher share of cereals in the rotation seemed to reduce nitrogen surpluses.

Interestingly, the strong positive correlation between nitrogen input and nitrogen surplus in the original data of all three case studies (Figure A6.18, A6.19, A6.20 in SM4.6), was only included in the final sPLS model in VK. In the sPLS-models with fifteen response variables, the correlation between nitrogen input and surplus was absent in all three case studies (Figure A7.1, A7.2, A7.3 in SM4.7).



Figure 4.7. sPLS model results for the first and second X-component in Flevoland. Groups indicate the different years. The left and bottom axis indicate the position of observations in the projected X-space. The top and right axes indicate the correlation of explanatory variables with the first and second component.



Figure 4.8. sPLS model results for the first and second Y-component in Flevoland. Groups indicate the different years. The left and bottom axis indicate the position of observations in the projected Y-space. The top and right axes indicate the correlation of response variables with the first and second component.

4.4 Discussion

4.4.1 Interpretation of results from a sustainability and resilience perspective

In this paper we aimed to study economic and environmental sustainability and resilience simultaneously and quantitatively. In particular, we aimed to identify resilience attributes at farm level, i.e. farm characteristics, that support sustainability and resilience regarding market and climatic conditions.

Intensity and farm performance levels and trends

Overall, intensity levels played out differently in the three case studies, thus limiting us in generalizing the role of intensity of crop management on economic and environmental farm performance. Higher intensity of farms in terms of euros spent was primarily associated with higher profits from crops in FL and ZE, indicating improved economic sustainability through intensification. In FL, the increased intensity and profit covaried with having additional crops next to the main crops potato, sugar beet and cereals, i.e. diversification. In ZE, this pattern of a relatively positive effect of crop diversity on profit was also visible in the original data, but not included in the final sPLS-model. In FL, a higher intensity level in terms of euros spent, including higher expenditure on crop protection products and energy, was associated with reduced nutrient inputs leading to a declining nitrogen surplus, indicating some gain in the environmental performance. By contrast, in ZE, intensity levels in terms of euros were positively associated with nitrogen input levels, but these were not related to any response variable on nitrogen surplus. In VK, intensity in terms of euros and in terms of nutrients applied were positively related, which positively affected energy yield, profit and nitrogen surplus. Only in VK, input intensity was linked to economic conditions, indicating that increasing production costs are potential direct drivers of intensification that lead to higher yields and

profits. Increasing production costs are indeed identified as a major challenge and intensification as an important strategy in VK (Chapters 2 & 6, Herrera et al. 2022).

Weather conditions and variability of farm performance

Intensity levels explained levels of farm performance, but not the year-to-year variability (residuals). Instead, weather conditions seemed to explain the year-to-year variability of farm performance.

In FL, farms seemed to benefit from drought in summer. In 2018, when drought in summer was experienced throughout Europe, the clay soils with their high water holding capacity and the opportunity of irrigation may have reduced the impact of drought, while prices were relatively good in this year. In FL and ZE, relatively high temperatures in spring seemed to be associated with the downward fluctuations in profits of crops per ha. In our dataset for FL and ZE, high temperatures in spring coincided often with high yearly temperatures and precipitation deficits.

The results for ZE also suggested that warm winters were actually beneficial for farm economic productivity, rather than being a weather extreme that causes early sprouting of potatoes in storage (Schaap et al., 2013). A possible explanation could be that warm winters, if extended into spring, lead to early sowing of potato and subsequently can lead to higher yields (Mulders et al., 2021; Silva et al., 2020). Yet another explanation could lie in the specific dataset under study: warm winters occurred seven times from 2006 till 2019, while high temperatures in spring and warm winters could therefore be an artefact, i.e. the coincidental opposite of the observed negative effect of high temperatures in spring. Longer time series would reduce the possibility of having results that could be considered an artefact.

In VK, residuals of crop yield and nitrogen surplus were affected by weather extremes. This suggests that nitrogen supply to fields is adapted to average conditions, resulting in nitrogen surplus peaks during or after years in which extreme weather events occurred. With expected increases of heat waves and droughts towards the future, adjusting nitrogen applications to possible lower yields becomes even more important. This finding may also apply to the other two case studies where nitrogen application is also high and correlated with nitrogen surplus, at least in the original dataset (Figures A6.18, A6.19, A6.20 in SM4.6). Unfortunately we were not able to verify this based on the final sPLS-models (Figure A6.3, A6.6, A6.9 in SM4.6) that seemed to mask the correlation between nitrogen supply and surplus.

4.4.2 Methodology

General reflections

sPLS can be seen as a projector and filter of high-dimensional data that accentuates certain patterns in the data. However, some patterns may also be overlooked. A general example is the loss of details in sPLS, compared to the PCA analysis. A specific example is the correlation between nitrogen input and nitrogen surplus in the correlation maps of the original (Figure A6.18, A6.19, A6.20 in SM4.6) and projected (Figure A6.3, A6.6, A6.9 in SM4.6) data: the correlation in the original data structure has disappeared in the projected data. Also the level of detail as provided by the PCA-analyses is not reached. For our case studies, the models reproduced generally well-known knowledge and experience that could be embedded in an already existing narrative. Including more management specific indicators and following individual farms as was done before by Martin et al. (2017) could improve model performance and the interpretation of results, but a large part of the variability is likely to remain unexplained (see e.g. Bouttes et al., 2018; Martin et al., 2017). At best, this positions the used methods as being explorative (hypothesis forming).

The method simplifies reality by assuming linearity over time and linearity regarding response to explanatory variables. Regarding time, the threshold regression analysis has compensated somewhat for this (see also SM4.6). Regarding explanatory variables such as input intensity levels, it should be noted that these are known to have a non-linear impact on food production and economic productivity. However, due to large differences in input use efficiency among farmers, de facto a linear function may be approaching the data well enough. Interaction effects, for instance of farm characteristics on the impact of weather extremes, could not be studied well. In our case studies, the combination of sPLS (instead of

PLS) and random effects improved model performance considerably, but also resulted in a focus on the general impact, rather than a farm specific impact, of weather conditions on farm residuals. In Figure 4.7 and 4.8, for instance, farms seem to be impacted in the same extent by weather conditions ,i.e. farm characteristics don't seem to influence this. However, it should be noted that weather conditions only explain a small part of variability. Moreover, sPLS (artificially) reduces co-variation between the different model components, compared to PLS as, for instance, was used by Martin et al. (2017). Studying interaction terms in multi-variate ordination techniques, such as PLS and redundancy analyses, are notoriously difficult (ter Braak and Šmilauer, 2015). In ter Braak and Šmilauer (2015), a few coarse methods are provided for (visually) assessing interaction effects for data from controlled experiments. Further development of such methods is needed before they can be applied to the datasets used in our analysis, i.e. relatively small, multi-level datasets with continuous and discrete values from an uncontrolled real life context.

sPLS in a sustainability and resilience context

By putting response variables in the context of resilience, a general idea about system's resilience could be obtained. It should be noted that the resilience of an individual farming system should in the end be evaluated in a broader context. For example, lower dependence on externally sourced nitrogen input may be good for reducing the environmental foot print and increasing resilience through increased autonomy. Some reduction in nitrogen input in the Netherlands is not expected to necessarily lead to yield decrease (Silva et al., 2021b; van Grinsven et al., 2019). Thus, there seems little risk of externalising environmental pressure to other regions through a decrease in production.

In our analyses, sPLS performed lower in more diverse regions, i.e. in FL and ZE, where farms were more different from one another and where crop prices are more variable compared to VK. This could imply that, when using relatively small datasets, sPLS should be applied to systems with rather uniform farms in a relatively stable economic environment, in order to detect patterns in farm data that is known to usually contain a lot of noise. Interestingly, diversity, in particular in the form of farming system heterogeneity, is considered important in the context of building resilience (Cabell and Oelofse 2012, Chapters 2 & 3). Moreover, stable economic environments are uncommon for most contemporary, intensive farming systems as most are exposed to (fluctuating market prices of) global markets (Giller et al., 2021; Nyström et al., 2019; Therond et al., 2017). Considering the reflections above, datasets containing more farms over a longer time span are needed to increase the usefulness of our methodology. However, even with large datasets of farms, finding patterns and good explanatory power is not guaranteed (Silva et al., 2020).

Small residuals were seen as indicative for farm stability and therefore positive for farm resilience regarding robustness. Some argue that stability is not the same as robustness and that more specific indicators are needed (e.g. Sneessens et al. 2019, Slipper et al. 2020). Interesting in the work of Sneessens et al. (2019) is the use of absolute benchmarks, e.g. for minimum wage reflecting economic performance, while our study and for instance Bouttes et al. (2018) look at deviations from the mean or trend without referring to standards. Similarly, yield indicators could be benchmarked against potential yields and environmental indicators to existing environmental standards. Using standards could put results of our type of work more into perspective of (societal) desired sustainability levels.

Selection of models and response variables influences results

Although sPLS is largely data-driven, the study design has influenced the results. With regard to the selection of components, the acceptable level of the Q2-score is arbitrary (Cao et al., 2017). We therefore presented the R2-values as well. Some of the components included the maximum or minimum number of indicators per component as specified a priori, i.e. for the sake of interpretability, arbitrariness was included here as well. Additional analyses suggested that model performance was relatively robust regarding the inclusion of strongly correlated explanatory variables (Table A5.3 in SM4.5). In contrast, in the case studies in this paper, strong correlations among response variables lowered model performance (Table A5.1 in SM4.5). More specifically, the strong correlation between nitrogen input and nitrogen surplus in the original dataset was disfavoured over correlations of other explanatory variables with response variables related to yield and economic response variables. This "finding" can be seen as an illustration how an abundance

of the relatively easy measurable indicators in the economic domain can mask patterns of generally less abundant and more difficult to measure environmental indicators. To avoid neglecting important environmental variables, overrepresentation of economic indicators should be discouraged.

4.5 Conclusions

Overall, our statistical analyses of farm accountancy data from three regions over a period of 14 years mostly confirmed already existing knowledge. Current levels of farm output and thus sustainability were mainly related to variables related to farm structure, in particular intensity-related indicators. Year-to-year variability of farm performance was mainly related to weather conditions and weather extremes. The usefulness of our method to test hypotheses on resilience attributes at farm level seems therefore limited, which may be at least partly due to the dataset.

The presented methods in this paper can be seen as a way to filter and project high-dimensional data and to accentuate patterns in the data. As such it is a useful way of getting to know the data. In the context of resilience of farms, while using a relatively small dataset, our methodology seems limited to a rather homogeneous farm population in a relatively stable economic environment. Larger datasets in terms of the number of farms and time span included should be used to increase the usefulness of our methodology. Researchers intending to apply this method in (arable) farming systems should be well aware of the influence they can have over the results through the selection of response variables. In particular regarding the relative abundance of economic indicators that could mask environmental indicators that are generally more difficult to measure and therefore less abundant.

Supplementary Materials

Supplementary materials 4: <u>https://zenodo.org/record/6511120#.Ym-pC9pByUk</u>

Assessing future sustainability and resilience of farming systems with a participatory method: a case study on extensive sheep farming in Huesca, Spain.

This chapter is published as:

Paas, W., San Martin, C., Soriano, B., van Ittersum, M.K., Meuwissen, M.P.M., Reidsma, P., 2021. Assessing future sustainability and resilience of farming systems with a participatory method: A case study on extensive sheep farming in Huesca, Spain. Ecological Indicators. 132: 108236. https://doi.org/https://doi.org/10.1016/j.ecolind.2021.108236

Abstract

Finding pathways to more sustainability and resilience of farming systems requires the avoidance of exceeding critical thresholds and the timely identification of viable alternative system configurations. To serve this purpose, the objective of this paper is to present a participatory, integrated and indicator-based methodology that leads researchers and farming system actors in six steps to a multi-dimensional understanding of sustainability and resilience of farming systems in the future. The methodology includes an assessment of current performance (Step 1), identification of critical thresholds whose exceedance can lead to large and permanent system change (Step 2), impact assessment when critical thresholds are exceeded (Step 3), identification of desired alternative systems and their expected improved performance of sustainability and resilience (Step 4), identification of strategies to realize those alternative systems (Step 5), and an assessment on the compatibility of alternative systems with the developments of exogenous factors as projected in different future scenarios (Step 6). The method is applied in 11 European farming systems, and the application to extensive sheep production in Huesca, Spain, is presented here, as its problematic situation provides insights for other farming systems. Participants in the participatory workshop indicated that their farming system is very close to a decline or even a collapse. Approaching and exceeding critical thresholds in the social, economic and environmental domain are currently causing a vicious circle that includes low economic returns, low attractiveness of the farming system and abandonment of pasture lands. More sustainable and resilient alternative systems to counteract the current negative system dynamics were proposed by participants; a semi-intensive system primarily aimed at improving production and a high-tech extensive system primarily aimed at providing public goods. Both alternatives place a strong emphasis on the role of technology, but differ in their approach towards grazing, which is reflected in the different strategies that are foreseen to realize those alternatives. Although the high-tech extensive system seems most compatible with a future in which sustainable food production is very important, the semi-intensive system seems a less risky bet as it has on average the best compatibility with multiple future scenarios. Overall, the methodology can be regarded as relatively quick, interactive and interdisciplinary, providing ample information on critical thresholds, current system dynamics and future possibilities. As such, the method enables stakeholders to think and talk about the future of their system, paying the way for improved sustainability and resilience.

5.1 Introduction

Agriculture in the European Union (EU) is generally highly specialized and intensive (Andersen et al., 2007), resulting in an abundant food production, but also often leading to the degradation of natural resources (Tilman et al., 2002). In addition, labour productivity and farm income is low in many farming systems in the EU (DG-AGRI, 2017). From a social point of view, quality of life in rural areas in the EU is often perceived to be low as well, especially in the poorer countries (Eurofound, 2019; Shucksmith et al., 2009). To improve sustainability, a balanced attention for social, environmental and economic system dimensions is important (Kharrazi et al., 2019; Kinzig et al., 2006). Inadequate management of natural resources, for instance, can be seen as a failure to understand how social, economic and environmental dimensions are interrelated (Allison and Hobbs, 2004). Interrelation of these dimensions often results in feedback loops in a system, resulting in non-linear behaviour. This makes it challenging to assess and interpret the effect of shocks, stresses and management options on the provision of system functions. In response to this challenge, several resilience frameworks have been developed to study agricultural systems (e.g. Callo-Concha and Ewert, 2014; Meuwissen et al., 2019; Tendall et al., 2015). Sustainability and resilience can be seen as two complementary concepts (Marchese et al., 2018; Meuwissen et al., 2019). Resilience in the form of robustness, adaptability or transformability is needed to maintain or improve sustainability. At the same time, sustainability is needed to ensure the access, availability and quality of resources to buffer shocks and set in motion adaptation or transformation.

For the context of a farming system (FS), Meuwissen et al. (2019) define resilience as the ability to ensure the provision of the system functions in the face of increasingly complex and accumulating shocks and stresses. By emphasizing the importance of system functions, Meuwissen et al. (2019), provide a practical way to combine the concepts of resilience and sustainability in a complementary way. To better understand the potential dynamics of farming systems, current as well as future sustainability and resilience need to be studied. Current resilience of European farming systems was for instance studied by Nera et al. (2020), Meuwissen et al. (2021) and in Chapters 2 & 3. Towards the future, system behaviour may differ according to the development of factors that are exogenous to the farming system (such as population growth and economic development), especially when shocks and stresses increase or when enabling conditions for changes are realized. Trespassing critical thresholds could for instance initiate cascading effects leading to a system decline (Kinzig et al., 2006). To avoid this, institutional actors may deliberately aim at changing threshold levels to enable innovation that provides an alternative to the dominant ways of producing (Westley et al., 2011).

Quantitative models are often used to assess, ex-ante, system performance and behaviour. Different types of studies and associated models can be distinguished (Van Ittersum et al., 1998). Based on statistical models, projections or predictions can be made about the average and probable performance for future conditions (e.g. Van Passel et al. 2017). However, because statistical models depend on (data) patterns from the past, only a limited range of all possible futures will be captured. Including a broader range of possible futures (scenarios) increases the opportunity to evaluate farming system resilience under different exogenous conditions that are all possible to happen. Incompatibility of farming systems with certain futures can be seen as a sign of non-resilience in case those systems have no capacity to adapt or transform. In itself, comparing farming systems with a broad range of futures directly contributes to foresight information supporting the capacity to anticipate shocks, which is seen as important for resilience (Mathijs and Wauters, 2020). In so-called explorations, optimization models (e.g. Rabbinge and van Diepen 2000, Ten Berge et al. 2000, Reidsma et al. 2015) and system dynamics models (Herrera, 2017b) can consider multiple possible futures, using scenarios capturing uncertainty on climate change and socioeconomic developments. However, these models need parameters which are sometimes also derived from statistical models based on past and current trends. Moreover, optimization models are of limited use for modelling dynamic transformations, as they are generally static.

Participatory methods can take into account multiple scenarios (Delmotte et al., 2013; Walker et al., 2002) and allow for input regarding transformational change (Quist and Vergragt, 2006) and resilience concepts such as critical and interacting thresholds (Resilience Alliance, 2010; Walker et al., 2002). It should be noted, however, that qualitative methods also are influenced by input from statistical sources and experts

that extrapolate past trends into the future. We argue that quantitative and qualitative approaches can be complementary. Participatory methods can be quick, interactive and flexible to start discussions about sustainability and resilience in the future, thus laying a base for further discussions and quantitative modelbased analyses (Chapter 2). Participatory methods allow for taking into account the voice of individual stakeholders as well as support stakeholder discussions to arrive at a common understanding and a shared vision for improvement of the system or problem under study. Stakeholder participation is important as stakeholders are usually involved in follow-up processes and thus need to agree with the problem definition and proposed action plan (Quist and Vergragt, 2006). Participatory input is valuable because system actors are able to provide empirical knowledge about their system (Delmotte et al., 2013) that reduce knowledge gaps of researchers (Sieber et al., 2018; Vaidya and Mayer, 2014). Vice versa, participatory methods are also important to identify the boundaries of local knowledge (Mosse, 1994). Stakeholder's perceptions are particularly precious, as they can explain or drive system dynamics as stakeholders are important components of socio-ecological systems (Walker et al., 2002). Hence, participatory methods can provide a first exploration of farming system structure, mechanisms, performance and behaviour in possible futures.

Discussions with stakeholders about future change can be challenging because stakeholder's mental models usually focus on maintaining the status quo with little imagination of alternative futures (Meuwissen et al., 2020). Other limitations for discussing farming system transformations may relate to the focus of experts on improving efficiency, vested interests, co-dependencies among system actors and institutional path dependence (Meuwissen et al., 2020). Participatory methods should therefore provide opportunities to go beyond the usual extent of stakeholder's mental models. Alternative systems, that avoid critical thresholds and increase sustainability and resilience simultaneously, should be explored, and new strategies to realize those alternative systems need to be compatible with possible future developments of exogenous factors as projected in different future scenarios. High compatibility of desired alternative systems with future scenarios increases the likelihood that those more sustainable and resilient systems will be realized. Consequently, this also decreases the likelihood that critical thresholds will be exceeded, resulting in farming systems with even lower sustainability and resilience levels.

We argue that a quick and flexible assessment of future resilience and sustainability of farming systems is still lacking in literature. In response to this research gap, this paper presents a participatory, integrated and indicator-based method to improve understanding of farming system sustainability and resilience. The method uses the concepts of critical and interacting thresholds to challenge stakeholders in a workshop setting to think about potential non-linear and undesired behaviour of their farming system. Following, stakeholders are elicited on desired alternative systems that avoid critical thresholds and thus improve sustainability and resilience (and vice versa). The method is flexible regarding: a. the information sources used as input for the workshop, b. the possibility to include case specific indicators and c. the stakeholder input during the workshop, i.e. alternation of individual input, small group discussions and plenary discussions. We illustrate the usefulness of the approach with an application to the extensive sheep farming system in Huesca, Spain. In this farming system, ongoing, interrelated economic, social as well as environmental developments are increasingly reducing the system's sustainability and resilience.

5.2 Methodology

5.2.1 FoPIA-SURE-Farm 2

The proposed methodology presented in this paper extends the Framework of Participatory Impact Assessment for Sustainable and Resilient European farming systems (FoPIA-SURE-Farm 1) approach for assessing sustainability and resilience of current systems (Chapters 2 & 3; Nera et al. 2020) with participatory assessments on resilience of European farming systems in the future (FoPIA-SURE-Farm 2). FoPIA-SURE-Farm 1 and 2 are based on the SURE-Farm resilience framework (RF; Meuwissen et al., 2019): 1) defining and delineating the farming system, 2) identifying main challenges, 3) assessing farming system functions, 4) assessing the system's resilience enhancing attributes (system characteristics that convey resilience to a system). While FoPIA-SURE-Farm 1 was mainly aimed at performance levels of main indicators, that represent main functions of the system, and resilience attributes, FoPIA-SURE-Farm 2 includes resilience concepts such as critical thresholds, interactions between thresholds (e.g. Kinzig et al., 2006), and regime shifts (e.g. Biggs et al., 2018).

In this paper we define the basis of a farming system as the farms producing the main products of interest in a regional context. Farming system actors included in the farming systems are the producers of main products and other actors that mutually influence one another (Step 1 RF). The perceived complementarity of sustainability and resilience is operationalized by distinguishing system challenges, function indicators and resilience attributes. In the context of resilience, challenges relate to the guestion "resilience to what?", such as resilience to weather extremes (Step 2 RF). Function indicators are case-study specific representatives for important system functions, such as "Food production" or "Maintaining natural resources", as direct metrics for those functions are often not available (Step 3 RF; Table A1 in SM5.1). In the context of resilience, function indicators relate to the question "resilience for what?". This relates to sustainability, which is defined as an adequate performance of all system functions across the environmental, economic and social domain (Chapter 2, Morris et al. 2011, König et al. 2013). Resilience attributes are characteristics that convey general resilience to a system (Step 5 RF; Table A2 in SM5.1). These resilience attributes can often be linked to system resources (Chapter 2), e.g. natural or social capital, that can only be maintained when system functions are performing adequately. To improve the flexibility of the methodology and the clarity and saliency of participatory input, just like for functions, case-study specific indicators may be used for resilience attributes, as well as for challenges. Based on workshop results, inductions are made about the resilience capacities of the studied farming system (Step 4 RF). For more details on the concepts used in this study, see Table A1.1 in the Appendix of Chapter 1.

FoPIA-SURE-Farm 2 consists of a preparation phase, a participatory workshop and an evaluation phase, and was developed for application and comparison across 11 European farming systems (Paas and Reidsma, 2020) (Figure 5.1). In this paper we present six key steps of the methodology (Figure 5.1). In Step 1, current performance and trends of function indicators and resilience attributes are assessed by the research team in the preparation phase. This assessment can be largely based on FoPIA-SURE-Farm 1 (Chapter 2), but other (grey) literature can also be used. In Step 2, critical thresholds of important system challenges, function indicators and resilience attributes are assessed by workshop participants. Based on Biggs et al. (2018) and Kinzig et al. (2006), we define critical thresholds as the levels at which challenges, function indicators or resilience attributes are expected to cause large and permanent system change. System's closeness to thresholds is consequently evaluated by the research team based on participants' comments and (grey) literature, e.g. based on ongoing trends identified in Step 1. In Step 3, performance of main function indicators and resilience attributes is assessed when critical thresholds of main challenges would be exceeded. Possibilities of interacting thresholds can be discussed during the workshop and in the evaluation phase, following the framework of Kinzig et al. (2006). Interacting thresholds are thresholds, that, when exceeded, lead to the exceedance of another threshold, i.e. there are cascading effects. In summary, Step 1, 2 and 3 provide an overview of possible system performance in case no adaptations for improved sustainability and resilience are made.



Figure 5.1. Workflow of FoPIA-SURE-Farm 2 during the preparation phase, participatory workshop and evaluation phase. Parentheses at the top of each block indicate who does the action. Dashed arrows and italic font indicate the respective parts of Step 2-5 that are conducted in the evaluation phase. *Step 1 can be based on outcomes from FoPIA-SURE-Farm 1 (Chapter 2) and/or other sources of information.

Keeping the sustainability and resilience of the current system and the impact of exceeding critical thresholds as a point of reference, Step 4 addresses possible desired changes of the farming system towards the future. Participants can indicate and discuss what alternative systems are possible when challenges would become more severe, and when/how certain function indicators and resilience attributes would improve compared to the current system configuration. Step 5 aims to gain information on the strategies that are needed to realize alternative systems. We indicate these strategies as "future strategies". Steps 2 to 5 correspond largely to the participatory workshop phase. In the workshop, individual, break-out and plenary sessions are alternated. Individual and break-out sessions are included to ensure that all participants can provide input, which can be used as input for further discussions in plenary sessions. The proposed session format in each step can be changed according to needs of the participants, as long as a balance between individual, break-out and plenary sessions is maintained. In Step 6, in the evaluation phase, researchers evaluate whether desired future systems, i.e. the current system maintained in the future and the alternative systems, are compatible with developments in Shared Social Pathways for European agriculture (Eur-agri-SSPs; Mitter et al. 2019, Mitter et al. 2020) and hence match exogenous developments at European level. The time horizon for the future is 2030 in all steps. In the next sections we present details of each of the six steps.

5.2.2 Current performance (Step 1) and critical thresholds (Step 2)

A pre-selection is made of most important system function indicators and resilience attributes, their qualitative description of performance (very low, low, moderate, good, very good performance) and developments (no change, strong or moderate negative or positive change) (Step 1). Step 1 can be based on FoPIA-SURE-Farm and/or other information sources. Participants individually evaluate the existence of critical thresholds related to function indicators, resilience attributes and challenges (Step 2). Walker and Salt (2012) mention that it is impossible to determine critical thresholds for resilience attributes because they all interact. However, we include resilience attributes as it stimulates thinking about resilience. Moreover, participatory input on thresholds can be interpreted as formulations of potential concerns for which management goals and strategies may be developed (Walker and Salt, 2012). In plenary sessions, individual input is discussed. Participants are free to discuss and conclude on the relative closeness of their

system to critical thresholds. In case closeness of the system to critical thresholds is not indicated by participants, the research team evaluates closeness based on the current performance levels, and magnitude of variation and/or trends. "Not close", "somewhat close" and "close" to thresholds are defined as respectively unlikely, somewhat likely and likely that the distance to critical thresholds will be trespassed in the coming ten years, based on knowledge on possible variation and/or trends. A fourth category is identified as current levels being already at or beyond the critical threshold ("at threshold or beyond").

5.2.3 What if thresholds of challenges are exceeded? (Step 3)

Per identified main challenge, it is evaluated in a participatory forecasting approach what the effect of a change beyond the indicated thresholds would be on main indicators and resilience attributes (Step 3). For this, the group is split in small groups of participants, each discussing one challenge. First, the expected direction of change of the challenge is clarified. Secondly, the relation between challenge and function indicator or resilience attribute is discussed. In each group, a moderator synthesizes this with a score of -, -, +-, + and ++ alongside arrows from challenges to function indicators and resilience attributes (Figure 5.2). A + relation implies that if the level of the challenge increases, the function indicator or resilience attribute). Verifications are also made in relation to possible interactions among and between function indicators and resilience attribute is indicated. This impact is scored referring to the expected performance level from 1-5, similar to FoPIA-SURE-Farm 1 (Reidsma et al., 2019a). In a plenary session, each moderator feeds back the results of the small group in a 1-minute pitch, after which participants can respond.

Based on the outcome of questions on critical thresholds and forecasting the impacts of exceeding them, the possibility of interacting critical thresholds is evaluated by researchers in the evaluation phase using the framework of Kinzig et al. (2006). Kinzig et al. (2006) specifically assess critical thresholds and cascading effects across scales for alternative future states of agricultural regions. Kinzig et al. (2006) distinguish the ecological, as well as the economic and social/cultural domain across the patch, farm and region scale. Thresholds of systems parameters can interact across domains and levels of integration (Kinzig et al., 2006). A good balance between developments in the different domains and levels may improve sustainability and resilience of a system (Walker and Salt, 2012). In systems with strong interactions between system variables at lower levels, vulnerability of the system at the focal level may increase (Resilience Alliance, 2010). This is especially the case when variables at lower levels are all aligned with regard to their closeness to critical thresholds (Resilience Alliance, 2010). An (almost) simultaneous exceedance of critical thresholds at lower levels may result in further cascading effects and ultimately result in an alternative, undesired system state at focal level, which in this study is the farming system. In the context of this paper we distinguish the environmental, economic and social domains and the field, farm and farming system levels.

5.2.4 Desired transformations of the farming system (Step 4)

In a forecasting approach for improved sustainability, results are largely based on dominant trends and causal mechanisms that often lead to low sustainability. Solutions for improved sustainability, therefore, ideally need to break these trends and causal mechanisms (Dreborg, 1996; Quist and Vergragt, 2006). In this part of the workshop, we therefore shift from a forecasting approach to a backcasting approach. A backcasting approach has greater problem-solving capacities in long-term challenges, because it is concerned less with what is likely to happen and more with what is desirable in the future (Quist and Vergragt, 2006). Picturing future systems may stimulate system actors to widen their perspectives and improve their understanding of the concept of sustainability (Dreborg, 1996). In this study, the backcasting approach is focused on alternative farming systems that have improved performance of function indicators and resilience attributes (Step 4).

To identify these alternative systems, all participants are asked to write on post-its alternative systems they desire if challenges cross thresholds and/or functions need improvement. This ensures that stakeholders can give their own input and are not directly influenced by others. If input is low, thinking

can be stimulated among participants by presenting alternative systems that are identified by the research team in the preparation phase. Based on the post-its, several alternative future systems are identified in a plenary session. These alternative systems may be combinations of suggestions of different participants. Some may be adaptations and some transformations of the current system. After giving them a name, per alternative system, one small group of participants is formed to further discuss which main function indicators and resilience attributes will change. In addition, changes in land use, sectors, objectives and other relevant aspects may be discussed. Participants in small groups also discuss the enabling conditions, i.e. how challenges and other drivers should change in order to be able to reach these alternative systems. Small groups consist of at least one moderator from the research team and three participants. In the evaluation phase, enabling conditions are categorized by researchers under the following domains: agronomic, economic, environmental, institutional, social.

5.2.5 Strategies to realize desired transformations (Step 5)

Taking alternative systems as the points of reference, the backcasting approach is continued by identifying strategies to realize the alternative systems, in the small groups. A strategy is seen and communicated to workshop participants as a "plan of action, or part of it, implemented by actors within and outside the farming system to maintain or reach a desired farming system in 2030". The workshops ends with a plenary session, in which participants are asked whether there is a shared vision about the future farming system. If such a shared vision is present, the discussion on the strategies to select is tailored towards this vision. If not, all possible alternatives and strategies are kept in mind. These strategies for future systems are compared with the strategies that have been implemented in the past and current system, as derived from FoPIA-SURE-Farm 1, to understand what should change.

5.2.6 Compatibility of future systems with Eur-Agri-SSPs (Step 6)

In the evaluation phase, carried out by researchers, the level of functions, resilience attributes, required strategies and enabling conditions in the different future systems are compared with future scenarios (Step 6). Future systems include the continuation of the current system in the future as well as the proposed desired alternative systems. As future scenarios we use the storylines of the Shared Socio-Economic Pathways adapted for European agriculture (Eur-Agri-SSPs; Mitter et al., 2019, 2020). The five Eur-Agri-SSPs include: agriculture on sustainable paths (1), on established paths (2), on separated paths (3), on unequal paths (4) and on high-tech paths (5). Mitter et al. (2019; 2020) take into account multiple indicators (Eur-Agri-SSP-indicators) that are categorized under the themes "Population", "Economy", "Policies & institutions", "Technology" and "Environment & natural resources". Per Eur-Agri-SSP-indicator researchers indicated how important an increase of this indicator is for the alternative system, where 0 is "not important", 1 is "somewhat important" and 2 is "very important". Expected developments of SSPindicators are taken from Mitter et al. (2020), where \forall , \rightarrow and \land were translated into the values -1, 0 and +1, indicating negative, no and positive changes, respectively. Multiplication of importance of positive developments for future systems {0; 1; 2} with expected developments of Eur-Agri-SSP-indicators {-1; 0; 1} is used as an approximation for compatibility. If, for instance, natural resources need to improve in a certain alternative system, this is aligned with the improvements foreseen for the Eur-Agri-SSP-indicator "natural resource management" in the sustainable paths scenario. This makes the alternative system and this scenario compatible, at least for this specific Eur-Agri-SSP-indicator. In a next step, compatibility scores are aggregated and transformed (sum of the compatibility scores divided by the sum of the importance scores) per theme (Population, Economy, etc.). Final compatibility scores per future system per Eur-Agri-SSP are an average of the overall section scores per theme, where values -1 to -0.66 imply strong incompatibility, -0.66 to -0.33 moderate incompatibility, -0.33 - 0 weak incompatibility, 0-0.33 weak compatibility, 0.33-0.66 moderate compatibility, and 0.66-1 strong compatibility.

5.3 Application to extensive sheep farming in Huesca, Spain

5.3.1 Case study description

The case study is the extensive sheep farming system located in the Huesca province, Northeast Spain. Huesca covers about 15,000 km² and two main regions can be distinguished: 1) The Pyrenees and pre-Pyrenees in the North, covering about 6,000 km², where agricultural activities are confined to extensive livestock; and 2) the southern part of the province, characterised by the plains of the Ebro depression (about 9,000 km²), where extensive farming (sheep, goat and cattle), intensive farming (pigs and broiler) and crop farming (rainfed and irrigated) are present.

In Huesca, the number of (ovine and caprine) decreased from 2,902 (1995) to 1,018 (2019) and the number of sheep from 923,399 (2005) to 521,501 (2019) (Gobierno de Aragón, 2019, 2016; MAPA, 2019a). The size of farms has shown an upward trend in the last years. The current size of a herd is between 200 - 1,000 sheep (Gobierno de Aragón, 2019). These trends are a result of the convergence of a range of economic, institutional, social and environmental challenges the farming system is facing. The extensive sheep farming system is highly dependent on EU and national subsidies, and hence, vulnerable to changing agricultural policy goals and increasing bureaucracy and control requirements. Regarding the social challenges, the case study area suffered a vast population decline over the last century (Bosque and Navarro, 2002) that comes along with a lack of skilled labour, social services and infrastructures. The low attractiveness of the farming system is increasingly limited in the access to pastures. The strategies that farmers have been implementing over time to deal with these challenges follow four management patterns, i.e. intensification, extensification, diversification and conservation (Bertolozzi-Caredio et al., 2021)

In addition to the provision of private goods, such as to ensure sufficient farm incomes and deliver highquality food at affordable prices the extensive farming system also provides public goods. Grazing helps to maintain and preserve the natural resources contributing to keep soil quality (Peco et al., 2017) and biodiversity by maintaining landscape heterogeneity (Ornai et al., 2020; Rodríguez-Ortega et al., 2014; Silva et al., 2019). Extensive livestock activity is also important to prevent forest fires by keeping the area clean from dry biomass (weeds and scrubs), which act as fuel in Mediterranean areas (Casasús et al., 2007; Ruiz-Mirazo and Robles, 2012). Grazing activities also provide recreational areas demanded by society (Bernués and Olaizola, 2012) and keep the rural areas attractive. As a result of the challenges mentioned in the previous paragraph, levels of functions in the farming system are generally perceived to be low (Chapter 3, Becking et al. 2019).

The clear presence of interacting economic, social and environmental domains makes the extensive sheep farming system in Huesca, an interesting case study for studying sustainability and resilience. In addition, there are signs of low sustainability, low resilience and consequently a pending decline of the farming system (Becking et al., 2019). The FoPIA-SURE-Farm 2 workshop was conducted on 14 February 2020 from 9.00 am till 3.00 pm with one break in the middle and lunch at the end. Eighteen people participated in the workshop, of which seven were farmers (five of the seven farmers belonged to an association). The rest of participants belonged to the agri-food value chain (veterinaries (3), cooperatives (1) and distributors (1)), and public sector (research institutes and Universities (3), and local public administration (4)).

5.3.2 Current sustainability and resilience performance (Step 1)

Participants agreed with FoPIA-SURE-Farm 1 results (Becking et al., 2019) on current performance of main indicators related to the functions of ensuring economic viability, food production and quality of life: gross margin, number of sheep and number of farms, respectively (Step 1). Two main resilience attributes discussed in the workshop were 'production coupled with local and natural capital' and 'diverse policies' (see also Tables B1 and B2 in SM5.1). As proxies for those resilience attributes, "availability of pastures" and "subsidies" were used, respectively, to ease the communication. The current performance of function indicators and presence of resilience attributes was considered low, with no change or moderately negative change (Table 5.1; 3rd column).

ц	2
ç	Ū
2	ם
ť	5

he	ge	
5	Jar	
4	C	
1	ΪVΘ	
tep	Jat	
S	Ъĝ	
õ	9	
ea	uo.	
as	str	
St	\rightarrow	
nlt	ì	
res	ge	c.
S (rar	te
m	Q	n S
/ste	ge	tio
S	an	na
ure	сh	va/
<i>ut</i>	Ve	ē.
L.	ati	the
₫ fe	eg	Ц.
tec	2	dc
<i>Sec</i>	ate	she
Xc	ler	rk.
1S 6	00	Ŵ
e O	2	he
inc	^	c ti
20	('L	7 E
0	j,	ait
erf	9	2
fр	ark	ы
0	(q	Sb
nt	ge	hа
ne	an	at
Ide	ĊP	42
/e/	Ve	8
lev	iti;	IO.
90	00	sf
an	9 4	io.
ИС	<i>uo</i> .	1ct
atic	str	gr
tuc	\leftarrow	de
SI	÷	ıre
ent	er	it c
irre	gre	9
C	ht g	a'
he	lig	E
n	e (10
e.	bu	2.
nc	ha	VS
ũ.	0	õ
£	1Ve	An
er	isci	D
Sp	ã	ho
Ite	ate	rks
ribi	ler	V0
itti	00	ē
e.	5	t
nc	$ \ \ \ \ \ \ \ \ \ \ \ \ \ $	12
ilie	Š,	ЪСC
res	0	air
g	ye.	bt
ar	e	ts o
DrS	вu	nlt
atc	Чa	res
dic	0 0	P.
in	Sn	t a.
uo	lie.	on
cti	du	d f.
'nn	⇒ ï	100
n J	1	nt
101	(YE	IS I
~	<i>jo</i> [MO.
5.1	opu	Arı
je	thc	÷.
rab	neı	rec
1.	-	\sim

				Expected f	uture develop	ments	
						Alternative	systems
						(Step 4)	
Function (F)/ Resilience attribute	Indicator	Current	Closeness to	Current	Critical	Semi-	High-tech
(RA) / Challenge (C)		level	threshold (Step 2)	system	thresholds	intensive	extensive
		(Step 1)		(Step 1)	exceeded	system	system
					(Step 3)		
F: Economic viability	Gross margin	Low	(Somewhat) close	↑	<u>ר </u> ר	۲,	R
F: Food production	Number of sheep	Very low	At threshold or	↑	→	R	¢
			beyond				
F: Quality of life	Number of farms	Low	At threshold or	R	→	۲,	<u>→</u>
			beyond				
RA: Production coupled with local and	Availability of	Low	(Somewhat) close	R	<u>↑</u> ⊼	٦	÷
natural resources	pastures						
RA: Diverse policies	Subsidies	Low	At threshold or	↑	<u>} א</u>	٦	R
			beyond				
RA: Socially self-organized	ı	Moderate	Not close	\uparrow	<u>} א</u>	↑	÷
RA: Supports rural life	ı	Low	Close	R	<u>→</u> ⊼	<u>↑</u> ⊼	۲,
RA: Infrastructure for innovation		Low	Close	R	<u>→</u> ⊼	۲,	۲,
RA: Reasonably profitable	ı	Low	At threshold or	٦	<u>ז ל</u>	R	۲,
			beyond				
C: Decreasing national lamb meat	Consumption per	ı	At threshold or	ı		ı	
consumption	inhabitant		beyond				
C: Increasing feeding costs		ı	At threshold or	ı		ı	
			beyond				
C: Increasing wild life attacks			Not close				
C: Lack of workforce	Workforce per farm		At threshold or	ı			
			beyond				

5.3.3 Critical thresholds (Step 2)

When discussing critical thresholds (Step 2), participants argued that these were already reached and that the farming system was on the edge of collapse/decline (Table 5.1). When participants resisted to participate individually, the flexibility of the methodology allowed for slightly adapting the procedure in Step 2². In order to stimulate the discussion and obtain values for thresholds, the trend and current value of the indicators according to the official statistics were presented to participants. In case of disagreement, participants were asked to define the current value of the indicators in a plenary session, which helped the researchers to determine how the discussed values were more or less close to the threshold. Based on the plenary discussions on thresholds, researchers deduced a number of enabling conditions that are needed to maintain the current system in the future. In the next sections, actual levels, developments and threshold levels of function indicators, indicators of resilience attributes and challenges are presented.

Main functions and related indicators

Economic viability: gross margin

Participants indicated that the gross margin is the decisive variable that determines whether the farming system is on the edge of collapse or not. Participants indicated that the gross margin threshold of the farms is 25-30 €/head. According to the literature, gross margins in the farming system vary among farms depending on feeding costs, size of herds (Milán et al., 2003; Pardos et al., 2008) and aids (Bernués and Olaizola, 2012; De Rancourt et al., 2006). This implies that not every farm is similarly close to the gross margin threshold. While the gross margin of the farms in the flat areas is at threshold and beyond (25-30 €/head), the distance of gross margins to the thershold appears larger in the farms located in the midmountain areas (40-45€ (MAPA, 2017)). The latter have lower feeding costs than the former because the herd feeding relies almost enterely on the availability of pastures. Herd size in mountain areas used to be higher allowing farmers to benefit from economies of scales. Farmers in mountain regions also receive least favoured area aids that increase their income.

Food production: number of sheep

Participants agreed that the current number of sheep has reached the tipping point in the area. There are currently about 521 thousand sheep heads in the province of Huesca, with a reduction of 43.7% since 2005 (Gobierno de Aragón, 2019; MAPA, 2019a). The decrease in the number of sheep in the farming system has not been as sharp as that of the number of farms. The reason that the decrease of sheep number has not been so marked in the last 10 years is because herds of quitting farms have been acquired by the farms that stayed.

The strategy of buying sheep from quitting farmers allowed other farmers to increase their margins and remain in the farming system. Pardos et al. (2007) found an average increment of 85 sheep per farm from the period 1996-2001 to period 2002-2005. Currently, farmers are investing a great effort and time managing between 500 - 1,000 sheep/shepherd, but the gross margins are not enough to hire new shepherds and increase the herd. Consequently, from now on the number of sheep is expected to decrease with each farm disappearing from the system.

Quality of life: number of ovine farms

Providing quality of life by means of creating rural employment with decent working conditions came up as one of the main functions of the farming system. This function is measured by the number of farms, as suggested by researchers and agreed upon by participants. Creating rural jobs contributes to keeping the rural areas attractive for residence and tourism. As also indicated by participants, the rural depopulation is an important challenge that this farming system has been facing since the last century (Bosque and Navarro, 2002). The depopulation seems to have more to do with the general socio-economic context of the farming system (lack of workforce, migration to urban centers, etc.), than with the sheep farming

² Due to the perceived closeness to critical thresholds in the studied system and participants' subsequent difficulty in following the normal procedures, the flexibility of the methodology was used to adapt procedures. While adapting the procedures, a balance between individual, small group and plenary activities was maintained to improve the chances that all participants could provide their input.

system itself (Bernués and Olaizola, 2012). The number of farms has decreased by 65% in the period 2005-2019, to the current value of 1,018 (Gobierno de Aragón, 2019, 2016), which is considered to be at the threshold or beyond.

Main resilience attributes and related indicators

Coupled with local and natural capital: availability of pastures

All participants agreed that the costs of feeding are strongly related to the availability of pastures. During the workshop, availability of pastures was assessed by looking at the total available surface of pastures (ha). In the province of Huesca the total amount of pastures has decreased by 65% in the period 2003-2018, with a current total of 160,000 ha in the province of Huesca (MAPA, 2019b). Participants concluded that the availability of pastures meets the farming system's needs, especially now the number of sheep has decreased.

However, in some areas such as the flat areas and those surrounding the Natural Parks and other protected areas (Sierra y Cañones de Guara Natural Park), the access to pastures is limited or nil. Although grazing contributes to modulate the vegetation dynamics (Bernués et al. 2005), bureaucracy and regulations limit the access to the pastures in the protected areas. Simultaneously, the increasing intensification of the agriculture in the flat areas is limiting the area of grazing lands. Moreover, the intensification of the farming system has led to the abandonment of lands, mainly in the mid-mountain areas. This abandonment causes a simplification and homogenization of the landscape due to the increase of the tree and shrub stratums, which lead to decrease in biodiversity and increase of fires (Lasanta-Martínez et al., 2005; Vicente-Serrano et al., 2000). Participants found it difficult to provide a minimum value of pasture surface they need for grazing, but they pointed out that the authorities must ease the access to pastures as well as compensate for environmental services delivered by the ovine farming system. Based on the input from participants, the research team estimated that the system is somewhat close to a critical threshold regarding the availability of pastures.

Diverse policies: subsidies

Participants explained that if basic payments would be lower than the current level, the gross margin would be null or negative, indicating that a critical threshold is reached. Farmers' incomes in the extensive sheep farming indeed depend greatly on aids (Bernués and Olaizola, 2012; De Rancourt et al., 2006). The basic payments was around $24 \in$ per sheep (MAPA, 2019c). Participants claimed that payments should increase at least by 30% to reach suitable gross margins. In fact, the decoupling of aids and the Common Agricultural policy (CAP) modulation have reduced the farms' income (Pardos et al., 2008).

Challenges

Lowering national lamb meat consumption

According to participants, the lamb consumption should not decrease more than the current level, indicating that the current level in fact is the critical threshold. Lamb meat consumption has declined strongly in the period 2006-2019 (50% of reduction), with a current value of 1.3 kg/inhabitant/year (MAPA, 2020). Participants mentioned that in the short term this challenge has a negative influence on the gross margin and the number of sheep, whereas, in the long term, it can lead to the closure of farms.

Participants identified several drivers that explain the lowering demand: consumers preferring other type of meats, mainly pork and chicken; disappearing culinary traditions; upcoming vegetarian and veganism trends; and the increasing campaigns against livestock farming influencing the negative perception of the sheep farming system (SM5.2). Overall, decreasing demand is indeed related to urban trends (Martin-Collado et al., 2019) and social-economic conditions such as consumer preferences and family structures (Corcoran, 2003). The quality of products from the case study area may give a competitive advantage (Bernués et al., 2006).

Increasing feeding costs

The feeding costs are a key element in the gross margin per head and at or beyond a critical threshold according to workshop participants. According to MAPA (2019c), the current average value of the feeding

costs is about 30€ per sheep in extensive sheep farms in Aragon. Participants agreed with this current value. According to Pardos et al. (2008) feeding costs depend on the type of farm. Intensive farms implementing feeding practices that rely more on feeds will deal with greater feeding costs (20-30 €/head) than extensive farms that rely more on the availability of pastures (14-17 €/head).

Droughts have been increasing in the last years (Hernández-Mora et al., 2012; Turner, 2005). Droughts are an important driver for increasing feeding costs, especially for those farms highly dependent on the availability of pastures for feeding the herds. For example, to overcome the low productivity of pastures caused by droughts in 2019, the European Commission allowed grazing in ecological focus areas (EFA) in Andalucía (Commission implementing decision (EU) 2019/1389, of 4 September 2019).

Increasing wild fauna attacks

Participants in the workshop are extremely worried about the increasing number of wolves and bears. The wild fauna attacks are recent and there are no clear statistics, but there is great concern about the potential impact. Participants did not provide the value of a critical threshold for wild fauna attacks in the ovine farming system. They indicated that the wild fauna attacks are more frequent in the mid-mountain than in the flat areas, where the attacks rarely occur. Participants mentioned that the attacks not only negatively affect the profitability of the farm, but also the farmers' quality of life as attacks imply more time and investments to take care of the herd. Based on the input from participants, the research team estimated that the system is not close to a critical threshold regarding wild fauna attacks.

Lack of workforce

The Annual Work Unit (AWU) per farm has shown a downward trend over the last years. The current value in the farming system is 1.9 AWU per farm. Participants agreed that this current value is the critical threshold for the workforce in the farming system. The low farm margins do not allow farmers to offer attractive labour conditions and hire personnel. Farmers run the farm alone or with family support. The socio-economic context of the farms such as the distance to major cities and the availability of public services in rural areas also explain the lack of workforce (García-Martínez et al., 2009).

The AWU per farm indicator is also indicative for the quality of life. A decrease in the AWU/farm value indicates a greater workload by the person(s) running the farm. Participants mentioned that the ovine farming system is very time consuming, mainly due to the shepherding. Shepherding is conditioned by the availability of pastures. In several occasions, pastures are far away from farms and farmers need to move long distances with the herds, spending a lot of time far from their families. The low number of shepherds limits the options to cooperate in shepherding and get time free.

5.3.4 Assessing the impact of exceeding critical thresholds (Step 3)

To compensate for the plenary input in Step 2, the research team decided that each participant should individually assess the impact when critical thresholds are exceeded (Step 3). In a plenary session all participants discussed the effects of exceeding critical thresholds of challenges and interactions between critical thresholds. Overall, exceeding the critical threshold of one of the challenges was expected to lead to moderate to strong decline in performance of main functions and resilience attributes (Table 5.1). Plenary discussion results are presented in detail in SM5.2.

In the evaluation step, interactions of thresholds across domains and scales (Figure 5.2) resulted in a vicious circle which explains the expected decline in system functioning when critical thresholds are approached and exceeded (Table 5.1). To adequately describe interacting thresholds in Figure 5.2, some additional indicators were added that came forward during the discussions with stakeholders. Figure 5.2 can be read as a summary of the information provided in the previous sections on thresholds of main function indicators, challenges and resilience attributes. Gross margin, a main function indicator of the system, plays a pivotal role in the interaction of thresholds and affects the number of farms and consequently the number of sheep in the area. Gross margins are directly affected by three main challenges: reducing subsidies, decreasing consumption and increasing feeding costs. Reducing gross margins and the closure of farms further reduces the available workforce, which reinforces the closure of remaining farms directly and indirectly via increasing feeding costs, which is why a lack of labour is seen

as a main challenge. The challenge of increasing feeding costs is indirectly affected by increasing occurrence of droughts and wild fauna attacks, two other identified challenges. These challenges reduce the access and use of pastures, a proxy for the resilience attribute "production being coupled with local and natural capital". Reduced access and use of pastures is eventually leading to shrub encroachment. Shrub encroachment is further stimulated when the number of sheep becomes insufficient to graze all available pastures. From a social perspective, the closure of farms and the decreasing workforce is expected to lead to a decreasing rural population.



Figure 5.2. Interacting thresholds between levels and domains for function indicators (rectangular shapes with sharp edges), indicators of resilience attributes (rectangular shapes with rounded edges), challenges (oval shapes) and additional indicators (rectangular shapes with dashed lines).

5.3.5 Alternative systems (Step 4)

Instead of providing defined alternative systems on post-its, participants proposed ideas in a plenary session, thus using the flexibility that the methodology is offering. Two main alternative systems, their goals, functions and resilience attributes (Table 5.1) and enabling conditions (Table 5.2) came up in the brainstorming.

Table 5.2. Relevance of enabling conditions categorized per domain. V implies that an enabling condition is relevant for maintaining the current system in the future and/ or moving to an alternative system. Tick marks in bold font are results obtained in the workshop. Tick marks in normal font are deductions from what has been said in the workshop in the evaluation phase. FS: farming system.

			Alternative systems		
			(Step 4)		
Domain	Enabling conditions	Current	Semi-	High-tech	
		system	intensive	extensive	
			system	system	
Agronomic	New technology applied to sheep FS farm	-	v	v	
	management				
Agronomic	Farmers training in new technology	-	v	v	
Agronomic	Improved sanitary conditions	V	v	V	
Agronomic	Improved animal handling	V	v	v	
Agronomic	Geo-localization technology	-	-	v	
Agronomic	Use of sub-products	-	v	-	
Economic	New financial products	V	v	v	
Economic	New commercialization channels and market	V	v	v	
	niches				
Economic	Public aids for public goods provision	V	-	v	
Environmental	Broader access to pastures and stubble fields	V	-	v	
Environmental	Sustainable pastures management	V	-	v	
Environmental	Research relationship nature-ovine FS	-	V	v	
Institutional	Reduced bureaucracy control	V	v	v	
Institutional	FS oriented legislation (sanitary,	V	v	v	
	environmental and urban)				
Institutional	Rural development	V	-	v	
Social	Public awareness of the contribution of FS	V	V	v	
Social	Improved cooperation among actors	V	-	v	

The first alternative system is the semi-intensive system. The main goal in this system is to improve the provision of private goods, i.e. increased meat production and improved labour conditions. Several enabling conditions at farm level were identified to reach this end (Table 5.2). This alternative system would fit better in the southernmost and flat areas where crop diversification is easier to implement.

The second alternative system is the high-tech extensive system. The aim is to improve farms' profitability by reducing feeding costs based on an improved pasture management. Participants highlighted the need for the innovation in herd geo-location, weather information and wild fauna surveillance (Table 5.2). In addition, subsidies are essential in this system to support the provision of public goods as well as a legal framework to regulate and protect the access to land for grazing purposes. This alternative system would be more suitable in the northernmost and mountainous locations, where there are more pasturelands and geography makes other types of farming systems less appropriate.

Current challenges, such as the reduced consumption of lamb meat by consumers, the lack of workforce and the increasing feeding costs, are still important in the future alternative systems. The feeding costs are more important in the semi-intensive alternative system due to a greater dependency of feed inputs (fodders) and lower dependency on the availability of pastures. On the other hand, wild fauna attacks will only pose a challenge in the high-tech extensive alternative system. In the alternative systems, all main functions are expected to increase in a moderate way (Table 5.1). The gross margins would increase in both systems, although margins seem to differ depending on the degree of intensification or extensification of the farms, as well as the areas where the farms are located. The increase in gross margin in both systems is the main change that is expected to allow to increase the number of sheep and farms, and are therefore moving away from other critical thresholds as well. The location of the farm determines the agroecological potential and the access to markets (Geoghegan et al., 1997). Thus, the semi-intensive alternative system is more likely in the flat areas where pastures are more scarce and payments for the less favourable areas are not applicable (Pardos et al., 2008). In the high-tech extensive alternative system, the production is not expected to change. However, its performance in less favoured areas (midmountains) and the provision of public goods services is supported by European subsidies that could increase the current margins. Greater gross margins would lead to a greater number of farms in the farming system, although this increase would be limited by the access to lands in the high-tech extensive system. The increase of the number of sheep is expected in both alternative systems, although this increment would be greater in the high-tech extensive alternative system. According to participants the lower production in this system would be compensated with greater herd sizes.

While some resilience attributes of the farming system ("infrastructure for innovation", "reasonable profitable" and "supports rural life") are expected to improve in both alternative systems, participants agreed that all the resilience attributes of the FS could improve in the high-tech extensive system (Table 5.1). The "social self-organization" resilience attribute in the high-tech extensive system would be improved as cooperation is needed to manage pastures and herds; it can also be argued that "production coupled to the local and natural capital" will improve as herd feeding will be coupled to the availability of pasture lands; and "diverse policies" will be enhanced as new policies will be tailored to support the provision of the public goods provided by the farming system. Moving towards the semi-intensive alternative scenario could constrain the resilience attributes "production coupled to the local and natural capital" and "diverse policies" leading to a deeper unbalance between the economic, social and environmental dimensions.

5.3.6 Strategies (Step 5)

Several current strategies, with currently low implementation levels, could be enhanced in the alternative systems. Some current strategies (in italics in Table 5.3) are compatible with the alternative farming systems. These strategies are mainly oriented to the economic domain, specifically related to the on-farm economic administration (investments in farm, savings, sales contracts, etc.) (Soriano et al., 2020).

Moreover, there were several new strategies identified during the workshop that match with current strategies (underlined in Table 5.3). Most of these strategies are economic strategies such as opening new marketing channels and developing new financial products and sales contracts that contribute to increase the robustness of the farming system to face hard times. Some institutional strategies are related to the public awareness campaigns about the positive contribution of the extensive sheep farming system to nature conservation and health. In the system, public awareness is expected to stimulate lamb meat consumption, which results in improved incomes. Public awareness is also expected to improve regulations for improving management of pastures, which in turn could lead to even more public awareness.

Most of the strategies proposed in the workshop are applicable for both systems and are mainly related to the need for improved technologies and innovation (normal font in Table 5.3). The number of proposed strategies was higher for the high-tech extensive system. The extra strategies in this system relate to the environmental and social domains, due to its more environmental-based and social nature. Institutional changes need to be made that improve the access to lands and the management of pasturelands, and the recognition of the farming system's contribution to the conservation of natural resources. This is expected to pay off in the economic domain, through subsidies and the lower feeding costs due to the use of pastures. Social measures are related to the promotion of generational renewal, which would increase the workforce in the farming system. The workforce availability improves the farmers' quality of life, stimulating the attractiveness of the farming system.

The quality of life is also improved with the implementation of new technology related to management of pastures and animal handling – in the semi-intensive alternative system the animal handling strategies are very important, mainly related to sanitary and production issues. The technology and innovation requires the cooperation between different actors in the exchange of knowledge and training in the technology (i.e., shepherds schools and GPS training in the high-tech extensive alternative system, and the management of more prolific breed and implementation of sanitary measures in the semi-intensive alternative system). The cooperation between farmers is also expected to increase the bargaining power and margins.

In any case, strategies regarding innovation and cooperation among system actors would be necessary, no matter what future system unfolds (no-regret strategies). It should be noted that the import of feed in the semi-intensive system reduces the coupling of production with local and natural resources. This could result in an opposite direction where, because of a worsening public image, less meat is consumed and regulations are getting stricter.

In both alternative systems, several strategies are oriented to technology implementation. The implementation of new technology generally does not allow for experimentation because of the great investments involved in new technology. For instance, in the high-tech extensive system the use of satellite images or the GPS per ewe is expensive. In the semi-intensive system, the replacement of more prolific ewes requires high investments. Strategies with low investment costs are related to the sanitary prevention, which lend robustness to the farming system (healthier animals that respond better to diseases), or the coordination among actors.

The probability of unfolding the high-tech extensive alternative system is expected to be larger than that of the semi-intensive system. The reason is that the semi-intensive system is going to compete with other intensive farming systems (e.g., pork) that are more profitable. The high-tech extensive system might highlight its importance in the contribution to the public goods and the conservation of the local breed *Rasa aragonesa*. As mentioned before, the greater availability of pastures makes the high-tech extensive system more suitable to mid-mountain areas. Farmers mentioned the high-tech extensive system as the preferable option in the future but also the most complicated to accomplish, especially without supporting policies in place. Besides, some of the technology for pasturelands and herds management is still in a development phase. In contrast, the lower presence (or absence) of pastures in flat areas of the farming system make the semi-intensive systems more appropriate in those areas. Participants pointed out that both alternative systems could attract young people to the farming system. Riedel et al. (2007) have related young farmers to a greater dynamism and technology adoption in the ovine production system and to the reduction of shepherding. Technology is indeed important in both alternative systems and (partly) replaces the need for actual shepherding.

Table 5.3. Current strategies and future strategies for different future systems. Current strategies are based on FoPIA-SURE-Farm 1. Strategies proposed for future systems that are currently being implemented are underlined. Current strategies (not explicitly indicated in the workshop for future systems) are indicated in italics. Strategies in normal font are the strategies that were proposed for future systems and that are not implemented in the present. Bold font checks indicate that these strategies were mentioned during the workshop for a specific system. Normal font checks indicate that, based on the discussions during the workshop, it seems likely that strategies will be applied in certain systems. FS: farming system

			Future s	vstems
Strategy	Domain	Current system	Semi- intensi ve system	High- tech extensi ve system
Use of technology for management efficiency improvement	Agronomic		v	v
(electronic readers, blood test, etc.)	•			
Research in more prolific and productive breeds	Agronomic	V	v	
Research for sanitary conditions of the ovine FS (new vaccines,	Agronomic		v	v
medicaments, etc.)	Agronomia	V	v	v
Use of technology for animal positioning (CDS, mobile phone, etc.)	Agronomic	v	v	v
	Agronomic			v
Farmers training in new technology	Agronomic	M	v	v
Financial products to cover market volatile prices	Economic	v	v	v
Financial products to cover droughts	Economic	V		v
Opening up a foreign market	Economic	v	v	v
Boosting of local consumption	Economic	V		v
Openness of local slaughternouses	Economic			v
Diversification (on-tarm)	Economic	V	v	
Alternative income sources (off-farm)	Economic	V		v
Investment in the farm assets	Economic	V	V	V
Costs reduction and flexibility	Economic	V	V	V
Sales contracts	Economic	V	V	V
Access to market information	Economic	V	V	V
Improvement of the access to pastures and stubble fields	Environmental			v
Use of technology for control of grazed pastures	Environmental			v
Research in methane emissions from ovine FS	Environmental		v	v
Use of technology for real-time communication with administration	Institutional		v	v
Trained administration staff in FS specificities	Institutional		v	v
Reduction of bureaucracy and excessive and specific regulations	Institutional		v	v
Tailored legislation in environmental management	Institutional			v
Tailored legislation in sanitary conditions	Institutional		v	v
New urban legislation	Institutional			v
Remuneration to the FS for contribution to public goods	Institutional			v
Improvement of legislation in relation to wild fauna	Institutional	V		v
Innovation of laws for products origin and certification	Institutional		V	v
Promotion of generational renewal (early retirements, access to land,	In atitudia na 1/Capial			v
etc.)	Institutional/Social		v	v
Creation of shepherd schools	Institutional/Social			v
Promotion of lamb meat consumption	Institutional/Social	V	v	v
Promotion of local breeds outside the FS	Institutional/Social			v
Improvement of awareness of FS contribution to public goods	Institutional/Social	V	V	v
Associations and cooperatives	Social	V	V	v
Improvement of quality of life (work intensity reduction with	Casial	M	v	
technology)	SUCIDI	v	v	v

5.3.7 Compatibility with Eur-Agri-SSPs (Step 6)

Based on the challenges, enabling conditions and strategies of the current and alternative systems, the extensive ovine farming system in the province of Huesca seems to be most compatible with a scenario on a pathway to higher sustainability with improved attention for the maintenance of natural resources (Eur-Agri-SSP1; Table 5.4), especially in the case of a high-tech extensive system. Compatibility with Eur-Agri-SSP1 is largely due to the increment of support for environmental services. As the current system is close to collapse, the compatibility with a scenario where the status quo is maintained as much as possible (Eur-Agri-SSP2) for the current state is limited. The establishment of the semi-intensive system is more compatible with Eur-Agri-SSP2 due to its production orientation. Eur-Agri-SSP3, with regional rivalry leading to amongst others slow technological process, is moderately to strongly incompatible with the current system and the alternative systems. In Eur-Agri-SSP3, specifically for the semi-intensive system, the lack of internationalization of markets, and for the high-tech extensive system the lack of environmental services valorization reduces compatibility. The semi-intensification of the farming system is evaluated as the only alternative system moderately compatible with Eur-Agri-SSP4, a scenario driven primarily by increasing social inequality, and Eur-Agri-SSP5, a scenario primarily driven by improvements in technology. The high-tech extensive system is even less compatible with Eur-Agri-SSP4 and Eur-Agri-SSP5 than the current system. Although the high-tech extensive system is most compatible with Eur-Agri-SSP1, the semi-intensive system seems the safest bet regarding its overall compatibility with all Eur-Agri-SSPs (for more detail see SM5.3 and SM5.4).

Table 5.4. Compatibility of the current system and alternative systems with different Eur-Agri-SSPs. With values -1 to -0.66 (dark red): strong incompatibility, -0.66 to -0.33 (light red): moderate incompatibility, -0.33 – 0 (orange): weak incompatibility, 0-0.33 (yellow): weak compatibility, 0.33-0.66 (light green): moderate compatibility, and 0.66-1 (dark green): strong compatibility.

	Eur-Agri-SS	Eur-Agri-SSPs					
Future systems	1: Sustainable paths	2: Established paths	3: Separated paths	4: Unequal paths	5: High- tech paths		
Maintaining the current system	0.51	0.32	-0.83	0.14	0.21		
Semi-intensive alternative system	0.63	0.66	-0.62	0.35	0.38		
High-tech extensive alternative system	0.73	0.43	-0.70	0.07	0.16		

5.4 Discussion

5.4.1 Insights from the case study

Critical thresholds and impacts when exceeding these (Step 1-3)

The outcome of the workshop suggested that, currently, the social, economic and environmental performance of extensive sheep farming system in Huesca, Spain is poor and declining. This is a common trend in Europe. Strijker (2005) explained that increasing opportunities outside agriculture, lower product prices, and higher land prices explained the continuous decline of extensive livestock grazing systems in several rural areas across Europe. Bernués et al. (2011) found that the lack of generational succession and the high opportunity cost of labour are also drivers of the disappearance of livestock farming in European Mediterranean countries. Most challenges, system functions and resilience attributes seem to be at or beyond critical thresholds, indicating simultaneously low sustainability and low resilience levels. Interactions between critical thresholds of challenges, functions and resilience attributes across levels and domains are perceived to be present. This emphasizes the importance of including multiple levels and domains when studying the sustainability and resilience of farming systems. This also emphasizes the complementarity between sustainability and resilience, albeit in a negative sense. Overall, the effect of exceeding thresholds is expected to strongly reduce system performance in terms of sustainability and

resilience. Economic viability at farm level plays a pivotal role regarding interacting thresholds. Participants indicated that exceeding the critical threshold for gross margin would result in a collapse of the farming system. This supports the idea that interacting indicators being close to critical thresholds at lower levels (field, farm) increase the vulnerability of the focal system (farming system) (Resilience Alliance, 2010). Interestingly, the level of gross margin is artificially maintained by subsidies that farms receive. This suggests a current focus on mainly economic sustainability, which in the long run may not be sustainable at all: subsidies may keep the fast responding "gross margin" away from critical thresholds, while the indicators relating to slower processes such as declining access to pastures in the environmental domain and lower attractiveness of the countryside in the social domain are not countered. Amalgamation of farms and livestock partly slows down the decline in sheep numbers and subsequent lower maintenance of the landscape. However, in the absence of subsidies and the limitations in managing huge herds, amalgamation is no longer profitable, which explains why participants expected a collapse. Biggs et al. (2018) mention that large shifts in socio-ecological systems are uncommon. The provisioning of agricultural subsidies could be seen as a main reason for continuing the status guo in some other agricultural systems in Europe as well. In the case of Huesca, change goes farm by farm, in terms of guitting and growing. However, there are limits to growth, relating to financial margins and availability of labour. Also the perspective of farmers that stay may change; it depends on how much the social fabric in a rural area is already eroded whether and how many farmers still can benefit from nearby facilities and off-farm work (Kinzig et al., 2006).

Alternative systems, strategies and Eur-agri-SSPs (Step 4-6)

Strong feedback mechanisms from the environment to the farming system seem not to be perceived by farming system actors. This seemed also lacking in participants' mental models in other case studies where the methodology was applied (Paas et al., 2020). This lack may reveal the boundaries of local knowledge. Instead of feedback from the environment, "lamb meat consumption" and "regulations" are perceived to provide strong feedback signals: a low natural state of pastures and dependence on feed imports pays off negatively via the public image of the extensive sheep farming system, which in turn may lead to lower lamb meat consumption and stricter regulations regarding pasture management. These feedback loops are expected to stay important in both proposed alternative systems that stimulate economic viability in order to steer away from other thresholds. It could therefore be argued that the alternative systems are adaptations in reaction to challenges rather than transformations in which farming system structure and functioning changes radically. Bernués et al (2005) also found adaptation alternatives to reinforce the sustainability and resilience of the extensive farming systems. They proposed adaptations such as to define work organisation schemes that allow variations in labour needs, to explore the product mix that facilitates to transfer risks, and to increase the utilisation of on-farm resources (fodder and grazing) and the productivity (lambs per ewe or kg per lamb).

In the high-tech extensive system, more attention is given to landscape maintenance, which increases the number of enabling conditions and strategies compared to the semi-intensive system. For the high-tech extensive system this implies continued dependence on subsidies and in general more dependence on cooperation with actors inside and outside the farming system. The higher level of enabling conditions and strategies of the high-tech system compared to the semi-intensive system reduce the likelihood of matching all developments in each specific Eur-Agri-SSP, which is reflected in the reduced compatibility with most Eur-Agri-SSPs. On the one hand, this could be interpreted as having low resilience. On the other hand, the high-tech scenario moves towards an improved balance between economic, social and environmental functions. Improving this balance is suggested to improve general system resilience (Walker and Salt, 2012). The semi-intensive system seems more resilient regarding its higher compatibility with Eur-Agri-SSPs. However, focus in this alternative system is mostly on economic functions, which could undermine general resilience. Participants also perceived that this alternative system has less chances of being realized, as there is more competition over land with other farming systems, compared to the hightech extensive system. In addition, lamb meat consumption and subsidies are expected to further reduce when production is becoming less pasture based. This leads to two methodological reflections. First, combining information on system trends and mechanisms, based on a forecasting approach, and requirements for realizing alternative systems based on a backcasting approach, shows the complementarity between the two approaches (Quist and Vergragt, 2006). Second, the local context seems

very important when assessing compatibility of alternative systems with Eur-Agri-SSPs at farming system level. At the same time, the methodology raises awareness that depending on which scenario is unfolding, the local context may change, which could leave certain alternative systems unviable. Shifting between system and scenario perspectives thus provides a means to triangulate stakeholder input with researchers' perspectives.

5.4.2 Methodology

A quick and flexible method

The proposed methodology provides a rapid and flexible way to assess multi-dimensional sustainability and resilience of future farming systems. Although qualitative in nature and covering many different topics, the methodology provides outputs that can be summarized and communicated in few key tables and a figure (Tables 5.1-4, Figure 5.2). The concept of critical thresholds is key to stimulate participants to think about potential permanent and large changes in their system. The notion of interaction between critical thresholds stimulates participants to think about interactions between challenges, functions and resilience attributes in the social, economic and environmental domain. Rapid resilience assessments are not widely available (Nemec et al., 2014) and are often inferring resilience solely based on expected presence of resilience attributes (e.g. Nemec et al. 2014, Tittonell 2020). Regarding the preparation phase, the method is flexible regarding the information sources used: results from FoPIA-SURE-Farm 1 or other sources of information. Other sources could be for instance (grey) literature, statistical databases and expert interviews. Provided time is managed strictly, the methodology turned out to be also flexible enough to be tailored to the local context and requirements with regard to changing individual, small group and plenary activities. Content-wise, the method also turned out to be agile with regard to including case study specific indicators, while the overarching concepts such as functions and resilience attributes allow for comparisons with other case studies. The same method was successfully applied in eight other SURE-Farm case studies in Europe (Accatino et al., 2020), allowing , for instance, to compare critical thresholds across case studies. In two case studies, desk studies were performed based on the method (Accatino et al., 2020), also representing the flexibility of the method. Unfortunately, time-wise, the workshops did not allow for an extensive discussion of all relevant elements and topics. An extension of one hour to the workshop would enable a better discussion on strategies to realize alternative systems, e.g. by discussing the prioritization of strategies, which actors and what resources need to be involved to implement strategies (Mathiis and Wauters, 2020), and whether there are trade-offs among strategies.

Influence of the research team

In case participants would not have assessed closeness to critical thresholds, the methodology suggests the research team to do this assessment. Such an assessment would be based on current levels and trends of main function indicators, resilience attributes and challenges. These levels and trends also serve as a points of reference (Table 5.1) and are based on previous work and other sources of information. This introduces an influence of the research team on the outcome of the workshop. Likewise, it should be noted that the method to assess compatibility of systems with scenarios, although transparent and useful for triangulating results, is also influenced by arbitrariness and subjectivity of researchers. For instance, when determining whether a development is important or very important for an alternative system, or when weighing the importance of the different groups of scenario indicators. The introduction of arbitrariness reduces the reproducibility of results. However, influence of researchers can also be explicitly accepted as a necessary part of an iterative, action-oriented process. In that process, researchers are actors aiming to develop, together with stakeholders, a shared, multi-dimensional understanding of current and future system performance (Wittmayer and Schäpke, 2014).

Participation and influence

The proposed methodology in this paper is designed to provide a voice to individual stakeholders as well as give room to develop a common understanding and vision for the studied farming system. Working with (a limited number of) participants also brings in subjectivity and arbitrariness. Suggestions from participants to make individual exercises plenary in this Spanish application may be the result of participants' interests to influence the flow and content of the workshop. For instance, to present private interests as formal knowledge (Mosse, 1994). In the case of extensive sheep farming in Huesca, Spain, collapse of the farming system seems pending, which may stimulate the expression of private as well as public interests to preserve the current system. The perception that the farming system has reached already certain thresholds could be a legitimate reason to avoid a discussion about where thresholds would lie exactly. The flexibility of the methodology allowed compensating the lack of individual input on critical threshold assessment by letting participants assess some other parts of the workshop individually. When discussing alternative systems, participants also expressed the need for a plenary discussion. Again, a pending collapse of the farming system may explain the need for immediate action, and thus not allowing for opportunities where time-consuming differences in opinion could arise. Another explanation could be that individual assessments were seen as dull, administrative tasks, resulting in reduced engagement and influence of certain stakeholder groups. Having plenary instead of individual input however also reduced the chance of having radical different ideas on alternative systems.

Adaptations or transformations?

By focusing on only a few system function indicators and resilience attributes, the likelihood of proposing alternative systems that integrate for instance new goals is reduced: importance of function indicators and resilience attributes in the current system may need to be re-evaluated in the light of possible alternative systems. In that sense, the followed methodology is to a certain extent path-dependent. This coincides with path dependency of social-ecological systems in general where actors have stakes and often change needs to be realized based on the resources that have been built up in the system so far. In the presented case study, alternative system goals shifted somewhat, but were largely emphasizing differences among goals of lowland and highland farming. Also in applications in eight other European farming systems, alternative systems proposed were adaptations rather than transformations. Only in an application where a desk study was performed (because of the Covid-19 situation) and main input was from experts, more radical transformations were proposed (Paas et al., 2020). Making farming system actors think about future change is indeed acknowledged as challenging (Meuwissen et al., 2020), but much needed in transition processes (Quist and Vergragt, 2006). As alternative to stakeholder participation, some foresight studies depend on expert opinions (e.g. Boland et al. 2013) or a literature study (e.g De Figueiredo et al. 2017). Although informative, these alternatives do not create a sense of ownership and engagement of local actors. Inviting radical thinkers from outside the system in a complementary workshop could help to challenge current mental models and to expose farming system actors to more radical ideas (Enfors-Kautsky et al., 2018; Westley et al. 2015). Another way to break free from established ways of thinking is to reframe the challenge (Enfors-Kautsky et al., 2018). For instance, by approaching it predominantly from an environmental perspective, a perspective that was not extensively discussed in the context of the workshop in the presented case study.

Representing the farming system

The representation of the farming system in the framework of Kinzig et al. (2006) gives a quick overview of important interactions for a system in decline. It should be noted, however, that getting an adequate system representation is always work in progress (Walker and Salt, 2012) and complementary methods are probably needed. For instance, the framework of Kinzig et al. (2006) does not provide a complete overview on possibilities to avoid or reverse a decline in system performance, based on knowledge of balancing and reinforcing processes in the system. The development of a causal loop diagram provides more insight on where these processes can be expected but is less intuitive and more complicated to interpret. Still, a causal loop diagram could help to qualitatively assess the impact of specific strategies. Further integration of causal loop diagrams and scenarios in system level (Herrera and Kopainsky, 2020).

5.5 Conclusions

The methodology presented in this paper leads researchers in six steps to a multi-dimensional understanding of future sustainability and resilience of a farming system. Taking the current system as

point of reference, the identification of interacting critical thresholds and assessing in a forecasting exercise the impact of exceeding these can explain how system sustainability and resilience can quickly decline (Step 1-3). Consequently, participants, being aware of this, are stimulated to think about alternative systems and their performance with regard to sustainability and resilience (Step 4). The alternative systems serve well as a point of reference in a back-casting exercise to identify the strategies that are needed to arrive at those alternative systems (Step 5). Although the workshop is originally designed to take five hours, taking more time for the workshop is advised as it will further improve understanding on the role of different strategies, actors and resources. Considering both feedback mechanisms (combining results from Step 1-5) and compatibility of alternative systems with Eur-Agri-SSPs (Step 6) provides a means of triangulation that allows for better understanding of strengths and weaknesses of the farming system, for instance with regard to the complementarity of sustainability and resilience of a system. Potential for decline (Step 1-3) and improvement (Step 4-6), simultaneously for sustainability and resilience, have been made clearly visible in the case study on the extensive sheep farming system that is included in this paper. Overall, the methodology can be regarded as relatively guick, interactive, flexible and interdisciplinary, enabling stakeholders to think and talk about the future sustainability and resilience of their system, paving the way for further discussions and also quantitative methods that can assess, exante, the impact of strategies and scenarios.

Supplementary Materials

Supplementary materials 5.1: <u>https://ars.els-cdn.com/content/image/1-s2.0-S1470160X21009018-mmc1.docx</u>

Supplementary materials 5.2: <u>https://ars.els-cdn.com/content/image/1-s2.0-S1470160X21009018-mmc2.docx</u>

Supplementary materials 5.3: <u>https://ars.els-cdn.com/content/image/1-s2.0-S1470160X21009018-mmc3.docx</u>

Supplementary materials 5.4: <u>https://ars.els-cdn.com/content/image/1-s2.0-S1470160X21009018-mmc4.xlsx</u>
Participatory assessment of critical thresholds for resilient and sustainable European farming systems

This chapter has been published as:

Paas, W., Accatino, F., Bijttebier, J., Black, J.E., Gavrilescu, C., Krupin, V., Manevska-Tasevska, G., Ollendorf, F., Peneva, M., San Martin, C., Zinnanti, C., Appel, F., Courtney, P., Severini, S., Soriano, B., Vigani, M., Zawalińska, K., van Ittersum, M.K., Meuwissen, M.P.M., Reidsma, P., 2021. Participatory assessment of critical thresholds for resilient and sustainable European farming systems. Journal of Rural Studies 88, 214–226. https://doi.org/10.1016/J.JRURSTUD.2021.10.016

Abstract

Farming systems in Europe are experiencing multiple stresses and shocks that may push systems beyond critical thresholds after which system change is expected to occur. These critical thresholds may lie in the economic, environmental, social and institutional domain. In this paper we take a participatory approach with involvement of farming system stakeholders to assess the presence of critical thresholds in 11 European farming systems, and the potential consequence of surpassing those with regard to system sustainability and resilience. First, critical thresholds of the main challenges, key system variables and their interactions in the studied farming systems were assessed. Second, participants assessed the potential developments of the key system variables in case critical thresholds for main system challenges would be exceeded. All studied systems were perceived to be close, at or beyond at least one identified critical threshold. Stakeholders were particularly worried about economic viability and food production levels. Moreover, critical thresholds were perceived to interact across system levels (field, farm, farming system) and domains (social, economic, environmental), with low economic viability leading to lower attractiveness of the farming system, and in some farming systems making it hard to maintain natural resources and biodiversity. Overall, a decline in performance of all key system variables was expected by workshop participants in case critical thresholds would be exceeded. For instance, a decline in the attractiveness of the area and a lower maintenance of natural resources and biodiversity. Our research shows that concern for exceeding critical thresholds is justified and that thresholds need to be studied while considering system variables at field, farm and farming system level across the social, economic and environmental domains. For instance, economic variables at farm level (e.g. income) seem important to detect whether a system is approaching critical thresholds of social variables at farming system level (e.g. attractiveness of the area), while in multiple case studies there are also indications that approaching thresholds of social variables (e.g. labour availability) are indicative for approaching economic thresholds (e.g. farm income). Based on our results we also reflect on the importance of system resources for stimulating sustainability and resilience of farming systems. We therefore stress the need to include variables that reflect system resources such as knowledge levels, attractiveness of rural areas and general well-being of rural residents when monitoring and evaluating the sustainability and resilience of European farming systems.

6.1 Introduction

Farming systems in Europe are experiencing multiple adverse shocks and stresses, such as weather extremes, price fluctuations and changes in policies and regulations. Under these multiple shocks and stresses, improving or even maintaining generally mediocre levels of sustainability of farming systems is increasingly challenged (Meuwissen et al., 2019).

The presence of critical thresholds adds dynamic complexity for farming system actors and policy makers. This is because beyond such thresholds, drastic system transformations may occur (Groffman et al., 2006; Kinzig et al., 2006) that are difficult to anticipate (Stockholm Resilience Centre, 2020) and to manage. For instance, the speed and scale of system processes after exceeding a critical threshold may be incompatible with the adaptation capacities of current institutions (Walker and Salt, 2012). Exceeding a critical threshold is most often undesirable as it generally leads to lower sustainability levels, e.g. a decline in biodiversity and human well-being (Biggs et al., 2018). Moreover, this state with lower sustainability levels may be more persistent resulting in reduced options to improve sustainability.

Timely knowledge on critical thresholds is therefore needed to prevent exceeding them (Resilience Alliance, 2010), but it is often difficult to anticipate the exceedance of a critical threshold (Stockholm Resilience Centre, 2020). In absence of clear knowledge on thresholds, Walker & Salt (2012) propose to work with thresholds of potential concern (TPCs) that inform management goals that aim to avoid those thresholds, without knowing exactly where they lie. In either case, the threshold level being known exactly or being a TPC, Monitoring is needed in order to detect the closing in on a critical threshold. Current monitoring frameworks of agriculture such as the Common Monitoring and Evaluation Framework (CMEF) in the European Union (EU), are mostly based on available statistics, leading to an overemphasis on economic data and an absence of data on social variables such as the well-being of farmers.

Participatory approaches could help to complement existing monitoring frameworks. Participatory input is a common way to define and assess environmental, economic as well as social indicators in an integrative way based on stakeholder perceptions (Chapter 2, van Calker et al. 2005, Morris et al. 2011, König et al. 2013). From a resilience perspective, closeness to critical thresholds of economic, environmental or social sustainability indicators can be seen as a sign of lower resilience. Perceived closeness to stakeholder-defined thresholds may hence be seen as a stress-signal of perceived low resilience. However, it should be kept in mind that perceived resilience is not always the same as resilience based on objectively defined and assessed resilience indicators (Jones, 2019; Jones and d'Errico, 2019). Although subjective, perceived resilience may explain stakeholder decision-making and resulting dynamics of the farming system. Closeness to critical thresholds may also inform the focus area of certain policies. Participatory input of farming system actors is also useful as it provides opportunities to take into account the local context and causal mechanisms at work. These are important to properly assess resilience and to realize adequate resilience-enhancing policies (Biesbroek et al., 2017).

In this study, we first further reflect on the importance of critical thresholds for resilience, and methods to assess these. Next, we assess in 11 European farming systems the closeness to critical thresholds of challenges and key system variables based on participatory input of stakeholders. The key challenges and system variables were defined based on the local context by researchers and stakeholders in previous studies (Chapters 2 & 3, Nera et al., 2020). We further use participatory input to assess the impact on main system variables in case critical thresholds of challenges are exceeded. Lastly, we use participatory input to reveal the interaction between critical thresholds, i.e. the exceedance of one threshold leading to the exceedance of another threshold. Based on the participatory input we discuss commonalities across farming systems. We finally use the commonalities to translate findings from a local context to national or EU-level policy recommendations and provide some suggestions for indicator development for the Common Agricultural Policy (CAP) 2021-2027.

6.2 Critical thresholds and resilience

In social-ecological systems (SES) research, there is ample evidence for the existence of critical thresholds whose exceedance leads to potentially undesired system transformations (Biggs et al., 2018; Rocha et al., 2015). Evidence in SES research is usually based on empirical data, theoretical models and statistics related to early warning signals (Rocha et al., 2015). Participatory approaches to identify critical thresholds are also proposed (Resilience Alliance, 2010; Walker et al., 2002; Walker and Salt, 2012). Still, large transformations or so-called regime shifts are not commonly observed in SES (Biggs et al., 2018; Carpenter et al., 2005). A hypothesis is that many SES are most of the time operating in a growth or consolidation phase, while their phases of decline and re-organization are usually short (Walker and Salt, 2012). Such a hypothesis may hold for the SES studied by Rocha et al. (2015) and Biggs et al. (2018), e.g. with regard to natural vegetation cover change in terrestrial systems or fish stock collapses in marine systems. In their studies, the focus is predominantly on passing critical thresholds in the environmental domain, as the degree of control over environmental processes or specific ecosystem services seems limited.

In SES such as contemporary European farming systems, anthropogenic inputs and human-induced adaptation processes are primarily aimed at controlling the level of food production. Transformations in farming systems may therefore be the result of gradually implemented adaptations in reaction to a changing environment, such as the gradual change towards agri-industrial entrepreneurship farming after the Second World War encountered in many European farming systems (Hardeman and Jochemsen, 2012). Therefore, in agricultural research, large transformations are often observed based on long-term historical studies on farming systems (e.g. Allison and Hobbs 2004, Termeer et al. 2019, Meuwissen et al. 2020), agricultural landscapes (e.g. Brown and Schulte 2011), or on a combination of both (e.g. Van Apeldoorn et al. 2013). Farming systems operate at a regional level (Meuwissen et al., 2019), a level for which Biggs et al. (2018) indicate that regime shifts develop slowly. This explains why large, gradual transformations can only be observed at longer time scales. In land use dynamics studies, large transformations can be simulated with quantitative models (e.g. Figueiredo and Pereira 2011, Brown et al. 2019). In these models, critical economic thresholds beyond which decision makers change activities are predefined inputs. However, apart from critical thresholds in the economic domain, critical thresholds in the social and environmental domain also need to be taken into account (Kinzig et al., 2006; Walker and Salt, 2012).

The work of Kinzig et al. (2006) is an example of how SES and agricultural systems research on critical thresholds and transformations can converge. Kinzig et al. (2006) and Walker and Salt (2012) propose to study transformations in agricultural regions by looking at interacting thresholds between field, farm and regional level and the social, economic and environmental domains. Critical thresholds are often associated with slow system processes, such as population dynamics and environmental changes (Resilience Alliance, 2010; Walker and Salt, 2012). Generally, indicators at higher levels of integration (e.g. countries) are dependent on slower processes than indicators at lower levels (e.g. farms) (Biggs et al., 2018). Indicators in the environmental domain are also often related to slow processes, while social indicators can be related to slow as well as fast processes (Walker and Salt, 2012). Warning signals of approaching critical thresholds of especially the slower processes in a system may go unnoticed or come too late (e.g. Van Der Bolt et al. 2018), while indicators related to faster processes are generally easier to measure. A distinction between thresholds of fast and slow variables and the identification of their interactions across levels of integration and the social, economic and environmental domain can therefore be useful to timely detect the approaching of critical thresholds.

6.3 Methodology

6.3.1 Farming systems and study design

This study is based on the "Framework of Participatory Impact Assessment for Sustainable and Resilient Farming Systems: future sustainability and resilience" (FoPIA-SURE-Farm 2; Paas and Reidsma 2020) applied to eleven European farming systems: large-scale arable farming in Northeast Bulgaria (BG-Arable), intensive arable farming in the Veenkoloniën, the Netherlands (NL-Arable), arable farming in East of England, United Kingdom (UK-Arable), large-scale corporate arable farming with additional livestock activities in Altmark, Germany (DE-Arable&Mixed), small-scale mixed farming in Nord-Est Romania (RO-Mixed), intensive dairy farming in Flanders, Belgium (BE-Dairy), extensive beef cattle systems in the Massif Central, France (FR-Beef), extensive sheep farming in Huesca, Spain (ES-Sheep), high-value egg and broiler systems in southern Sweden (SE-Poultry), small-scale hazelnut production in Lazio, Italy (IT-Hazelnut), and fruit and vegetable farming in the Mazovian region, Poland (PL-Horticulture).

FoPIA-SURE-Farm 2 consists of a preparation phase, a stakeholder workshop and an evaluation phase. The preparation and evaluation phase were exclusively conducted by the case study research teams. The research teams have been studying the resilience in their own case studies between June 2017 and August 2020. Stakeholder workshops were conducted in nine case studies between November 2019 and March 2020. This was a second round of workshops in a series of two, where the first round was focused on current and the second on future sustainability and resilience of farming systems. Participation in workshops was limited to farming system stakeholders, i.e. farmers and other actors that are influenced by and influence those farmers (Meuwissen et al., 2019), to make sure that participants had a good understanding of the local context. Farmers and participants from the government, (processing) industry, NGOs, agricultural advisors and researchers were present in the workshops (SM6.1). Farmers were the best represented stakeholder group. The stakeholder workshops lasted about half a day. Individual workshop reports are presented as Supplementary Materials to Paas et al. (2020) in Accatino et al. (2020). In BE-Dairy and FR-Beef, desk studies were performed, because planned workshops had to be cancelled due to measures that were put in place in the context of the COVID-19 outbreak.

6.3.2 Challenges, function indicators and resilience attributes

In this paper, we distinguish between system challenges, function indicators and resilience attributes. In the context of resilience, challenges relate to the question "resilience to what?" (Carpenter et al., 2001; Meuwissen et al., 2019), e.g. resilience to weather extremes. Challenges can affect the system regarding the functions it provides. Function indicators are case-study specific characteristics of important system functions, such as "Food production" or "Maintaining natural resources", as direct metrics for those functions are often not available (Meuwissen et al., 2019; for a complete overview of system functions see Table 1.1). In the context of resilience, function indicators relate to the question "resilience for what purpose?", e.g. resilience to maintain "Food production". Good values for function indicators can be seen as signs of high sustainability (Chapter 2, König et al. 2013). Challenges can also affect the system regarding its resilience attributes, i.e. characteristics that convey general resilience to a system (Chapters 1 & 2, Cabell and Oelofse, 2012, Walker and Salt, 2012; Table 2.5). Resilience attributes address the question "what enhances resilience?" (Meuwissen et al., 2019). High presence of resilience attributes is associated with high resilience. We argue that studying challenges, function indicators, resilience attributes and their possible interactions provides an opportunity to operationalize sustainability and resilience as complementary concepts (Chapter 5). For more details on the concepts used in this study, see Table A1.1 in the Appendix of Chapter 1.

For benchmarking purposes, case study research teams conducted an assessment of the current performance levels and trends of a few main *function indicators* and *resilience attributes* of the farming system. Main *function indicators* and *resilience attributes* were determined in the first round of workshops with farming system stakeholders, which were conducted one year earlier within the same research project (Chapters 2 & 3). In these previous workshops, eight system *functions* were determined (Meuwissen et al. 2019) and *indicators* were selected in relation to these functions. Perceived importance of both *functions* and *function indicators* was assessed by stakeholders, resulting in main *function indicators* important to

functioning of the system. For a set of 13 *resilience attributes*, the presence and contribution to resilience was assessed by stakeholders, resulting in an overview of perceived impact that attributes have on the resilience of the farming system. Contrary to the first round of workshops, the assessments in the second round of workshops were limited by the involved researchers to a few main *function indicators* and *resilience attributes* as critical system changes are expected to be determined by a small set of key variables (Kinzig et al., 2006). The main *challenges* of the respective farming system were also listed and described in each case study workshop. Participants were presented with and asked to comment on proposed main *challenges*, and (performance levels of) main *function indicators* and *resilience attributes* as obtained in the preparation phase, and the expected developments. As they are results of our first round of workshops, we present these here in order to keep a clear distinction from the results obtained in the second round of workshops and the evaluation phase.

Challenges were encountered in the agronomic, economic, environmental, social and institutional domain. We regard the challenges from the institutional domain as exogenous, where challenges from other domains may be endogenous as well as exogenous to the system. Common *challenges* in the economic domain across most case studies were low commodity prices and price fluctuations or high production costs. In the environmental domain, extreme weather events were experienced as a challenge in the studied arable, perennial and mixed crop-livestock systems. When extreme weather was mentioned in case studies, the occurrence of drought was defined as the most important extreme event. Environmental *challenges* damaging main products in case studies were encountered in NL-Arable (plant parasitic nematodes), ES-Sheep (wildlife attacks) and IT-Hazelnut (pests that reduce yield quantity and quality). A challenge in the social domain in multiple case studies was the low attractiveness of the area and labour availability. In the institutional domain, laws and legislations, and their continuous change, were experienced as *challenges* in most studied systems (SM6.1).

Main *function indicators* differed per case study to take into account the local context, but were representative for system functions, allowing for comparisons across case studies (Paas et al., 2019). *Function indicators* for "Economic viability" and "Food production" were most commonly discussed across case studies. *Function indicators* for "Natural resources" were mainly discussed in the arable systems, but also in SE-Poultry and IT-Hazelnut. *Function indicators* for "Attractiveness of the area" were mainly discussed in case studies in which rural isolation or outmigration was experienced (BG-Arable, DE-Arable&Mixed, IT-Hazelnut). In IT-Hazelnut for instance, the retention of young people was perceived to be representative for this *function*. The number of farms in ES-Sheep was perceived to be representative for "Quality of life". The happiness-index-of-farmers in UK-Arable was perceived to be representative for "Quality of life" and also relates to social isolation and to acknowledgement to and acceptance of farmers by society (SM6.1).

Resilience attributes were selected by researchers based on stakeholder perceptions in the first round of workshops. In those workshops, a pre-defined list of 13 attributes (Table 2.5) was used and could, therefore, be directly compared across farming systems. *Resilience attributes* that were discussed in most case studies were "Infrastructure for innovation", and "Production coupled with local and natural capital". *Resilience attributes* related to diversity, policies or connection with actors outside the farming system were least discussed. In SE-Poultry and PL-Horticulture the "Functional diversity" and "Response diversity" was emphasized. In DE-Arable&Mixed, RO-Mixed and to a lesser extent in IT-Hazelnut, "Support rural life" relating to the embeddedness of the farming system in the rural society was discussed because of rural isolation and/or outmigration that is experienced (see also previous paragraph). In ES-Sheep and IT-Hazelnut, the resilience attribute "Diverse policies" was discussed due to the pressure experienced from environmental regulations that reduce the competitive advantage because of higher production costs (SM6.1).

Levels of most of the main *function indicators* and *resilience attributes* are currently perceived to be slightly decreasing. In the perceived moderately performing systems IT-Hazelnut, SE-Poultry and NL-Arable (Chapter 3), overall moderately positive indicator developments were expected. In the perceived low

performing systems ES-Sheep and PL-Horticulture (Chapter 3), and also in UK-Arable, negative developments were expected.

6.3.3 Assessing critical thresholds in farming systems

With reference to current performance and ongoing trends it is interesting to know between what levels the main system *challenges, function indicators* and *resilience attributes* need to stay in order to maintain the current system configuration. Critical thresholds were defined as levels beyond which performance of all other key system functions is expected to drop below acceptable levels. Although multiple types of critical thresholds can be distinguished, all types have in common that system change after exceeding them is large and that reversing that change is challenging and costly (Kinzig et al., 2006). To not overcomplicate the concept in a participatory setting, we therefore defined a critical threshold as a point beyond which large and permanent, system change is expected. This change can have a positive as well as a negative connotation. However, as *challenges* are the point of departure in this study, overall change has predominantly a negative connotation.

Workshop participants were asked to individually note down critical thresholds of the main system *challenges, function indicators* and *resilience attributes*. Participants were encouraged to provide quantitative assessments of critical thresholds. When asked for by participants, members of the research team could suggest units for expressing critical thresholds. Notes with the stakeholders' assessment of critical thresholds were collected and posted on a wall and were left there for the remainder of the workshop. Notes were discussed in plenary sessions to explore possible critical thresholds and to reach consensus on critical thresholds. Stakeholders' notes of enabling conditions that help avoiding the exceedance of critical thresholds, rather than estimations of values for critical thresholds, were included in the plenary discussions and are summarized in a separate paragraph in this paper.

Closeness of *challenges, function indicators* and *resilience attributes* to critical thresholds was evaluated by the research team based on participants' comments and (grey) literature, e.g. based on ongoing trends identified in the preparation phase before the workshop. The position relative to the threshold was considered to be either "not close", "somewhat close" or "close" when it seemed respectively unlikely, somewhat likely or likely that the distance to critical thresholds would be trespassed in the coming ten years, based on knowledge on possible variation and/or trends. We relate proximity measures to likelihoods to indicate the approximative nature of our approach. An indicator that is "close", for instance, is likely to exceed a threshold within ten years, but exceedance can also happen after 30 years, which, however, is less likely. A fourth category of indicating the position relative to the threshold was "at or beyond". Detailed argumentation about the evaluation of closeness to critical thresholds is provided in SM6 2.

After discussing critical thresholds, farming system performance was assessed in case critical thresholds of main *challenges* would be exceeded in the near future. For each identified *challenge*, sub-groups of a moderator and at least three participants were formed on a voluntary basis. In those subgroups, the impact of exceeding the critical threshold of a challenge on main *indicators* and *resilience attributes* was discussed. A research team member functioned as moderator and used a poster to draw arrows between the *challenges* and main *indicators* and *resilience attributes* that were expected to be impacted. The strength of the expected impact was indicated by adding ++, +, -, ---, representing a strong positive, moderate positive, moderate negative and strong negative expected impact. As the impacts of exceeding thresholds were determined for the current system, challenges and their impact were discussed in the context of other challenges that are already present in the system. In this paper, therefore, we present and consider the overall impact of exceeding challenge thresholds as the impact of simultaneous stresses that have a combined effect at system level (Homer-Dixon et al., 2015; Walker and Salt, 2012).

The possibility of interactions between critical thresholds of *challenges*, *indicators* and *resilience attributes* was discussed during the workshops. Based on this, and based on the information acquired in the previous step and from literature, research teams aimed to reveal interacting thresholds across domains (environmental, economic and social) and levels of integration (field, farm, farming system) that cause farming system dynamics. Interacting thresholds are thresholds that, when exceeded, lead to the exceedance of another threshold (Kinzig et al., 2006). Determining whether thresholds were interacting

was based on qualitative argumentation by researchers using input from workshops. Detailed information on interacting thresholds per farming system is provided in SM6.3.³

To be able to concisely compare results from 11 case studies, our focus in this paper is on reporting and discussing the perceived relative closeness to critical thresholds and their interactions. The actual thresholds as noted down and discussed by stakeholders during the workshop are often very case-specific. Moreover, the precise level of critical thresholds was in most cases challenging to assess as stakeholders differed in opinion, and used different metrics. The assessments of thresholds are therefore mainly used to illustrate the methodology and our findings.

6.4 Results

6.4.1 Closeness to critical thresholds

More than half of the identified challenges were perceived to be "close" or "at or beyond" critical thresholds (Table 6.1). For extreme weather, closeness differed between farming systems: NL-Arable, IT-Hazelnut, PL-Horticulture, were perceived "somewhat close" to, DE-Arable&Mixed and BG-Arable seemed "close" to and RO-Mixed seems "at or beyond" the perceived critical thresholds. For the environmental challenge "pest & diseases", NL-Arable, challenged by plant parasitic nematodes, and IT-Hazelnut, challenged by phytophatologies, were perceived to be "somewhat close" to critical thresholds. For *challenges* in the social, economic and institutional domain, participants perceived more often that critical thresholds were reached than for the environmental domain. In ES-Sheep, participants indicated that for all challenges critical thresholds were reached, except for wildlife attacks (no threshold defined). In DE-Arable&Mixed, the lack of infrastructure and low attractiveness of the area were perceived to be at or beyond a critical threshold. In SE-Poultry, the perceived mismatch between economic viability on the one hand and the high production standards and strict environmental regulations on the other hand made participants indicate that for both challenges critical thresholds were reached. Continuous change of laws and regulations was seen as a main challenge in NL-Arable, UK-Arable, PL-Horticulture as well as BG-Arable, Participants in these case studies, for instance, perceived a critical threshold in the case that certain crop protection products would be banned before replacements had become available. A policy implication here would be to study a reasonable time for phasing out/in of policies. In DE-Arable&Mixed, SE-Poultry and RO-Mixed, inadequate alignment of policies and regulations at national and EU level was mentioned: national production quality standards increase production costs, while abiding with EU trade regulations allows for cheaper imports from countries with lower production standards and constraints.

³ Minor deviations from the methodology described above occurred in multiple case studies. BE-Dairy & FR-Beef: Desk study instead of a workshop. ES-Sheep: Participants argued that the system was already on the edge of collapse/decline. To still stimulate the discussion, the individual assessment of critical thresholds was turned into a plenary discussion. To this end, researchers presented participants with the statistics on the current values of the *challenges, function indicators* and *resilience attributes*. In case of disagreement with the presented values, participants were asked to provide the perceived current value of the indicator and the distance to its threshold. To balance plenary and individual activities, the researchers' team asked participants to individually assess s interactions between challenges, function indicators and attributes when critical thresholds were exceeded. Once participants reflected on this, they discussed their ideas in a plenary session. NL-Arable: Critical thresholds of resilience attributes were not discussed plenary due to time constraints. PL-Horticulture: Modified (aggregated) function indicators were used compared to the outcome of the previous workshop to achieve more structured and focused responses. Therefore four indicators were outlined based on the previous results, some consisting of several indicators of relatively high importance defined within the previous approach. SE-Poultry: Separate workshops were conducted for the egg and broiler production.

		Position relative to perceived critical threshold			No threshold defined	Not discussed	Total [†] (n)	
Challenge	Domain	Not close	Some- what close	Close	At or beyond			
Change in technology	Agronomic			1				1
Low prices and price fluctuations	Economic	1	2	2	1			6
High production costs	Economic			2	1			3
Extreme weather	Environmental	1	2	2	1			6
Pests & diseases	Environmental		1	1				2
Wildlife attacks	Environmental	1						1
Continuous change of laws	Institutional		3	2				5
Economic laws & regulations	Institutional	1	1		2			4
Environmental laws &	Institutional		1	1	1			3
regulations Lack of infrastructure	Social				1			1
Low attractiveness of	Social				1			1
Low labour	Social		1	1	1			3
Changes in consumer	Social				1		1	2
Total (n)		4	11	12	10	-	1	38

Table 6.1. Number of times challenges were assessed being in a certain position relative to the perceived critical threshold (aggregated results across 9 case studies; only main challenges were discussed in each farming system).

[†]For BE-Dairy and FR-Beef desk studies were conducted instead of workshops. Results from these case studies are hence not included in this table.

Participants could define critical thresholds for most system function indicators (Table 6.2); for instance, critical thresholds for the yield per hectare, an indicator related to the function "Food production", e.g. in BG-Arable, RO-Mixed and NL-Arable, Systems were perceived to be "close" to critical thresholds for "Food production" and "Economic viability" and "somewhat close" to those for "Natural resources" and "Attractiveness of the area". In IT-Hazelnut, for instance, the threshold for "Gross margin" relating to the function "Economic viability" was assessed to be 5,000 Euros per hectare, but was expected to differ from farm to farm. Based on current variability of markets and climate, it is likely that the value will someday drop below the indicated threshold, which makes that the system may be close to this critical threshold. For the seemingly low performing systems PL-Horticulture and ES-Sheep, some indicator levels were perceived to be at or beyond the threshold. In these systems, immediate action seems required, e.g. with regard to product prices and availability of labour in the area. Reaching critical thresholds for soil guality, an indicator representing "Natural Resources", was a concern in UK-Arable and NL-Arable. In those systems, participants mentioned that continuous adaptation is needed to prevent further degradation. In NL-Arable, a participant from the regional water board indicated that in the long-term water availability would decline, thus the system would approach a threshold. Most other participants took a more mediumterm stance and therefore proximity to this threshold was considered somewhat close. Overall, there was rarely a disagreement between participants about threshold levels. In BE-Dairy, where a desk-study was performed, water quality and greenhouse gas emissions were perceived to be beyond acceptable levels set by European and regional policy makers. Farmers in BE-Dairy are likely to disagree with these externally determined thresholds. In SE-Poultry, DE-Arable&Mixed, ES-Sheep and NL-Arable, participants indicated

that critical thresholds for economic viability differ from farm to farm. Hence, exceeding critical thresholds in these case studies may foremost imply the disappearance of economically less competitive farms from the farming system, rather than an immediate decline of the entire farming system performance.

Table 6.2. Number of times function indicators were assessed being in a certain position relative to the perceived critical threshold (aggregated results across nine farming systems; only main function indicators were discussed in each farming system).

		Position relative to perceived critical threshold				No threshold defined	Not discusse d	Total [†] (n)
Function indicator	Domain	Not close	Some- what close	Close	At or beyond			
Food production	Economic		1	4	3		1	9
Bio-based resources	Economic				1			1
Economic Viability	Economic		3	7	1		1	12
Quality of life	Social	1			1			2
Natural Resources	Environmental		4	1	2		1	8
Biodiversity & habitat	Environmental	1		1		2		4
Attractiveness of the area	Social		3			1		4
Animal health & welfare	Environmental			1			1	2
Total (n)		2	11	14	8	3	4	42

[†]For BE-Dairy and FR-Beef, desk studies were conducted instead of workshops and results from these case studies are hence not included in this table.

For resilience attributes, relatively fewer critical thresholds were defined than for function indicators (Table 6.3; 22 out of 37 vs. 35 out of 42). Thresholds of resilience attributes were mostly (semi-) qualitatively determined. For instance, in DE-Arable& Mixed "Supports rural life" was assessed to be on the lower end of a 1 to 5 scale where 1 implied very low and 5 implied a very high support. Participants indicated that a further decline in support would imply crossing a critical threshold. Overall, when defined, resilience attributes seem less close to critical thresholds than function indicators. From a methodological point of view, resilience attributes might be harder to grasp, and therefore more difficult to define and also perceived to be less close to critical thresholds than function indicators. From a theoretical point of view, the distance to critical thresholds could suggest that under the current challenges, resilience capacities are still sufficient to, for instance, start an adaptation or transformation process that steers away from critical thresholds of system challenges and indicators. However, the presence of some attributes e.g. "Reasonably profitable", when discussed and when a critical threshold was defined, was perceived to be close to a critical threshold, similar to the function "Economic viability" in most case studies (previous section). For the resilience attribute "Diverse policies", i.e. policies that equally support robustness, adaptability and transformability (Chapter 2), the systems in ES-Sheep and IT-Hazelnut were perceived to be at or beyond a critical threshold. In IT-Hazelnut the system was perceived to be close to a critical threshold regarding "Infrastructure for innovation". In IT-Hazelnut, current innovation levels were perceived already high, but would benefit from more to ensure further adaptation and improvement. For most other resilience attributes the system was perceived to be (somewhat) close to critical thresholds.

Table 6.3. Number of times resilience attributes were assessed being in a certain position relative to the perceived critical threshold (aggregated results across 9 farming systems; only main resilience attributes were discussed in each farming system).

	Position relative to perceived critical threshold				No threshold defined	Not discus sed	Total [†] (n)
Resilience attribute	Not close	Some- what close	Close	At or beyond			
Reasonably profitable			3			1	4
Production coupled with local and natural capital		2	1		2	1	6
Functional diversity					1	1	2
Response diversity		1			1	1	3
Exposed to disturbances			1			1	2
Heterogeneity of farm types			1		1		2
Supports rural life		2	1				3
Socially self-organized	1	2	1				4
Appropriately connected with actors outside the farming system	1				1		2
Legislation coupled with local and natural capital		1					1
Infrastructure for innovation			2	1	3		6
Diverse policies				2			2
Total (n)	2	7	10	3	10	5	37

[†]For BE-Dairy and FR-Beef, desk studies were conducted instead of workshops and results from these case studies are hence not included in this table.

While noting down and discussing critical thresholds, participants often mentioned enabling conditions that help avoiding the exceedance of critical thresholds, rather than precise values for critical thresholds. Enabling conditions can be seen as general notions of how system specific problems can be solved for the current system. Enabling conditions in the agronomic domain were mentioned only in BG-Arable, NL-Arable and ES-Sheep; e.g. improving productivity levels (BG-Arable) and availability of geo-localization technologies (ES-Sheep). Enabling conditions in the economic domain were e.g. creating access to new markets (ES-Sheep, IT-Hazelnut, NL-Arable), environmental payments (NL-Arable, ES-Sheep) and improving input/output price ratios (SE-Poultry, RO-Mixed, PL-Horticulture, NL-Arable, IT-HazeInut). Enabling conditions in the environmental domain were e.g. low occurrence of extreme weather events (BG-Arable, IT-Hazelnut, NL-Arable, PL-Horticulture, RO-Mixed), improved soil guality (NL-Arable, UK-Arable) and ecological and resource management regulations (IT-Hazelnut, RO-Mixed, ES-Sheep). Specifically in UK-Arable, emphasis was put on enabling conditions in the environmental domain. Enabling conditions in the institutional domain included good governance practices of authorities (BG-Arable, DE-Arable&Mixed, ES-Sheep, NL-Arable, PL-Horticulture, RO-Mixed, SE-Poultry) and access to knowledge, finance and/or land (BG-Arable, DE-Arable&Mixed, PL-Horticulture, RO-Mixed). Enabling conditions in the social domain were e.g. related to rural demographics and/or availability of labour (BG-Arable, IT-Hazelnut, PL-Horticulture, RO-Mixed, SE-Poultry, ES-Sheep, DE-Arable&Mixed) and more horizontal and vertical cooperation and social self-organization (BG-Arable, ES-Sheep, PL-Horticulture, RO-Mixed, UK-Arable). Specifically, in BG-Arable and RO-Mixed emphasis was put on enabling conditions in the institutional and social domain.

6.4.2 Interacting thresholds and impact of exceeding these

In all case studies, interacting thresholds across level and/or domain were observed (Figure 6.1; SM6.3). More details on the interacting thresholds are presented in the SM6.3. Common interactions between critical thresholds occur between field-environmental and field-economic, from field-economic to farmeconomic, from farm-economic to farm-social, from farm-social to farming system-social, and from farming system-social to farm-social (Figure 6.1), Generally, an environmental issue at field level, for instance, decreasing soil quality (NL-Arable, UK-Arable), pest diseases (NL-Arable, IT-HazeInut), wildlife attacks (ES-Sheep), or drought (DE-Arable&Mixed, PL-Horticulture, RO-Mixed, BG-Arable) is so much of a shock or stress that it leads to yields that are too low to sustain an adequate level of farm income (see SM6.3). In a majority of the farming systems, high input prices and decreasing output prices and sales further diminish the farm income. Too low incomes at farm level were in all case studies resulting in reduced attractiveness of farming, farmers guitting or the lack of finding a successor for the farm. In UK-Arable, also reduced farmer happiness due to lack of recognition was mentioned as a reason for guitting a farm. Farmers guitting their farm without having a successor was in multiple farming systems also considered to contribute to a smaller rural population at farming system level (FR-Beef, ES-Sheep, RO-Mixed, BG-Arable, IT-Hazelnut, PL-Horticulture; Figure 6.1). Interestingly, although socially oriented function indicators and resilience attributes were less often formally included in the discussions, they eventually appeared when explaining how challenges impact the farming system. Having less farms in the farming system was also associated with a lower maintenance of natural resources and a less attractive countryside (ES-Sheep, FR-Beef; SM6.3). Interactions with critical thresholds in the environmental domain at farm and farming system level were mentioned in a few other case studies. In NL-Arable, at farm level in the environmental domain a narrow rotation in which starch potato is grown every second year was expected to lead to increased pressure of plant parasitic nematodes (SM6.3). In UK-Arable, low income at farm level was expected to lead to declining soil health at field level (SM6.3). In IT-Hazelnut and SE-Poultry, environmental regulations were expected to improve the maintenance of natural resources at farming system level, but also to push farm income levels below a threshold through increased costs (SM6.3). Overall we observed that environmental thresholds certainly feature, but differ in the level at which they play a role and in what direction they evolve. In farming systems for which access to land is an issue (e.g. BE-Dairy, PL-Horticulture), guitting of farmers may also be an opportunity, provided land becomes available on the market for sale or to be leased. In ES-Sheep, quitting of farmers was experienced as a serious issue. In IT-Hazelnut, the retention of young people on the farms was specifically mentioned as something that could support the rural life and vice versa (SM6.3). Both low economic viability at farm level and low attractiveness of farming and a smaller rural population were considered to reduce the access to labour at farm level in BG-Arable, SE-Poultry, PL-Horticulture, DE-Arable&Mixed, RO-Mixed, and ES-Sheep. Access to labour in BG-Arable, PL-Horticulture and RO-Mixed was important for the continuation of activities on farms, as lack of labour was expected to push yields below acceptable levels (Figure 6.1). In BG-Arable lack of labour could be overcome by implementing new technologies, but this would require a labour force with higher levels of education and qualification which is even harder to find. Lack of labour was also expected to push production costs beyond critical thresholds in SE-Poultry and RO-Mixed. Hence, in multiple systems, low economic viability, attractiveness of farming, rural depopulation and low level of services at farming system level, and low access to labour seem to be part of a vicious cycle.



Figure 6.1. A synthesis of main interactions across scales and domains for 11 European farming systems (based on the framework of Kinzig et al., 2006).

Following from Figure 6.1, it can be made plausible that after exceeding critical thresholds of *challenges*, a decline in performance of system's main *function indicators* and *resilience attributes* was expected by workshop participants in most case studies (see SM6.1 for details). Across farming systems, the functions "Food production", "Economic viability", and the "Natural resources" were in most case expected to decline moderately or strongly (SM6.1). Especially system functions in arable systems were perceived to be moderately to strongly affected. In ES-Sheep, ongoing decline of function performance was expected to be aggravated. When discussed in case studies, "Biodiversity & habitat" and "Animal health & welfare" were on average expected to be less impacted compared to other functions.

When exceeding critical thresholds of challenges, also a decline in *resilience attributes* was expected in most case studies, mainly because of a decline in profitability, production being less coupled with local and natural capital, a declining support of rural life and lower levels of self-organization (SM6.1). By contrast, participants in BG-Arable and SE-Poultry generally expected improvements in *resilience attributes* after critical thresholds are exceeded (SM6.1). For instance, infrastructure for innovation was expected to develop positively in BG-Arable and SE-Poultry, while it was expected to develop negatively in other case studies (DE-Arable&Mixed, ES-Sheep, NL-Arable, UK-Arable). In the case of BG-Arable, participants expected that the current low profitability of farmers will not allow investment in new infrastructures for innovation.

6.5 Discussion

6.5.1 Closeness to critical thresholds

All studied farming systems were perceived to be "close" or "at or beyond" at least one critical threshold for challenges, function indicators or resilience attributes (Tables 6.1-3). The actual state of the system may be more or less close to a threshold than the participant's perception. Obviously, for case studies that are perceived to be "at or beyond" critical thresholds while still continuing business as usual, the actual state must be at a different position than perceived. Still, perceived closeness can be seen as a clear stress signal, indicating that change is needed, expected or even already experienced. An example refers to the ban of crop protection products before alternatives are available. This stress signal could instigate a study about a reasonable time to phase in/out regulations regarding the use of crop protection products before actually implementing them. Perceptions of being close to or at critical thresholds also indicate that, from the perspective of farming system actors, immediate action is needed to preserve the farming system or guide it in its transition, thus avoiding a situation where sustainability is even lower. Looking at multiple challenges puts individual challenges into perspective. To give an example, climate change may be a problem causing regime shifts in many socio-ecological systems (Biggs et al., 2018), but for the studied farming systems this is not the only *challenge* and often also not perceived to be the most urgent, except for some arable systems (Table 6.1). This supports the notion that climate change should be studied in the context of other drivers (Hermans et al., 2010; Mandryk et al., 2012; Reidsma et al., 2015). At a global level, reducing anthropogenically induced climate change is, of course, urgent and agricultural systems' contribution to it must be reduced. Some challenges experienced by FS actors, especially farmers, may also be implicitly caused by climate change; for instance changing legislation and high input costs. For most of the farming systems in our study, climate awareness of some stakeholders, such as conventional farmers, is however not likely triggered due to the impact of climate change on their system per se. When deliberated in an appropriate manner with those stakeholders, new legislation in the context of fighting climate change may however have considerably more effect regarding changing stakeholder perceptions.

Function indicators for food production and economic viability were often perceived to be close to critical thresholds. This confirms the need to closely monitor economic indicators as is done in the CMEF of the CAP (European Commission, 2015). When discussed, social function indicators were generally perceived to be "not close" or "somewhat close" to a critical threshold, except for ES-sheep where participants experienced that a critical threshold was exceeded (e.g., quality of life through number of farms, which lead to work generation) (Table 6.2). Environmental function indicators were in most cases perceived to be "not close" or "somewhat close" to critical thresholds (Table 6.2). Only in arable systems, environmental functions were experienced "close" or "at or beyond" critical thresholds. This was mainly related to the capacity of soils (at farm or field level) to deal with an excess or lack of water, often due to climate change. Participants in workshops of arable systems indicated that a lot of effort was already required to maintain rather than to improve the current soil quality. Arable systems, in need for soil improvement to avoid critical thresholds, would benefit from enabling conditions at national and EU level that foster the maintenance of natural resources. Mitter et al. (2020), based on a mechanistic scenario development approach for European agriculture, expect improved attention for natural resources only in a scenario following a "sustainability pathway" out of five possible future scenarios. Current conditions and their future development hence do not seem to support a resilient future of arable systems. Overall, perceived closeness to critical economic thresholds could explain the perceived lower importance of social and environmental functions compared to economic and production functions (Chapter 3).

Defining critical thresholds seemed most difficult for *resilience attributes* (Table 6.3). According to Walker and Salt (2012) it is actually impossible to determine critical thresholds for *resilience attributes* because they all interact. However, *function indicators* also interact, but were easier to assess for participants. We argue that difficulties in determining critical thresholds are probably more an indication of the perceived redundancy of *resilience attributes* for system functioning: presence and contribution to resilience was low to moderate according to stakeholders' perceptions (Chapters 2 & 3). This could be related to a control rationale (Hoekstra et al., 2018), in which keeping a relatively stable environment and improving efficiency is more important than increasing the presence of *resilience attributes*. It should be noted, however, that participants often could indicate enabling conditions that improve the *resilience attributes*. This could be an indication that participants are aware of the importance of *resilience attributes*, but are in need for more concrete, locally adapted indicators that represent the *resilience attributes*. In any case, suggesting improvements for *resilience attributes* could be seen as an implicit acknowledgment by participants that building capacities for adaptation or transformation is required.

Perceived thresholds may be different than the real threshold. For the systems that are perceived to be "at or beyond" critical thresholds, it is not necessarily too late to adapt in case the real threshold is actually at a different level than the perceived one. The extensive sheep system in Spain was judged to be close to a collapse, but alternative systems and strategies to reach those have been proposed (Chapter 5). In IT-Hazelnut, introduction of new machinery in the past has made farming more attractive for the younger generation, thus avoiding depopulation (Nera et al., 2020). Further developments in IT-Hazelnut regarding local value chain activities at farming system level rather than farm scale enlargement, are aimed to further stimulate economic viability and the retention of young people in the area (Nera et al., 2020; Paas et al., 2020). In PL-Horticulture, the case study is relatively close to Poland's capital where access to land is limited, system actors aim at increasing the economic viability via vertical and horizontal cooperation at farming system level, which keeps re-attracting seasonal laborers from nearby Ukraine, where wages are lower, to the region. The common factor in these examples of adaptation is that resources are needed to implement them. Be it financial, human, social or other forms of resources. The examples above also suggest that coming back to a desired state, even after exceeding a critical threshold, is possible, provided the disturbance causing the exceedance does not last too long (e.g. Van Der Bolt et al. 2018), and adaptation strategies are available (e.g. Schuetz, 2020). The notion of a critical threshold being a combination of magnitude (level) and duration was not discussed much in the workshops but could help to further define critical thresholds. For instance with regard to the number of years the farming system can deal with extreme weather events as was done in NL-Arable.

It is worth noting that *challenges* are perceived to be more often "at or beyond" perceived critical thresholds than *function indicators* and *resilience attributes*. From a system dynamic perspective this could suggest that the studied farming systems have some buffering capacity to deal with disturbances (Meadows, 2008). An example of this is the farm expansion in area and number of animals in many farming systems that compensates for the loss of farms from the system. From a methodological perspective, it could be argued that the participatory assessment of critical thresholds of *challenges* is easier than for *system functions* and *resilience attributes*. Critical thresholds of *challenges* are linked to important *function indicators* and *resilience attributes* and, therefore, may serve as warnings in the mental models of farming system stakeholders.

6.5.2 Interaction of critical thresholds

Based on workshop results and further reflections, interactions between critical thresholds are expected to (in)directly affect the economic viability at farm level, a central critical threshold observed in all farming systems (Figure 6.1). Economic viability at farm level is a relatively fast and measurable indicator. This gives another argument for monitoring income and other economic indicators in the monitoring frameworks such as the CMEF. The lack of a consistent pattern with regard to environmental thresholds indicates the importance of the local context.

In all farming systems, exceeding the critical threshold for economic viability at farm level affects the attractiveness of the sector, the number of farm closures and the availability of farm successors, which in turn in about half of the case studies contribute to lower availability of (qualified) labour and/or depopulation, which finally can reinforce low economic viability. Hence, a vicious cycle is initiated. This suggests that processes related to the economic and social domain can be driving dynamics of farming systems as well as being reinforced by those dynamics. This potentially can turn a relatively slow social process into a fast process. Social processes are therefore indeed important to monitor (Walker and Salt, 2012). This is already acknowledged in, for instance, in DE-Arable&Mixed, where participants emphasized the attractiveness of the area, specifically regarding the development of infrastructure.

Through its interactions with processes in other domains and levels, economic performance can be seen as an indirect driver as well as a warning signal for approaching critical thresholds in other domains and levels. In all farming systems food production was perceived to directly impact economic viability. Therefore, from the perspective of many farming system actors participating in our workshops, focus on food production and economic viability (FoPIA-SURE-Farm 1), which are based on relatively fast and measurable processes (Walker & Salt, 2012), seems often more justified than focusing on the more slowly developing social functions such as providing an attractive countryside. However, this may be due to the fact that (conventional) farmers were in most case studies the best represented stakeholder group, thus possibly masking the voices of other stakeholder groups that were represented less. In any case, social and environmental functions should not be overlooked as a focus on one domain will likely lead to missing important interactions with critical thresholds in other domains (Kinzig et al., 2006). For example, improving economic viability through scale enlargement and intensification, meaning fewer farms and often replacing labour by technology, often leads to a less attractive countryside. Regarding the environmental domain, focus on economic farm performance can even be dangerous as it could ignore externalized risk. For instance in UK-Arable and NL-Arable soil quality, the base of crop production and hence economic performance, was considered close to critical thresholds, while prohibition of certain crop protection products was seen as a challenge for the farming system, rather than the damage these products cause to surrounding ecosystems. Another example of externalized risk in one of our case studies is the pollution of water bodies in IT-Hazelnut. On their own, farmers may initially not have the willingness or capacity to look beyond the farm level. In IT-Hazelnut, farmers, through interaction with environmental actors, are now addressing these environmental issues. Building on this example, we argue that for instance societal dialogues and policy deliberations on improving sustainability and resilience need input from specific social and environmental actors, possibly even from outside the farming system. This seems necessary to counter-balance the bias towards economic performance at farm level by most of the participating farming system actors in most of our workshops.

In the more remote case studies, e.g. DE-Arable&Mixed and BG-Arable, attractiveness of the area seems low anyway. Consequently, improving prices alone, for instance, may not improve the availability of the necessary labour, thus reducing the emphasis on economic performance. Extensive rural development seems necessary to maintain the functioning of these farming systems. Mitter and et al. (2020), based on their mechanistic scenario development approach, expected no or negative developments regarding rural development in all future scenarios of European agriculture. The notion that both mechanisms at European and farming system level are not wired to address rural development, shows how the low attractiveness of an area can persist once it has come about.

Avoiding exceedance of critical thresholds without further adaptation or transformation, implies a performance at or below the current low to moderate levels for most system *function indicators* and *resilience attributes* (Chapter 3). A potential exceedance of a critical (and interacting) threshold in the coming ten years is expected to lead to negative developments for most system *function indicators* and *resilience attributes*. Negative developments of *function indicators* are expected in the economic, social as well as the environmental domain. On average, across all farming systems, we did not observe any differences in the magnitude of the effect between domains for *function indicators*. This consistent development confirms the idea that the different domains are interacting.

The consistent expected developments for *function indicators* and *resilience attributes* after exceeding critical thresholds suggest a perceived interaction between them. One could argue that a system needs resources to react to shocks and stresses (Meadows, 2008; Walker and Salt, 2012), especially for adaptation and transformation. These resources can only be adequately realized when there is an enabling environment and when system functions are performing well. The other way around, *resilience attributes* can be seen as "resources" to support system functions on the way to more sustainability. For instance, existing diversity of activities and farm types makes visible what works in a specific situation, openness of a system helps to timely introduce improved technologies, and connection with actors outside the farming system may help to create the enabling environment for innovations to improve system functioning (Table 2.5).

6.5.3 Farm level responses to reaching critical thresholds of challenges

Impact of *challenges* is primarily experienced at the farm level, resulting in the disappearance of (certain) farms from the farming system. In multiple case studies (SE-Poultry, DE-Arable&Mixed, NL-Arable), participants indicated that identified critical thresholds would be perceived differently among farmers. As mentioned before, farm closure generally leads to a less attractive countryside, a long-term process that is currently not perceived the most important issue in most studied farming systems, according to stakeholder input. Increasing farm size could be seen as a solution to compensate for the loss of farms and farmers in the farming system. Increasing the farm size is often associated with the advantage of economies of scale. For multiple farming systems in our study (NL-Arable, UK-Arable, SE-Poultry, BE-Dairy, ES-Sheep), production margins are low, which could further stimulate this thinking. However, from the farm level perspective, beyond a certain size, further economies of scale are not realized in some of the studied farming systems, i.e. there are limits to growth dependent on the rural context. In BE-Dairy, for instance, increasing farm size seems to be limited due to environmental standards. In ES-Sheep, further reduction of the farmer population is perceived to be harming the farming system, e.g. through reduction of facilities such as farmer networks, agricultural research initiatives, etc., but also hospitals, schools, etc. Besides, to further increase farm size, farmers in ES-Sheep depend on extra labour that is not available because of low attractiveness of the countryside, while investment in labour saving technology does not pay off with the current market prices. This is an example of the reflection of Kinzig et al. (2006) that a seemingly reversible threshold (no hysteresis effect) becomes irreversible because a certain management option to reverse processes is not available anymore. Based on Figure 6.1, we argue that this specific example may be true for more farming systems where a lack of labour force is experienced and investment in labour saving technology are not likely to pay off (e.g. RO-Mixed).

6.5.4 Implications for monitoring resilience

Social indicators

The importance of the social domain of farming systems makes us argue that indicators in this domain should be monitored. The option for countries in CAP2021-27 to shift 25% of the budget from income support (Pillar I) to rural development (Pillar II) provides the opportunity to adapt policies and investments to rural development needs. For instance for the more remote farming systems such as DE-Arable&Mixed and BG-Arable. We argue that a large shift of budget across the two pillars is already an indication of the perceived need to improve rural living conditions and can thus be used for monitoring. Although relating to economic values, the allocation of budget to rural development can thus be seen as the importance that is attributed to support processes in the social domain. Caution is needed however, as Pillar II also supports processes related to the environmental domain. Surveys among (agricultural) experts at national and regional level that record how much of the budget should be shifted from pillar I to II is a further step in assessing the performance of farming systems in the social domain. This implies introducing subjectivity in the CMEF on the evaluation side, while the choice of the parameter (shift of budget) is defined objectively, i.e. externally. Jones (2019) remarks that objectively defined and subjectively evaluated resilience assessments are relatively robust, easy and quick, while the limitations lay mainly in having to deal with bias, priming and social desirability. Other possibilities for objectively defined and subjectively evaluated indicators may lie in including indicators on living conditions and guality of life in rural areas based on Eurofound studies (Eurofound, 2021, 2019). These type of indicators also have the advantage of being entirely in the social domain, i.e. they don't indirectly refer to economic values such as the shift in budget from Pillar I to Pillar II as discussed above.

Monitoring resources

A common reflection in the discussion section so far is that having adequate system resources seems essential for stimulating system *resilience attributes* and dealing with challenges. In cases of low farming system resilience, building system resources may initially depend largely on external resources. This implies a role for regional, national and EU government bodies, i.e. a pro-active role for actors in the institutional domain outside the farming system. Given the tendency to focus on economic performance at farm level, external resources in the form of economic subsidies should be increasingly conditional regarding environmental and social functioning of the farming system. The emphasis on (accessible)

resources for building resilience is also acknowledged in several recent resilience frameworks (Duchek, 2020; Mathijs and Wauters, 2020), for instance with regard to knowledge and innovation systems (AKIS; Mathijs and Wauters, 2020). To elaborate on the example of AKIS, we argue that, rather than only monitoring and evaluating the amount of budget and the number of people that benefit from improved AKIS (as is currently done in for instance the CMEF), also the amount of this resource and stakeholders' access to it should be known and evaluated regularly. Similarly, other social and institutional resources need to be monitored next to economic and environmental resources.

6.5.5 Reflection on methodology

Given the challenges regarding assessing and discussing critical thresholds in workshops (stakeholder participation, differing stakeholder opinions, differing metrics, farm-specificity of thresholds, expert judgments of case study researchers on proximity to those thresholds), all identified critical thresholds could be seen as "Thresholds of potential concern" (TPCs; Walker and Salt 2012 citing Biggs and Rogers, 2003). In our case these TPCs would express the concerns of a selection of farming system stakeholders. TPCs can be seen as a set of evolving management goals that are aimed at avoiding critical thresholds that are expected, e.g. from experiences in other systems, but are not known. In case thresholds are considered beforehand as TPC's, O-methodology (McKeown and Thomas, 2013) may be an interesting participatory method to define which TPC deserves most priority. Estimating main functions of a system by assessing critical thresholds as TPCs, reduces the presence of clear sustainability goals. This makes the threshold assessment less dependent on externally determined values and criteria than most sustainability assessments (see e.g. Binder et al. 2010). Implicitly, the goal is to avoid a decline in sustainability and resilience levels of the current system, which may give the participating system actors the trust to provide details, expose interrelatedness between sustainability domains, and also come up with solutions. Regarding the latter, it should be noted that avoiding exceedance of critical thresholds does not automatically imply that a system is steering away from mediocre performance. This is why after assessing critical thresholds, participants should also be stimulated to think about adaptations to improve their system to desired sustainability and resilience levels (Chapter 5). Be it by steering away or actual exceeding critical thresholds to arrive at higher sustainability levels. Chapter 5 suggests a back-casting approach, but other solution-oriented methods such as participatory multi-criteria decision analysis may also be appropriate (Belton and Stewart, 2002). In any case, starting with a threshold assessment before solutionoriented participatory methods may create path-dependency, resulting in adaptations that lead to a reconfirmation of the current system where a transformation might actually be more appropriate. This path-dependency is likely to be reinforced by only inviting participants from within the farming system. Farming system actors are for instance probably biased regarding depopulation and a loss of attractiveness of the rural area, as it is related to farm closure. Considering the possibility that the closure of individual farms could be good for the farming system as a whole might go beyond the mental models of some farming system actors. Participatory methods involving so-called "critical friends" that have no direct stake in the system might help to overcome this obstacle (Enfors-Kautsky et al., 2018). Involving external actors is especially required in unsustainable systems that persist through the agency of only a subset of stakeholders.

It should be noted that critical thresholds are never static as they depend on the context (Kinzig et al., 2006; Resilience Alliance, 2010). The need for labour, for instance, depends on the level of automatization in agriculture. Critical thresholds may change because of slowly changing variables (Kinzig et al. 2006 citing Carpenter et al. 2003), which is also acknowledged in this study by presenting interacting thresholds across levels and domains in multiple case studies. Different domains could be addressed by including a variety of social, economic, institutional and environmental *challenges, function indicators* and *resilience attributes*. Using the framework of Kinzig et al. (2006) forced in particular researchers in some case studies to reflect on critical thresholds in the social domain, while focus of participants was more on economic and environmental processes. The framework of Kinzig et al. (2006) can hence show where knowledge of stakeholders is limited. This is an asset as exposing the limits of local knowledge is often lacking in participatory settings (Mosse, 1994). Explicitly adding the institutional domain and a level beyond the farming system to the framework of Kinzig et al. (2006) may further reveal the limits of knowledge and improve the understanding of farming system dynamics. To further stimulate co-production of knowledge,

the figures with interacting thresholds (e.g. Figure 6.1) could be fed back to farming system stakeholders in a follow-up workshop. In addition, farming system actors could be stimulated to think about representative indicators for resilience attributes. These representative indicators could add local meaning and thus improve stakeholders' understanding and assessment of the resilience attributes and resilience mechanisms (see also Chapter 5).

Becoming aware about a threshold can help reducing the likelihood of exceeding one (Resilience Alliance, 2010). Indeed, assessing critical thresholds may bring the awareness that is needed to move away from the conditions that have caused them. Participatory methods that are more specifically aimed at social processes could bring about awareness of system actors. However, interrelatedness with processes in other domains are consequently likely to be lost out of sight. Still, specific attention for social processes in the conducted workshops can improve the integrated nature of the assessments, for instance by pre-selecting at least one indicator related to a social function and a resilience attribute related to social conditions. For some case studies in this study, this would imply a suggestion that new functions and system goals are needed. Although top-down, this could initiate the process of system actors picking up this signal as being valuable (belief formation) and the process of redirecting the system as a whole to an alternative state (conversion; Biesbroek et al. 2017).

The study presented in this paper is a resilience assessment that is partly objectively and partly subjectively defined: we worked with a set of *function indicators* and *resilience attributes* selected in a previous workshop by stakeholders based on lists prepared by researchers (Chapters 2 & 3). Such an approach may not be feasible at European scale, but has proven effective for postulating candidate indicators for monitoring frameworks such as the CMEF. More participatory workshops in a diverse range of European farming systems are advised to find more of these indicators that can enrich those monitoring frameworks. It should be noted however, that assessments inclining towards a subjective definition and evaluation of resilience are poorly researched and that translation issues and cultural biases can limit these kind of assessments (Jones, 2019). Further elaboration and study of participatory methodologies is therefore necessary to improve its use for evaluating sustainability and resilience at farming system, national and European level. Specifically the desired or acceptable degree of objectivity vs. subjectivity in assessments across different levels (field, farm, farming system) and domains (economic, environmental, social) should be discussed.

6.6 Conclusion

In our participatory approach, all 11 studied systems in the European Union were perceived to be "close to", "at or beyond" at least one identified critical threshold (Tables 6.1-3). In particular, critical thresholds in the economic domain were considered to be (almost) reached. This could explain the economic orientation of farming system stakeholders and the current CMEF of the CAP. Overall, a strong decline in system performance was expected if critical thresholds would be exceeded. We conclude that concern for exceeding critical thresholds is justified, even though precise determination of a threshold position based on a participatory approach is difficult. Stakeholder perceptions on critical thresholds provide useful information as they serve as a stress signal and can be used as a starting point for a dialogue with farming system actors. We suggest that critical thresholds could be seen as a "thresholds of potential concern" for which management and policy goals may be developed. For instance, policies to attract more agricultural workers to an area to avoid a shortage of labour. Those policy and management goals should include the development of metrics that provide rigorous information on that specific threshold. The analysis of critical thresholds provides a basis for early thinking about possible alternative configurations of the systems. In this regard, the results can be used to reflect collectively about farming system trajectories, as to system functions and the often-competing goals of the different stakeholders. Therefore, the results of the analysis can be used to develop a contextualized, shared vision and to identify, within each farming system of interest, where to focus regarding increasing the resilience and sustainability of the farming system.

Critical thresholds were perceived to interact across levels of integration (field, farm, farming system) and domains (social, economic, environmental) in all case studies (Figure 6.1). Common across case studies was the central role of economic performance at farm level, which was mainly affected by price levels and

yield levels. This is another confirmation of the importance of economic indicators in the CMEF. However, in all case studies, exceeding the critical threshold of economic performance at farm level was associated with social issues such as lower attractiveness of farming, lower availability of successors or farm exit. In some farming systems, these social consequences were also experienced as critical thresholds contributing to lower labour availability reinforcing the low economic performance or contributing to depopulation, which encourages the loss of attractiveness of farming. This reinforcing effect may speed up the erosion of resources in the social domain. Social indicators are therefore important to consider when assessing the sustainability and resilience of farming systems.

A recurrent theme in our discussion section is the importance of system resources for stimulating sustainability and resilience of farming systems. For instance with regard to creating buffering capacities, building *resilience attributes* or finding the means to implement resilience enhancing strategies. We therefore stress the need to include system resource indicators such as soil quality, habitat quality, knowledge levels, attractiveness of rural areas and general well-being of rural residents when monitoring and evaluating the sustainability and resilience of European farming systems. In cases of low farming system resilience, building system resources may initially depend on actors in the institutional domain outside the farming system. In case of economic subsidies, these should be increasingly conditional on the environmental and social functioning of farming systems.

Supplementary Materials

Supplementary Materials 6.1: <u>https://ars.els-cdn.com/content/image/1-s2.0-S0743016721003223-</u> mmc1.docx

Supplementary Materials 6.2: <u>https://ars.els-cdn.com/content/image/1-s2.0-S0743016721003223-</u> mmc2.xlsx

Supplementary Materials 6.3: <u>https://ars.els-cdn.com/content/image/1-s2.0-S0743016721003223-</u> mmc3.docx

General discussion

7.1 Objectives and design of the study

The aim of this thesis is to operationalize the SURE-Farm resilience framework (Meuwissen et al., 2019) with new and (semi-)quantitative methods and to assess the sustainability and resilience of current and future European farming systems. The following research questions are central in this thesis:

- Is there a balance between social, economic and environmental functions in European farming systems in terms of importance and performance?
- Are European farming systems approaching critical thresholds?
- What resilience capacities do and should European farming systems have?
- What strategies enhance sustainability and resilience of European farming systems?

In this chapter I first discuss the research findings. I start with outcomes in relation to the research questions that are common across case studies and subsequently zoom in on specific case studies. This part of the discussion is largely based on SURE-Farm deliverable 5.7, which concerns a policy brief that I co-authored as second author (Reidsma et al., 2021). Based on the integrated work presented in this thesis I can provide a plausible narrative about current and future sustainability and resilience of European farming systems. This plausibility is primarily based on, and hence limited by, the case studies and the level of detail involved.

In the second section I reflect on the strengths and limitations of the resilience framework and methodology used in this thesis. In the third section I discuss the relevance and implications of my work in a wider context. In the fourth section, I provide recommendations for assessing sustainability and resilience of European farming systems. I end with the main conclusions of my work.

7.2 Sustainability and resilience of European farming systems

7.2.1 Sustainability and critical thresholds

In the studied farming systems there was generally an emphasis on food production and economic viability, where the maintenance of natural resources was often perceived to come third in terms of importance (all chapters). The sustainability of the farming systems was generally perceived to be moderate (Chapter 3). Moreover, most farming systems were perceived to be close to economic, environmental and/or social critical thresholds (Chapter 6). A variety of challenges, such as extreme weather events, low prices, price volatility, high production costs and continuous change of laws and regulations, pushed many farming systems to the limits. In addition, critical thresholds were perceived to interact across system levels (field, farm, farming system) and domains (social, economic, environmental), with low economic viability leading to lower attractiveness of the farming system, and in some farming systems making it hard to maintain natural resources and biodiversity.

Economic viability is perceived to be a central function of farming systems (Chapters 2, 3, 5 & 6), but in many systems it is perceived to be close to its critical threshold (Chapter 6). Food production needs to be high to ensure economic viability. Moreover, past strategies (e.g., increasing farm size and intensity) often focused on improving production and economic viability (e.g. Chapter 2). In Chapter 6 I argued that these functions are part of fast system processes and that concerns about more immediate stress-signals from these processes result in trade-offs with slower processes related to public goods (e.g., improvement of soil quality and social well-being in the farming system). These trade-offs could reinforce the already strong emphasis put on food production and economic functions by stakeholders (Chapters 2 & 3).

Accumulating challenges cause some systems to be close to an undesired decline, such as in the case of the extensive sheep production in Huesca, Spain, where a vicious circle leads to low economic returns, low attractiveness of the sector and abandonment of pasture lands (Chapter 5). In other systems, such as the starch potato system in the Veenkoloniën, the Netherlands, innovation and self-organisation have, so far, prevented the system from an undesired decline, but nematode pressure, droughts, high production costs and stricter regulations keep the system close to critical thresholds (Chapter 6; Herrera et al., 2022).

7.2.2 Current resilience and strategies

Based on an assessment of the presence of resilience attributes and strategies implemented in the past in 11 European farming systems, resilience was judged to be low to moderate (Chapters 2 & 3). Arable systems and horticultural systems scored relatively low, while the high-value equ and broiler system in southern Sweden, and the small-scale hazelnut system in Viterbo, Italy, scored higher. In Sweden, this was mainly because of production and legislation being well-coupled to local and natural capital, while in Italy, the system was reasonably profitable and socially self-organized (Nera et al., 2020). In most farming systems, presence of resilience attributes and historical dynamics of main functions suggest a certain robustness, which prevails over transformability (Chapters 2 & 3). In this context, adaptability is mostly employed for keeping stability and realizing (slow) incremental improvements. The focus on robustness is reflected in the policies in place for the studied farming systems (Buitenhuis et al., 2020b, 2020a; Feindt et al., 2019), while farmers in the studied farming systems perceived themselves as having slightly more adaptability than robustness (Spiegel et al., 2021, 2020). In Chapter 2 I argue that the focus on robustness fits well with a control-rationale that is focussed on efficient production in a stable context (Hoekstra et al., 2018). However, the economic and environmental context is not stable and expected to become increasingly instable. Moreover, given the perceived low to moderate sustainability and resilience of current farming systems, in particular regarding their closeness to critical thresholds, adaptation or even transformation seems necessary. A shift towards a resilience rationale seems, therefore, required for most of the farming systems studied in this thesis. The self-reported adaptability of farmers (Spiegel et al., 2021) is an essential capacity at farm level that may induce a shift in thinking at farming system level. provided the adaptability is not primarily employed for increased robustness.

Past strategies to cope with challenges were often geared towards maintaining profitability, such as intensification and scale enlargement (Chapter 3). Such past strategies were to a lesser extent geared towards building the important resilience attributes related to social self-organization, infrastructure for innovation, response and functional diversity, and coupling production to local and natural capital (Chapter 3). These strategies kept farming systems robust, but there are limits to the success with regards to increasing farm size and intensity (Chapters 3 & 5). In the extensive sheep production system in Huesca, Spain, for instance, further expansion of farms was not feasible due to labour shortage (Chapter 5). Systems focused on production and economic functions may seem to enhance resilience in the short-term, but they negatively affect other resilience attributes and deteriorate resilience in the long-term (Chapters 3 & 6). In the hazelnut production system in Viterbo, Italy, for instance, this is acknowledged and new strategies to improve sustainability and resilience are directed towards self-organization at the farming system level (Chapter 2; Nera et al., 2020). Strategies aiming for long-term sustainability and resilience thus need to consider how to nurture environmental and social dynamics that are needed to sustain and enhance economic viability of farming systems (e.g., natural resources and labour are needed to maintain profitable yields). Hence, there is a clear need for alternative systems with a balanced attention for economic, social and environmental dimensions.

7.2.3 Future sustainability, resilience and strategies

Assessments on alternative systems are in this thesis only elaborated for the extensive sheep production system in Huesca, Spain (ES-Sheep; Chapter 5). Many of the observations for ES-Sheep are similar to the observations done in other farming systems, using the same participatory approach (Accatino et al., 2020). For instance, the balance between social, economic and environmental functions was not always kept by stakeholders when envisaging alternative systems (Chapter 5; Accatino et al., 2020). Envisaged alternative systems in ES-Sheep included a focus on intensification, specialization, technology, product valorisation, collaboration and an attractive countryside (Chapter 5). In other farming systems, alternative systems also included a focus on diversification and organic and/or nature-friendly farming (Accatino et al., 2020). Each type of these alternative systems may enhance the performance of some system functions, but, in the end, all functions need to perform at adequate levels. Keeping in mind the interactive nature of system functions and attributes simultaneously. Technological innovation is required, but should be accompanied with structural, social, agro-ecological and institutional changes (Chapter 5; Accatino et al., 2020). In ES-Sheep and other studied farming systems, stakeholders indicated that particularly enabling conditions in

the social and institutional domains were lacking to make change happen (Chapter 5; Accatino et al., 2020). Enabling conditions that should be addressed in ES-Sheep and other systems are the access to knowledge, more effective bureaucracy, improving the consistency and transparency of policies and regulations, and providing compensation for the delivery of public goods (Chapter 5; Accatino et al., 2020). From the above it becomes clear that all actors inside as well as outside the farming system need to be involved to realize more sustainable and resilience farming systems.

Obviously, not all farming systems can implement all changes as presented in the previous paragraphs. In particular the stability of the farming system itself and its context should play a role in implementing changes. Depending on the stability of the context, I assume in Chapter 2 that agricultural systems are/move somewhere on the gradient between having a technological control rationale in a stable context and an ecological resilience rational to deal with contexts of uncertainty (Hoekstra et al., 2018)⁴. Many contemporary, intensive and specialized systems seem to be too much on the "control" side, i.e. the degree of technology use has improved production, but social and environmental sustainability limits are approached (Chapter 6), while the economic and environmental context is increasingly variable. In these systems, for instance, supporting nature-based pest management through diversification is likely to reduce pesticide use, but a trade-off with (control over) yield may occur (van der Werf and Bianchi, 2022). On the other hand, very extensive systems may benefit from more technologization and control as it can help to do "better than nature" (Ford Denison and McGuire, 2015) or help the co-existence of agriculture with wild-life (e.g. placing electric fences to avoid wolf attacks on sheep; Chapter 5).

The capacity of farming systems to move along the control-resilience continuum is important for a broader discussion on agricultural sustainability. From a sustainability perspective, literature suggests that agricultural performance at national or continental level can be improved when different types of farming systems, including more technology-based intensive as well as more ecological-based extensive systems, are spatially divided (Accatino et al., 2019; Bakker et al., 2021; Schulte et al., 2019). However, from a resilience perspective, concentration of specific farm types in one region would also reduce the farm heterogeneity, which is considered an important resilience attribute (Chapters 2 & 3; Cabell and Oelofse, 2012). The costly process of spatial planning and re-allocation, therefore, should explicitly consider the sustainability and resilience of farming systems in combination with the stability of the economic, social and environmental context.

7.3. Methodological strengths and limitations

7.3.1 Framework

Farming system

The flexible definition of a farming system regarding the geographical delineation allowed for study cases that were different in size and number and types of actors. Yet, during the cross-case study comparison, differences in culture, size of the farming system etc. made it difficult to compare case studies directly (Chapter 2). Instead, I compared common patterns across case studies, e.g. regarding important resilience attributes in Chapter 2 and critical thresholds in Chapter 6. The definition of a farming system that was used also provides flexibility when looking at a single case study: the system can be defined at any level at which emergent properties are expected regarding a topic of interest. As such, (farming) system research should not be presented as a particular study domain, but rather as a way of thinking that supports defining a problem in a multi-stakeholder setting or support the scaling of innovations. Parts of the framework, for instance, already shaped researchers' thoughts, beyond the SURE-Farm project, on studying the link between financial assistance for drought-affected agriculture and resilience in tropical Asia (Goodwin et al., 2022) and on studying the response and resilience of agri-food systems in twenty-five Asian countries (Dixon et al., 2021).

⁴ This also relates to the concept of adaptive cycles: in a growth/conservation phase (in a relatively stable environment), a control strategy (focussed on robustness and efficiency) works, but in a decline/reorganization phase a resilience rationale (focussed on adaptability and transformability) is needed.

Farming systems research emphasizes the production aspect of agriculture, rather than on the consumption aspect. Work on agricultural resilience is often embedded in the notion of food systems (e.g. Ericksen, 2008; Prosperi et al., 2016; Tendall et al., 2015), which addresses both the production and consumption aspects related to agriculture. Food systems include, beside farming systems, the whole value chain of agricultural produce, including, amongst others, food processors, super markets and consumers (Berkum et al., 2018; von Braun et al., 2021). Within SURE-Farm work, it became clear that these actors are indeed important to improve farming system sustainability and resilience (Chapters 2 & 5; Meuwissen et al., 2020). Consumers, for instance, are important actors that determine the demand for food. By implicitly placing food demand outside the farming system boundary it could be perceived as a factor that cannot, or even should not, be influenced or controlled. However, to stay within planetary or regional boundaries for the environment and human well-being, changing food demand through changing consumption patterns is part of the solution space (Springmann et al., 2018).

Challenges

In the framework, there is a focus on challenges affecting the system negatively, rather than as an opportunity for positive system change. In the resilience literature, a pending system change is often seen as an opportunity (e.g. Westley et al., 2011) for which preparations can be taken (Enfors-Kautsky et al., 2018). Challenges as an opportunity originate in the metaphor of the adaptive cycle that has been shortly discussed in the discussion of Chapter 6. The idea of seeing challenges as an opportunity is that after a crisis, in which part of the system resources have been released, actors could use their agency to influence the re-organization phase for the benefit of sustainability and resilience. Walker et al. (2020), for instance, identified opportunities for improved social and environmental sustainability at global level after the COVID-19 crisis. In agricultural research the notion of opportunities after unexpected events has also appeared (Darnhofer, 2021).

Realizing desired change through a crisis requires a thorough understanding of the system regarding its main values, variables and drivers (Walker et al., 2020) as well as an understanding of the agency of individual actors in social-ecological systems (Westley et al., 2013). Based on the work presented in this thesis, such a claim cannot be made. In contrast, in Chapter 6, it is pointed out that critical thresholds whose exceedance precede a release of a system are very difficult to assess. Moreover, the exceedance is expected to greatly negatively impact the sustainability and resilience of farming systems (Chapter 6). It should also be noted that the adaptive cycle, which provides the origin of seeing challenges as opportunities, is a metaphor, i.e. so far there is very little empirical evidence that could support a working model for the adaptive cycle in socio-ecological systems in general (Cumming and Collier, 2005; Cumming and Peterson, 2017), or in agricultural systems in specific (e.g. van Apeldoorn et al., 2011). After all, considering the uncertainties and stakes, a risk avoiding strategy relating to challenges seems the best option in order to avoid a forced system transformation. However, unexpected events are increasingly common in a globalized world (Darnhofer, 2021). These events may push farming systems beyond critical thresholds. In such situations, adaptability is needed. Based on the findings in this thesis, farm heterogeneity and experimentation at levels below the farming system are recommended (Chapters 2 & 3). These resilience attributes could help to have production technologies available in case critical thresholds are exceeded, i.e. improve adaptability and transformability at farm and farming system level. To benefit from unexpected change at farming system level, further elaboration of theories, conceptual frameworks and methods seems necessary (Darnhofer, 2021).

Functions

The list of eight functions proofed practical: they were easily recognized and acknowledged by participants and representative indicators were relatively easy to determine. This provided a good base for discussion with farming system stakeholders. However, when it came to evaluating the importance and performance of functions, participants sometimes argued that all functions were connected and therefore that an evaluation was difficult (e.g. Chapter 2).

In the context of this thesis, functions were primarily categorized in the economic, environmental or social domain. Alternatively, a categorization according to ecosystem services (Table 7.1) could have been

interesting: existing resilience theories are mainly based on resilience studies that are primarily focussing on the ecological functioning of socio-ecological systems (e.g. Bennett and Peterson, 2005; Carpenter et al., 2005) and, more importantly, can be strongly embedded in the ecosystem services framework (e.g. Biggs et al., 2012; Schlüter et al., 2015). However, some of the proposed functions by Meuwissen et al. (2019) are difficult to categorize according to ecosystem services (Table 7.1). Moreover, apart from providing ecosystem services, farming system functioning can also be associated with **dis**services provided **to** ecosystems (Huang et al., 2015). Indeed, the categorization in economic, social and environmental domains, can, without confusing stakeholders, include notions of agricultural disservices by means of addressing trade-offs between different domains, e.g. between food production (economic) and biodiversity & habitat (environmental)(e.g. Chapters 5 & 6)⁵.

Function	Description	Domain	Type of ES	
	Private goods			
Food production	Deliver healthy and affordable food products	Economic	Provisioning	
Bio-based resources	Deliver other bio-based resources for the processing sector	Economic	Provisioning	
Economic viability	Ensure economic viability (viable farms help to strengthen the economy and contribute to balanced territorial development)	Economic	Cultural?	
Quality of life	Improve quality of life in farming areas by providing employment and offering decent working conditions	Social	Cultural?	
	Public goods			
Natural resources	Maintain natural resources in good condition (water, soil, air)	Environmental	Supporting	
Biodiversity & habitat	Protect biodiversity of habitats, genes, and species	Environmental	Regulating	
Attractiveness of the area	Ensure that rural areas are attractive places for residence and tourism (countryside, social structures)	Social	Cultural	
Animal health & welfare	Ensure animal health & welfare	Environmental	Regulating? Cultural?	

Table 7.1. System functions considered in the resilience framework of Meuwissen et al. (2019) and linkage to the eco-system services (ES) framework that distinguishes provisioning, regulating, supporting and cultural ES.

Robustness, adaptability and transformability

Resilience thinking needs metaphors in the absence of complete understanding of a system's potential behaviour. Robustness, adaptability and transformability and their accompanying metaphors of balls in stability landscapes (Figure 1.2) proved useful for farming systems. Using these three resilience capacities may suit the thinking process of farming systems actors better than the more complicated metaphor of

⁵ The categorization in terms of domains instead of ecosystem services is an important adaptation for studies on the resilience of farming systems (having a strong technological component) compared to studying resilience studies on ecosystems.

(interacting) adaptive cycles (See also the discussion under the sub-section "Challenges" in this section). The metaphor of adaptive cycles is best used by researchers in the evaluation phase. For instance to determine the position in the adaptive cycle (Reidsma et al., 2019b) and the capacities present in the system to deal with the different phases in the adaptive cycle (Cabell and Oelofse, 2012; Le Goff et al., 2022; Tittonell, 2020).

Yet, it should be noted that the distinction between adaptability and transformability was sometimes difficult to make in participatory settings (Chapter 2). Although the concepts of adaptability and transformability could be merged for the sake of convenience (Chapter 2), I think this should not be done so easily. Addressing all three capacities in the studied systems really pointed at something that was lacking: transformative capacity. Considering this lack, notions of a system's identity (Cumming and Collier, 2005; Cumming and Peterson, 2017) should be used with care in farming systems research. Marshall et al. (2012), for instance, mention that identity, in the form of attachment to an area or occupation, may be positive for adaptation to incremental change, but reduce transformational capacity. Identity, when based on history, may also result in the exclusion of much needed new actors that can help transforming the system, as they don't belong in the original picture (Enfors-Kautsky et al., 2018).

In this thesis, some farming systems seem rather unsustainable and therefore in need of transformation (e.g. ES-Sheep; Chapter 5). In Chapters 2 & 6 it is already mentioned that mere adaptations sometimes do not seem enough, but that lock-in interests of farming system actors can impede possible transformations. Indeed, power issues play an important role, while they have been scarcely addressed in this thesis. I already mentioned the "critical friends" approach to break the lock-in (Chapter 6). Critical friends are important to question and reframe dominant agendas for change, but complementary actors may be needed (Chambers et al., 2022). For instance, actors that can navigate conflicting agendas or elevate marginalized agendas for change (Chambers et al., 2022).

Resilience attributes

In this thesis I adapted an existing list of resilience attributes. This resulted in 22 attributes from which 13 were selected and discussed in participatory workshops (Chapters 2, 3, 5 & 6). The 13 resilience attributes were initially categorized according to five resilience principles selected by Meuwissen et al. (2019) based on previous resilience research on socio-ecological systems (e.g. Walker and Salt, 2012). During the analyses and interpretation of workshop results, a categorization according to system resources in the form of different kind of capitals (e.g. financial, social) turned out to be more intuitive for farming systems.

It is challenging to determine what would be the right set of resilience attributes for farming systems: the importance of resilience attributes turned out to be context and stakeholder dependent (Chapters 2 & 3). One could also think of additional resilience attributes related, for instance, to the remoteness of the area (Chapter 6), foresight capacities (e.g. relating to policy changes) and the type of product (e.g. fresh or long lasting, seasonal or available throughout the year). From a practical point of view, discussing 13 resilience attributes in a participatory workshop is already challenging. In such a setting, including more attributes is not recommended, so selection becomes important.

A generic advice is to make sure that all different forms of capital are represented by the resilience attributes that will be evaluated for a farming system. That way it resonates with the balanced attention for system functions in the economic, social and environmental domain. Another generic advice is to make sure that all resilience attributes relate to the same level of analysis. In the case of European farming systems, it is recommended to include the resilience attributes related to being reasonable profitable, production being coupled with local and natural resources, farm heterogeneity, social-self organization and infrastructure for innovation, as these were perceived important in most case studies (Chapters 2 & 3). In particular, being reasonably profitable and having infrastructure for innovation were perceived important for transformability (Chapters 2 & 3). Further research is necessary to identify the much-needed resilience attributes for improved transformability. In the case of farming systems, experimentation could be such an attribute (Chapter 2).

7.3.2 Participatory methods

The participatory methods that were developed and used in this thesis were supported by the resilience framework. This yielded a very structured and rather exhaustive, but still flexible, approach regarding the presentation of functions and resilience attributes covering the economic, social and environmental domain. As discussed in Chapters 2 & 5, this resulted in knowledge exchange from researchers to participants. The structure provided by the framework also stimulated the knowledge exchange from participants to researchers regarding the farming system-specific actors, challenges, function indicators, and, in the case of ES-Sheep, local indicators for resilience attributes (Chapter 5).

The participatory methods had multiple limitations. I see the stakeholder selection and participation reducing the representativeness and reproducibility of the research as the main limitation for getting an objective and comprehensive picture of a farming system (Chapters 2, 5 & 6). As a consequence of the selection process, radical thinkers were largely absent, which limits the likelihood of transformative change in cases where current sustainability and resilience is low (Chapter 5). As mentioned before, multiple actors with different skill sets or personality traits may be needed to quide transformation (Chambers et al., 2022). In addition, researchers themselves could play an active role in guiding transformations, i.e. move beyond the more or less neutral role of the "honest broker" (Wittmayer and Schäpke, 2014). This implies that researchers become politically involved and explicitly advocate for values important for society in addition to scientific values and standards (Pielke, 2007). In general this leads to a vague distinction between knowledge and ethics and between (scientific) guality and (societal) truth, i.e. a post-normal understanding of science (Ravetz, 2002). In particular the combination of (scientific) uncertainty and (societal) urgency surrounding the topic of sustainability and resilience of agricultural systems justifies such a development. It should be noted, however, that a post-normal understanding of science in transdisciplinary research requires researchers to navigate important and urgent, but potentially conflicting, demands from science, society and the self (Sellberg et al., 2021). In particular young scientists in transdisciplinary studies should be aware of this in order to engage sustainably at a professional, political and personal level (Sellberg et al., 2021).

Overall, the methodology can be regarded as relatively quick, interactive and interdisciplinary, providing ample information on critical thresholds, current system dynamics and future possibilities. The participatory workshops supported the construction of joined visions for the farming systems, at least regarding relevant strategies and enabling conditions for change. Altogether, the participatory approach used in this thesis leans towards the descriptive and explanatory, while exploration and (re-)design are hardly addressed (see e.g. Giller et al., 2008). Related to this, the approach is more leaning towards information extraction rather than rapport building with local stakeholders (Mosse, 1994). To become more complete as a participatory approach, additional activities and tools need to be developed that direct towards a joint vision and action plan for change. For such a vision and plan, the right set of goals, strategies, resources and actors involved need to be determined (Mathijs and Wauters, 2020). Existing approaches, such as the Wayfinder approach (Enfors-Kautsky et al., 2018), could be used for inspiration, provided it is first adapted to the context of farming systems.

7.3.3 Quantitative methods

The quantitative method to assess farm resilience in Chapter 4 was functional for filtering high-dimensional data but was limited for empirically confirming the resilience attributes in agricultural systems. Empirical evidence based on quantitative data remains important to check the assumptions of generic resilience theory in an agricultural context, e.g. to assess the role of diversity (Dardonville et al., 2020) and intensity (Dardonville et al., 2022a). Fortunately, there are multiple methods available for quantitative resilience assessments based on empirical data (see e.g. Dardonville et al., 2021; Gil et al., 2017). Within the SURE-Farm consortium, for instance, Slijper et al. (2021) studied the effect of different forms of payments on the resilience capacities of European farms. They found negative and positive effects of payments on robustness, while no effect was observed for adaptability or transformability (Thomas Slijper et al., 2021).

Gil et al. (2017) point out that, when using statistical methods, the real cause-effect relation is often not found. In addition, these methods are ill-equipped to deal with social risks as these are difficult to quantify.

Hence, a statistical method integrating economic and environmental farming system functions, let alone integrating social functions, is still not available (Chapter 4). Based on my experiences with the analyses in Chapter 4, I have come to the conclusion that multiple statistical analyses with a single response variable are recommended over integrating multiple response variables in one analysis. Quantitative trade-off analyses should, therefore, be conducted with other types of methods such as dynamic simulation models (e.g. Herrera et al., 2022; Pinsard et al., 2021) and optimization models (e.g. Shi et al., 2021). It should be noted, however, that these alternative methods are often not empirical and therefore their practical relevance is questioned (Klapwijk et al., 2014). Complementary participatory approaches, such as the methods presented in this thesis (Chapters 2 & 5), have the potential to improve the practical relevance of models in trade-off analyses (Klapwijk et al., 2014).

Models representing system behaviour over time are important in the context of resilience (Herrera et al., 2022). In particular systems thinking theory and subsequent system dynamics modelling could provide the right context to integrate different kinds of quantitative and qualitative knowledge (Sterman, 2000). Based on these knowledge sources, system dynamic models can be used for simulating different scenarios regarding, for instance, the implementation of new technologies and policies, but also changing values of system actors. In some first attempts to apply system dynamics modelling in a qualitative way, I noticed that by including different, interacting framework indicators (e.g. functions indicators, resilience attributes), sustainability and resilience get mixed up (Accatino et al., 2020). On the one hand this operationalizes sustainability and resilience as complementary concepts. On the other hand, it may prevent researchers and stakeholders to get a quick overview of important system variables, resulting in a reduced interest in the subject. Systems dynamics modelling should, therefore, always be conducted within the context of a framework, such as the one used in this thesis, that filters and categorizes the most important system variables.

7.4 Relevance in a wider context

In Chapter 1 I argued that the global/European production system has low sustainability and decreasing resilience, which, as I argue, is intrinsically linked with the design of the global Economic system. This image has been confirmed by the research chapters in this thesis. An important finding for the studied farming systems was the importance of a balanced attention for the social, economic and environmental domain. This balanced attention stretches beyond the boundaries of the farming system. In this thesis we have already seen the focus on economic performance at individual farm level (Chapter 2) and the tendency to forget about social indicators and sometimes environmental indicators when there is a plenitude of economic indicators (Chapter 5). At a global level, something similar is happening due to the dominant role of markets in guiding human activity (Raworth, 2017). For instance with regard to an overemphasis on economic growth in the form of gross domestic product.

Raworth (2017) points out that, besides markets, the three other pillars supporting human well-being and welfare should be considered equally: the households, the government and the collaborative management of the commons, i.e. a large share of the formal and informal institutions that operate mainly in the social and environmental domain. In the context of sustainability and resilience of farming systems, farm households are indeed very important regarding supplying an unpaid or cheap, yet motivated, knowledgeable, responsive and flexible labour force that is willing to work long hours in order to stay economically viable (van Vliet et al., 2015), being a source for generational renewal (Coopmans et al., 2021; van Vliet et al., 2015), influencing decision making (Urquhart et al., 2019) and providing quality of life to farmers, farm workers and rural communities (van Vliet et al., 2015). Without trying to idealize the concept of family farming, it can be concluded that the family farm household is of particular importance for the current functioning of farming systems and the wider food system. In this context, he decline of family farms and the rise of large scale farm corporations (e.g. Giller et al., 2021; Nyström et al., 2019) seem developments that could remove crucial actors for making farming systems more sustainable and resilient.

In general, the national government is an actor that can take risks that no other actor can take, which makes it an important actor for creating welfare and change, in particular through innovation (Mazzucato,

2018). Although the role of government is not exactly clear yet regarding farming systems (and probably never will be), it has become clear in this thesis (e.g. Chapters 3 & 5) that farming system actors cannot make the necessary changes alone (lack of agency). There is a need for actors outside the farming system to change the economic and institutional context (changing the structure) in order to allow for change. Unified, collaborating national governments at European level could partly realize this. Hopefully, the full implementation of the EU's new green deal and it's Farm to Fork strategy will be an example of this. Especially when realized under the current and, to a large extent, unexpected global challenges: a persistent pandemic, rising geo-political tensions and unprecedented market dynamics.

When thinking about the commons as important for farming system functioning. I still sometimes believe that this is a bit farfetched, as farms mostly operate individually and are to a considerable extent detached from their local biophysical environment (while in most cases actually being considered as land-based systems). This, from a resilience and sustainability perspective, seems exactly the issue; a low degree of self-organization and a low degree of production coupled to local and natural capital, i.e. low degrees of two important resilience attributes. The commons, i.e. that what we share and manage together for our welfare and well-being, comprise mainly the social and environmental environment. These are the domains that are most easily neglected as has become clear in this thesis. Fortunately, previous research on the use of commons has provided an overview of boundary conditions and rules to prevent the erosion of common resources (Ostrom, 1990). Apart from ecological commons, other types of commons could be considered, such as knowledge, digital and cultural commons. In fact, the understanding of what commons really are, how to manage them and how important they are for human well-being could be interpreted as a cultural common. Doubting the important role of commons in European agriculture, as I did at the start of this paragraph, could even be seen as an indication that this common is actually eroded. A boundary condition for its comeback is a less dominant role of markets and their main actors in economies. Identifying and monitoring environmental and social commons would be an important first step towards a more prominent role and recognition of commons for the creation of welfare and well-being. In the "Further research" section below, I, therefore, put forward a basic structure of a policy monitoring and evaluation framework, for instance for the Common Agricultural Policy (CAP) of the European Commission, that can help stakeholders and governments realize what their commons actually are and how to monitor them.

7.5 Further research

7.5.1 Attention for resources in future resilience research and policies

As discussed in Chapter 6, having adequate resources in all domains seems essential for improving system resilience attributes and dealing with challenges. Overall, in (agricultural) economics there is a large emphasis on inputs, throughputs and outputs, i.e. on system flows or system performance (Sterman, 2000). This is for instance very visible in the CAPs Common Monitoring and Evaluation Framework (CMEF) whose indicators primarily describe money flows, even regarding environmental objectives, while the status of relevant resources is hardly addressed. Emphasizing system flows leads to event-oriented cause-effect analyses that neglect crucial feedback loops in a system as the flows are hardly associated with the resources that they affect and are affected by (Meadows, 2008; Sterman, 2000). Elaborating further on this, Meadows (2008) emphasizes the buffering capacity of system resources and their importance as leverage points to realize change and resilience in systems. Further sustainability and resilience research on farming systems should, therefore, explicitly have attention for environmental, economic and social resources.

The resilience attributes used in this thesis provide a good base for including resources in the analyses and monitoring of farming systems (see also section 7.3.1). While applying systems thinking to the resilience framework used in this study, it can be argued that functions of a system relate more to its flows, where attributes relate more to its stocks/resources (Chapters 2 & 6). In that sense, attributes are a source of resilience, but also a manifestation of system flows over time, i.e. system sustainability. This is in fact the operationalization of sustainability and resilience as complementary concepts as proposed in the resilience framework (Chapter 6).

Taking this system's perspective of interacting functions and attributes, I elaborate why and how resources can be included in existing monitoring frameworks on sustainability and resilience. As an example I take the CMEF, the monitoring framework of EU's agricultural policy instrument, the CAP. The CMEF has been developed to measure the implementation and impact of the CAP while considering contextual factors (European Commission, 2015). The CMEF was primarily developed based on a sustainability perspective regarding economic growth, management of natural resources and socio-economic development of rural areas (European Commission, 2015). Although the previous editions of the CAP and also the new CAP (CAP 2021-27) aim to address environmental, economic as well as social aspects of European agriculture, the indicators in the current CMEF mainly cover monetary investments in the environmental and economic domain. Regarding the impact indicators, i.e. the main indicators to evaluate the effectiveness of the CAP, social indicators are completely absent. In addition, when addressed elsewhere in the CMEF, the indicators in the social domain are often expressed in economic terms, for instance as investment in rural facilities.

Fortunately, there are already a few examples of resource indicators in the current CMEF. For instance, the percentage of organic matter in top soils could be explicitly regarded as an environmental resource. Based on the work in this thesis, the resilience attribute describing the degree to which production is coupled with local and natural capital should be added as an environmental impact variable (Chapters 2 & 3). This will not be an easy task. Ongoing work of e.g. Dardonville et al. (2022b) is a starting point for (quantitative) indicators related to this resilience attribute. Current CMEF contextual indicators that may be distinguished as social resource indicators are the number of farms and agricultural training of farm managers. Based on the work in this thesis, the resilience attribute that describes the degree to which farming system actors are capable of self-organization should be considered as a resource and added as a social impact variable (Chapters 2 & 3). Asking farming stakeholders about perceived self-organization is a good start, but more objective metrics adapted to farming systems may be needed as well. Also the resilience attribute related to the level of infrastructure for innovation is an important resource that should be included (Chapters 2, 3 & 6). Work regarding Agricultural Knowledge and Information Systems could provide useful indicators for this resilience attribute (e.g. EIP-Agri, 2018). Other current CMEF indicators (Pillar I result indicators) that could be regarded as resource indicators are, for instance, structural diversity and crop diversity (currently still measured as amount of subsidies and area under subsidy). Based on the results presented in this thesis alone, I cannot confirm the importance of these indicators. Existing literature on the role of diversity in agriculture shows ambivalent results (Dardonville et al., 2020). Given the lack of diversity in multiple contemporary EU farming systems geared towards operating under a control rationale (Chapter 2), different forms of diversity should still be regarded as a potential solution for improving resilience. For instance crop diversity regarding economic buffering capacity for when a crop fails (Abson et al., 2013).

7.5.2 The complementary role of subjective and objective resilience assessments

A final observation regarding the indicators in the CMEF is that they are objectively defined and objectively assessed, i.e. they are defined and measured by actors outside the farming systems under study (Jones, 2019). Moreover, they are based on available quantitative data. This may be one of the reasons why social indicators are less covered in the framework, as they are generally hard to measure. Including local data and perceptions could overcome this shortcoming, provided that explicit attention is given to social indicators (Chapters 2 & 3).

Participatory approaches have the potential to include local data and perceptions and thus complement existing monitoring frameworks, such as the CMEF. This thesis has, for instance, shown that participatory methods can provide resource indicators in the economic, environmental and social domain that are relevant across a diverse set of farming systems in Europe (previous section; 7.4.1.). A subsequent question that arises is: what degree of influence of local actors should be sought when assessing farming system resilience? The objectivity-subjectivity continuum for assessing resilience as presented in Jones (2019) helps to reflect on this question (Figure 7.1). Jones (2019) distinguishes between *defining* and *evaluating* resilience. Both activities can be done subjectively, i.e. by actors inside, and objectively, i.e. by actors outside the farming system (Jones and d'Errico, 2019). *Defining* resilience relates to the choice of methods and indicators that are used to measure resilience. *Evaluating* resilience relates to the actual assessment of resilience according to the defined methods and indicators.



Figure 7.1. Suggested emphasis on objectively or subjectively defining and evaluating resilience indicators per domain and level of analysis. Source: combined from Kinzig et al. (2006) and Jones (2019).

I suggest, based on the reflections in the previous paragraphs, that from the environmental to the economic to the social domain, the desired degree of subjectivity in *defining* resilience should increase. This is in line with Van Calker et al. (2005) who, in the context of a sustainability assessment, suggest an increasing role for experts regarding ecological dimensions and an increasing role for stakeholders regarding social dimensions of farming systems. As mentioned in Chapter 6, farming system actors may not have the capacities or willingness to evaluate resilience at farming system level. For evaluating resilience, the degree of objectivity needs, therefore, to be higher at the farming system level, compared to farm and field level where most farming system actors operate (Figure 7.1). The emphasis on subjectivity or objectivity for evaluating resilience should also be dependent on the domain, with an increasing degree of subjectivity towards the social domain. This is partly a practical choice to improve the availability of data on the social domain. Overall, a diagonal arrow for evaluating resilience is presented in Figure 7.1. A good yield at field level, for instance, depends on the goals a farmer has, but can also be compared with potential and water-limited yields (Van Ittersum et al., 2013). At farm level, objectively evaluated farm income is justified as much as the subjective appreciation of a farmer for his current income: both actual income and personal appreciation of that income is important for a farmer's potential desire to guit or continue the farm. Also in the environmental domain, thresholds of soil organic matter at farm level, for instance, could be evaluated subjectively as well as objectively (Hijbeek et al., 2017). At farm level, it is also important to compare and consider objective (e.g. Slijper et al., 2021) and subjective (Spiegel et al., 2021) evaluations of resilience: large differences imply differences in perceptions of farming system actors and actors outside the farming system that need to be bridged. An emphasis on objectively evaluating resilience at farming system level in the environmental domain is required as feedbacks of agricultural practices may not be experienced at farm and field level because they are too slow or because the feedback is transferred to actors beyond the farming system (see e.g. Chapter 6), e.g. citizens or wildlife populations.

7.6 Conclusions

Based on the application of new and (semi-)quantitative methods developed in this thesis the following conclusions on the sustainability and resilience of European farming systems – and the methods to assess sustainability and resilience - can be drawn:

Current sustainability and resilience:

- European farming systems were perceived to have low to moderate sustainability and resilience (Chapters 2 & 3) and operate close to critical thresholds (Chapters 5 & 6).
- In the studied farming systems there was an overemphasis on (short-term) economic viability and a lack of attention for (long-term) social variables, while robustness was perceived to prevail over adaptability and transformability (Chapters 2 & 3).
- According to stakeholders, main building blocks for current resilience in most case studies were the
 resilience attributes related to having production coupled with local and natural resources,
 heterogeneity of farm types, social self-organization, reasonable profitability, and infrastructure for
 innovation. The latter two were perceived as particularly important for transformability (Chapters 2 &
 3).
- Past strategies of farming systems were often geared towards making the system more profitable, and to a lesser extent towards the other important resilience attributes (Chapter 3).

Future sustainability and resilience:

- For improving sustainability and resilience, future farming systems need a more balanced attention for economic, social and environmental domains, and an enabling environment. (Chapters 5 & 6).
- In terms of strategies, technological innovation is often required, provided it is implemented simultaneously with social, agro-ecological and institutional strategies that consider the long-term (Chapters 3 & 5).
- To implement such strategies, all involved actors inside and outside the farming system need to collaborate (Chapters 3 & 5).

Understanding and assessing farming system sustainability and resilience:

- Sustainability and resilience of farming systems remains a challenging topic, regarding its complexity
 in terms of detail (different domains, many concepts and variables) and dynamics (non-linearity,
 thresholds, interactions).
- The research presented in this thesis confirms the usefulness of the resilience framework in reducing this complexity through a step-wise approach tailored to farming systems.
- The participatory approaches presented in this thesis (Chapters 2 & 5) contribute mainly to describing and explaining sustainability and resilience of farming systems.
- These methods provide, therefore, a good basis for exploring future farming systems (Chapter 5).
- The quantitative approach (presented in Chapter 4) confirmed the impact of weather extremes on
 economic and environmental farm performance, but was limited in explaining resilience, and raised
 awareness about the influence researchers have on the results through the selection of response
 variables.
- Based on the work and reflections presented in this thesis I see scope for better understanding and assessing farming system sustainability and resilience through:
 - system thinking theory that explicitly includes the notion of common pool resources and the power and agency of all actors, i.e. including the household, government and market actors, inside and outside the (open) system boundaries; and
 - use of participatory integrated assessments including different levels and domains, complemented by objective and/or subjective approaches that are applied to a single domain and/or level of analysis. The level of subjectiveness of these complementary approaches should be dependent on the level and domain of analysis.
- System thinking theory and participatory integrated assessments should, therefore, be implemented in existing monitoring and evaluation frameworks, such as the Common Monitoring and Evaluation Framework of the Common Agricultural Policy.

References

- Abson, D.J., Fraser, E.D.G., Benton, T.G., 2013. Landscape diversity and the resilience of agricultural returns: A portfolio analysis of land-use patterns and economic returns from lowland agriculture. Agric, Food Secur. 2. 2. https://doi.org/10.1186/2048-7010-2-2
- Accatino, F., Paas, W., Herrera, H., Appel, F., Pinsard, C., Shi, Y., Schutz, L., Kopainsky, B., Bańkowska, K., Bijttebier, J., Black, J., Gavrilescu, C., Krupin, V., Manevska-Tasevska, G., Ollendorf, F., Peneva, M., Rommel, J., San Martín, C., Severini, S., Soriano, B., Valchovska, S., Vigani, M., Wauters, E., Zawalińska, K., Zinnanti, C., Meuwissen, M., Reidsma, P., 2020. D5.5 Impacts of future scenarios on the resilience of farming systems across the EU assessed with guantitative and gualitative methods.
- Accatino, F., Tonda, A., Dross, C., Léger, F., Tichit, M., 2019. Trade-offs and synergies between livestock production and other ecosystem services. Agric. Syst. 168, 58–72. https://doi.org/10.1016/J.AGSY.2018.08.002
- Alexandrov, A., 2010. Characteristics of single-item measures in likert scale format. Electron. J. Bus. Res. Methods 8, 1–12.
- Alkan Olsson, J., Bockstaller, C., Stapleton, L.M., Ewert, F., Knapen, R., Therond, O., Geniaux, G., Bellon, S., Correira, T.P., Turpin, N., Bezlepkina, I., 2009. A goal oriented indicator framework to support integrated assessment of new policies for agri-environmental systems. Environ. Sci. Policy 12, 562– 572. https://doi.org/10.1016/j.envsci.2009.01.012
- Allen, C.R., Birge, H.E., Angeler, D.G., Tony Arnold, C.A., Chaffin, B.C., Decaro, D.A., Garmestani, A.S., Gunderson, L., 2018. Quantifying uncertainty and trade-offs in resilience assessments. Ecol. Soc. 23, 3. https://doi.org/10.5751/ES-09920-230103
- Allison, H.E., Hobbs, R.J., 2004. Resilience, adaptive capacity, and the "lock-in trap" of the Western Australian agricultural region. Ecol. Soc. 9, 3. https://doi.org/10.5751/ES-00641-090103
- Anderies, J.M., Folke, C., Walker, B., Ostrom, E., 2013. Aligning key concepts for global change policy: Robustness, resilience, and sustainability. Ecol. Soc. 18, 8. https://doi.org/10.5751/ES-05178-180208
- Andersen, E., 2017. The farming system component of European agricultural landscapes. Eur. J. Agron. 82, 282–291. https://doi.org/10.1016/j.eja.2016.09.011
- Andersen, E., Elbersen, B., Godeschalk, F., Verhoog, D., 2007. Farm management indicators and farm typologies as a basis for assessments in a changing policy environment. J. Environ. Manage. 82, 353– 362.
- Arodudu, O., Helming, K., Wiggering, H., Voinov, A., 2017. Towards a more holistic sustainability assessment framework for agro-bioenergy systems — A review. Environ. Impact Assess. Rev. 62, 61–75. https://doi.org/10.1016/j.eiar.2016.07.008
- Ashkenazy, A., Calvão Chebach, T., Knickel, K., Peter, S., Horowitz, B., Offenbach, R., 2018. Operationalising resilience in farms and rural regions – Findings from fourteen case studies. J. Rural Stud. 59, 211–221. https://doi.org/10.1016/j.jrurstud.2017.07.008
- Bakker, M.M., de Vries, W., Ros, G.H., Kros, J., Kuhlman, J.W., Mashhoodi, B., de Vries, S., Witte, J.-P., 2021. Zoneren biedt landbouw toekomstperspectief. Milieu Doss. 2021, 39–44.
- Balmann, A., Dautzenberg, K., Happe, K., Kellermann, K., 2006. On the dynamics of structural change in agriculture: Internal frictions, policy threats and vertical integration. Outlook Agric. 35, 115–121. https://doi.org/10.5367/00000006777641543
- Balmann, A., Odening, M., Weikard, H.P., Brandes, W., 1996. Path-dependence without increasing returns to scale and network externalities. J. Econ. Behav. Organ. 29, 159–172. https://doi.org/10.1016/0167-2681(95)00055-0
- Becking, J., Soriano, B., Bardaji, I., 2019. FoPIA-SURE-Farm Case-study Report Spain, in: Paas, W., Accatino, F., Antonioli, F., Appel, F., Bardaji, I., Coopmans, I., Courtney, P., Gavrilescu, C., Heinrich, F., Krupin, V., Manevska-Tasevska, G., Neumeister, D., Peneva, M., Rommel, J., Severini, S., Soriano, B., Tudor, M., Urquhart, J., Wauters, E., Zawalinska, K., Meuwissen, M., Reidsma, P. (Eds.), D5.2 Participatory Impact Assessment of Sustainability and Resilience of EU Farming Systems. Sustainable and resilient EU farming systems (SureFarm) project report, EU Horizon 2020 Grant Agreement No. 727520.
- Belton, V., Stewart, T.J.T.A.-T.T.-, 2002. Multiple Criteria Decision Analysis : an Integrated Approach. https://doi.org/10.1007/978-1-4615-1495-4 LK - https://wur.on.worldcat.org/oclc/852788641
- Bennett, E.M., Peterson, G.D., 2005. A systems model approach to determining resilience surrogates for case studies. Ecosystems 8, 945–957.
- Berkum, S. van, Dengerink, J., Ruben, R., 2018. The food systems approach: sustainable solutions for a sufficient supply of healthy food, Wageningen Economic Research memorandum: 2018-064. Wageningen Economic Research, LEI Internationaal Beleid.
- Bernués, A., Olaizola, A., 2012. La ganadería en los Pirineos: Evolución, condicionantes y oportunidades, in: Lasagabaster, I. (Ed.), Los Pirineos: Geografía, Turismo, Agricultura, Cooperación Transfronteriza y Derecho. Universidad del País Vasco, pp. 29–67.
- Bernués, A., Riedel, J.L., Asensio, M.A., Blanco, M., Sanz, A., Revilla, R., Casasús, I., 2005. An integrated approach to studying the role of grazing livestock systems in the conservation of rangelands in a protected natural park (Sierra de Guara, Spain). Livest. Prod. Sci. 96, 75–85. https://doi.org/10.1016/j.livprodsci.2005.05.023

- Bernués, A., Riedel, J.L., Casasús, I., Olaizola, A., 2006. The conservation of natural resources as an extrinsic quality attribute of lamb in Mediterranean Areas, in: Ramalhi Ribeiro, J., Horta, A., Mosconi, C., Rosati, A. (Eds.), Animal Products from the Mediterranean Area, EAAP Publication 119. Wageningen Academic Publishers, pp. 73–82.
- Bernués, A., Ruiz, R., Olaizola, A., Villalba, D., Casasús, I., 2011. Sustainability of pasture-based livestock farming systems in the European Mediterranean context: Synergies and trade-offs. Livest. Sci. 139, 44–57. https://doi.org/10.1016/j.livsci.2011.03.018
- Bertolozzi-Caredio, D., Bardaji, I., Coopmans, I., Soriano, B., Garrido, A., 2020. Key steps and dynamics of family farm succession in marginal extensive livestock farming. J. Rural Stud. 76, 131–141. https://doi.org/10.1016/J.JRURSTUD.2020.04.030
- Bertolozzi-Caredio, D., Garrido, A., Soriano, B., Bardaji, I., 2021. Implications of alternative farm management patterns to promote resilience in extensive sheep farming. A Spanish case study. J. Rural Stud. 86, 633–644. https://doi.org/10.1016/J.JRURSTUD.2021.08.007
- Biasi, R., Botti, F., 2010. Hazelnut landscape transformation in the northern Latium: the study case of the Monti Cimini's hills. Corylus Co. 2, 39–48.
- Biermann, F., Kim, R.E., 2020. The Boundaries of the Planetary Boundary Framework: A Critical Appraisal of Approaches to Define a "Safe Operating Space" for Humanity. Annu. Rev. Environ. Resour. 45, 497–521. https://doi.org/10.1146/annurev-environ-012320-080337
- Biesbroek, R., Dupuis, J., Wellstead, A., 2017. Explaining through causal mechanisms: resilience and governance of social-ecological systems. Curr. Opin. Environ. Sustain. 28, 64–70. https://doi.org/10.1016/j.cosust.2017.08.007
- Biggs, R., Peterson, G.D., Rocha, J.C., 2018. The Regime Shifts Database: a framework for analyzing regime shifts in social-ecological systems. Ecol. Soc. 23, 9. https://doi.org/10.5751/ES-10264-230309
- Biggs, R., Schlüter, M., Biggs, D., Bohensky, E.L., BurnSilver, S., Cundill, G., Dakos, V., Daw, T.M., Evans, L.S., Kotschy, K., Leitch, A.M., Meek, C., Quinlan, A., Raudsepp-Hearne, C., Robards, M.D., Schoon, M.L., Schultz, L., West, P.C., 2012. Toward Principles for Enhancing the Resilience of Ecosystem Services. Annu. Rev. Environ. Resour. 37, 421–448. https://doi.org/10.1146/annurev-environ-051211-123836
- Binder, C.R., Feola, G., Steinberger, J.K., 2010. Considering the normative, systemic and procedural dimensions in indicator-based sustainability assessments in agriculture. Environ. Impact Assess. Rev. 30, 71–81.
- Blomkvist, P., Larsson, J., 2013. An analytical framework for common-pool resource large technical system (CPR-LTS) constellations. Int. J. Commons 7, 113–139.
- Boland, M.J., Rae, A.N., Vereijken, J.M., Meuwissen, M.P.M., Fischer, A.R.H., van Boekel, M.A.J.S., Rutherfurd, S.M., Gruppen, H., Moughan, P.J., Hendriks, W.H., 2013. The future supply of animalderived protein for human consumption. Trends Food Sci. Technol. 29, 62–73. https://doi.org/10.1016/j.tifs.2012.07.002
- Bosque, M.A., Navarro, V.P., 2002. El proceso de desertización demográfica de la montaña pirenaica en el largo plazo: Aragón. Ager Rev. Estud. sobre despoblación y Desarro. Rural = J. depopulation Rural Dev. Stud. 2, 101–138.
- Bouttes, M., San Cristobal, M., Martin, G., 2018. Vulnerability to climatic and economic variability is mainly driven by farmers' practices on French organic dairy farms. Eur. J. Agron. 94, 89–97. https://doi.org/10.1016/j.eja.2018.01.013
- Brand, F.S., Jax, K., 2007. Focusing the meaning(s) of resilience: Resilience as a descriptive concept and a boundary object. Ecol. Soc. 12, 23. https://doi.org/10.5751/ES-02029-120123
- Brown, C., Seo, B., Rounsevell, M., 2019. Societal breakdown as an emergent property of large-scale behavioural models of land use change. Earth Syst. Dyn. 10, 809–845. https://doi.org/10.5194/esd-10-809-2019
- Brown, P.W., Schulte, L.A., 2011. Agricultural landscape change (1937-2002) in three townships in Iowa, USA. Landsc. Urban Plan. 100, 202–212. https://doi.org/10.1016/j.landurbplan.2010.12.007
- Buitenhuis, Y., Candel, J., Feindt, P.H., Termeer, K., Mathijs, E., Bardají, I., Black, J., Martikainen, A., Moeyersons, M., Sorrentino, A., 2020a. Improving the resilience-enabling capacity of the Common Agricultural Policy: policy recommendations for more resilient EU farming systems. Eurochoices 19, 63–71. https://doi.org/10.1111/1746-692X.12286
- Buitenhuis, Y., Candel, J.J.L., Termeer, K.J.A.M., Feindt, P.H., 2020b. Does the Common Agricultural Policy enhance farming systems' resilience? Applying the Resilience Assessment Tool (ResAT) to a farming system case study in the Netherlands. J. Rural Stud. 80, 314–327. https://doi.org/10.1016/J.JRURSTUD.2020.10.004
- Cabell, J.F., Oelofse, M., 2012. An Indicator Framework for Assessing Agroecosystem Resilience. Ecol. Soc. 17, 18. https://doi.org/10.5751/ES-04666-170118
- Callo-Concha, D., Ewert, F., 2014. Using the Concepts of Resilience, Vulnerability and Adaptability for the Assessment and Analysis of Agricultural Systems. Chang. Adapt. Socio-Ecological Syst. 1, 1–11. https://doi.org/10.2478/cass-2014-0001
- Campbell, B.M., Beare, D.J., Bennett, E.M., Hall-Spencer, J.M., Ingram, J.S.I., Jaramillo, F., Ortiz, R., Ramankutty, N., Sayer, J.A., Shindell, D., 2017. Agriculture production as a major driver of the Earth system exceeding planetary boundaries. Ecol. Soc. 22, 8. https://doi.org/10.5751/ES-09595-220408
Cao, K.-A. Le, Rohart, F., Gonzalez, I., Dejean, S., 2017. mixOmics: Omics Data Integration Project.

- Carpenter, S., Walker, B., Anderies, J.M., Abel, N., 2001. From Metaphor to Measurement: Resilience of What to What? Ecosystems 4, 765–781. https://doi.org/10.1007/s10021-001-0045-9
- Carpenter, S.R., Westley, F., Turner, M.G., 2005. Surrogates for resilience of social-ecological systems. Ecosystems 8, 941–944.
- Casasús, İ., Bernués, A., Sanz, A., Villalba, D., Riedel, J.L., Revilla, R., 2007. Vegetation dynamics in Mediterranean forest pastures as affected by beef cattle grazing. Agric. Ecosyst. Environ. 121, 365– 370. https://doi.org/10.1016/j.agee.2006.11.012
- Chambers, J.M., Wyborn, C., Klenk, N.L., Ryan, M., Serban, A., Bennett, N.J., Brennan, R., Charli-Joseph, L., Fernández-Giménez, M.E., Galvin, K.A., Goldstein, B.E., Haller, T., Hill, R., Munera, C., Nel, J.L., Österblom, H., Reid, R.S., Riechers, M., Spierenburg, M., Tengö, M., Bennett, E., Brandeis, A., Chatterton, P., Cockburn, J.J., Cvitanovic, C., Dumrongrojwatthana, P., Paz Durán, A., Gerber, J.D., Green, J.M.H., Gruby, R., Guerrero, A.M., Horcea-Milcu, A.I., Montana, J., Steyaert, P., Zaehringer, J.G., Bednarek, A.T., Curran, K., Fada, S.J., Hutton, J., Leimona, B., Pickering, T., Rondeau, R., 2022. Co-productive agility and four collaborative pathways to sustainability transformations. Glob. Environ. Chang. 72, 102422. https://doi.org/10.1016/J.GLOENVCHA.2021.102422
- Chatellier, V., 2021. Review: International trade in animal products and the place of the European Union: main trends over the last 20 years. Animal 15, 100289. https://doi.org/10.1016/J.ANIMAL.2021.100289
- Coopmans, I., Dessein, J., Accatino, F., Antonioli, F., Bertolozzi-Caredio, D., Gavrilescu, C., Gradziuk, P., Manevska-Tasevska, G., Meuwissen, M., Peneva, M., Petitt, A., Urquhart, J., Wauters, E., 2021. Understanding farm generational renewal and its influencing factors in Europe. J. Rural Stud. 86, 398–409. https://doi.org/https://doi.org/10.1016/i.jrurstud.2021.06.023
- Corcoran, K., 2003. Marketing red meat in the European Union: extending the options. Final Report of project FAIR (SME) FA-S2-CT98-9093 (CRAFT programme).
- Cumming, G.S., Collier, J., 2005. Change and identity in complex systems. Ecol. Soc. 10, 29.
- Cumming, G.S., Morrison, T.H., Hughes, T.P., 2017. New Directions for Understanding the Spatial Resilience of Social–Ecological Systems. Ecosystems 20, 649–664. https://doi.org/10.1007/s10021-016-0089-5
- Cumming, G.S., Peterson, G.D., 2017. Unifying Research on Social-Ecological Resilience and Collapse. Trends Ecol. Evol. 32, 695–713. https://doi.org/10.1016/j.tree.2017.06.014
- Dardonville, M., Bockstaller, C., Therond, O., 2021. Review of quantitative evaluations of the resilience, vulnerability, robustness and adaptive capacity of temperate agricultural systems. J. Clean. Prod. 286, 125456. https://doi.org/10.1016/J.JCLEPRO.2020.125456
- Dardonville, M., Bockstaller, C., Villerd, J., Therond, O., 2022a. Resilience of agricultural systems: biodiversity-based systems are stable, while intensified ones are resistant and high-yielding. Agric. Syst. 197, 103365. https://doi.org/10.1016/J.AGSY.2022.103365
- Dardonville, M., Legrand, B., Clivot, H., Bernardin, C., Bockstaller, C., Therond, O., 2022b. Assessment of ecosystem services and natural capital dynamics in agroecosystems. Ecosyst. Serv. 54, 101415. https://doi.org/10.1016/J.ECOSER.2022.101415
- Dardonville, M., Urruty, N., Bockstaller, C., Therond, O., 2020. Influence of diversity and intensification level on vulnerability, resilience and robustness of agricultural systems. Agric. Syst. 184, 102913. https://doi.org/10.1016/J.AGSY.2020.102913
- Darnhofer, I., 2021. Resilience or how do we enable agricultural systems to ride the waves of unexpected change? Agric. Syst. 187, 102997. https://doi.org/https://doi.org/10.1016/j.agsy.2020.102997
- Darnhofer, I., 2014. Resilience and why it matters for farm management. Eur. Rev. Agric. Econ. 41, 461– 484. https://doi.org/10.1093/erae/jbu012
- Darnhofer, I., 2010. Strategies of family farms to strengthen their resilience. Environ. policy Gov. 20(4), 212–222.
- De Figueiredo, H.S., Meuwissen, M.P.M., Van Der Lans, I.A., Oude Lansink, A.G.J.M., 2017. Beyond upgrading typologies – In search of a better deal for honey value chains in Brazil. PLoS One 12, e0181391. https://doi.org/10.1371/journal.pone.0181391
- De Rancourt, M., Fois, N., Lavín, M.P., Tchakérian, E., Vallerand, F., 2006. Mediterranean sheep and goats production: An uncertain future. Small Rumin. Res. 62, 167–179. https://doi.org/10.1016/j.smallrumres.2005.08.012
- Dearing, J.A., Wang, R., Zhang, K., Dyke, J.G., Haberl, H., Hossain, M.S., Langdon, P.G., Lenton, T.M., Raworth, K., Brown, S., Carstensen, J., Cole, M.J., Cornell, S.E., Dawson, T.P., Doncaster, C.P., Eigenbrod, F., Flörke, M., Jeffers, E., Mackay, A.W., Nykvist, B., Poppy, G.M., 2014. Safe and just operating spaces for regional social-ecological systems. Glob. Environ. Chang. 28, 227–238. https://doi.org/10.1016/J.GLOENVCHA.2014.06.012
- Delmotte, S., Couderc, V., Mouret, J.-C., Lopez-Ridaura, S., Barbier, J.-M., Hossard, L., 2017. From stakeholders narratives to modelling plausible future agricultural systems. Integrated assessment of scenarios for Camargue, Southern France. Eur. J. Agron. 82, 292–307. https://doi.org/http://dx.doi.org/10.1016/j.eja.2016.09.009
- Delmotte, S., Lopez-Ridaura, S., Barbier, J.M., Wery, J., 2013. Prospective and participatory integrated assessment of agricultural systems from farm to regional scales: Comparison of three modeling approaches. J. Environ. Manage. 129, 493–502. https://doi.org/10.1016/j.jenvman.2013.08.001

DG-Agri, 2017. Farm structures. European Commission, DG Agriculture and Rural Development, Brussels. DG-AGRI, 2017. Agricultural and farm income. European Commission, DG Agriculture and Rural Development. Brussels.

Dinno, A., 2017. conover.test: Conover-Iman Test of Multiple Comparisons Using Rank Sums. R package version 1.1.5.

Diogo, V., Reidsma, P., Schaap, B., Andree, B.P.J., Koomen, E., 2017. Assessing local and regional economic impacts of climatic extremes and feasibility of adaptation measures in Dutch arable farming systems. Agric. Syst. 157, 216–229. https://doi.org/10.1016/j.agsy.2017.06.013

- Dixon, J.M., Weerahewa, J., Hellin, J., Rola-Rubzen, M.F., Huang, J., Kumar, S., Das, A., Qureshi, M.E., Krupnik, T.J., Shideed, K., Jat, M.L., Prasad, P.V.V., Yadav, S., Irshad, A., Asanaliev, A., Abugalieva, A., Karimov, A., Bhattarai, B., Balgos, C.Q., Benu, F., Ehara, H., Pant, J., Sarmiento, J.M.P., Newby, J.C., Pretty, J., Tokuda, H., Weyerhaeuser, H., Digal, L.N., Li, L., Sarkar, M.A.R., Abedin, M.Z., Schreinemachers, P., Grafton, Q., Sharma, R.C., Saidzoda, S., Lopez-Ridaura, S., Coffey, S., Kam, S.P., Win, S.S., Praneetvatakul, S., Maraseni, T., Touch, V., Liang, W. Ii, Saharawat, Y.S., Timsina, J., 2021. Response and resilience of Asian agrifood systems to COVID-19: An assessment across twenty-five countries and four regional farming and food systems. Agric. Syst. 193, 103168. https://doi.org/10.1016/J.AGSY.2021.103168
- Douxchamps, S., Debevec, L., Giordano, M., Barron, J., 2017. Monitoring and evaluation of climate resilience for agricultural development – A review of currently available tools. World Dev. Perspect. 5, 10–23. https://doi.org/10.1016/i.wdp.2017.02.001
- Dreborg, K.H., 1996. Essence of backcasting. Futures 28, 813–828. https://doi.org/10.1016/S0016-3287(96)00044-4
- Duchek, S., 2020. Organizational resilience: a capability-based conceptualization. Bus. Res. 13, 215–246. https://doi.org/10.1007/s40685-019-0085-7
- EIP-Agri, 2018. Agricultural Knowledge and Innovation Systems: stimulating creativity and learning. AIP-Agri Service Point.
- Enfors-Kautsky, E., Järnberg, L., Quinlan, A., Ryan, P., 2018. Wayfinder: a resilience guide for navigating towards sustainable futures [WWW Document]. GRAID Program. Stock. Resil. Centre. URL https://wayfinder.earth/ (accessed 6.1.21).
- Ercin, E., Veldkamp, T.I.E., Hunink, J., 2021. Cross-border climate vulnerabilities of the European Union to drought. Nat. Commun. 12, 3322. https://doi.org/10.1038/s41467-021-23584-0
- Ericksen, P.J., 2008. What is the vulnerability of a food system to global environmental change? Ecol. Soc. 13, 14. https://doi.org/10.5751/ES-02475-130214
- Eurofound, 2021. Living conditions and quality of life [WWW Document]. URL https://www.eurofound.europa.eu/topic/living-conditions-quality-life
- Eurofound, 2019. Is rural Europe being left behind? European Quality of Life Survey 2016, Publications Office of the European Union, Luzembourg.
- Eurofound, 2017. 6th European Working Conditions Survey Overview report (2017 update). Publications Office of the European Union, Luxembourg.
- European Commission, 2018. CAP Context Indicators 2014-2020.
- European Commission, 2015. The Monitoring and Evaluation Framework for the Common Agricultural Policy: 2014-2020 [WWW Document]. URL http://publications.europa.eu/resource/cellar/00da6abf-7c75-11e5-9fae-01aa75ed71a1.0021.03/DOC 1
- European Commission, 2013. Structure and dynamics of EU farms: changes, trends and policy relevance. EU Agricultural Economics Briefs. No 9.
- European Court of Auditors, 2021. Sustainable water use in agriculture: CAP funds more likely to promote greater rather than more efficient water use. Special Report. Publications Office of the European Union, Luxembourg, Luxembourg.
- European Court of Auditors, 2020. Biodiversity on farmland: CAP contribution has not halted the decline. Publication Office of the European Union, Luxembourg, Luxembourg.
- European Environment Agency, 2020. Global climate change impacts and the supply of agricultural commodities to Europe.
- Eurostat, 2018. Farms and farmland in the European Union statistics [WWW Document]. URL https://ec.europa.eu/eurostat/statistics-

 $explained/index.php?title=Farms_and_farmland_in_the_European_Union_-$

- _statistics#Farms_in_2016 (accessed 4.5.22).
- Eurostat, 2013. Nutrient Budgets Methodology and Handbook. Version 1.02. Eurostat and OECD, Luxembourg.
- Fader, M., Gerten, D., Krause, M., Lucht, W., Cramer, W., 2013. Spatial decoupling of agricultural production and consumption: quantifying dependences of countries on food imports due to domestic land and water constraints. Environ. Res. Lett. 8, 014046. https://doi.org/http://dx.doi.org/10.1088/1748-9326/8/1/014046
- FAO UNDP and UNEP, 2021. A multi-billion-dollar opportunity Repurposing agricultural support to transform food systems. Rome, FAO.
- Feindt, P.H., Termeer, C.J.A.M., Candel, J., Buitenhuis, Y., 2019. D4.2 Assessing how policies enable or constrain the resilience of farming systems in the European Union: case study results. Sustainable and resilient EU farming systems (SureFarm) project report, EU Horizon 2020 Grant Agreement No.

727520. https://doi.org/https://surefarmproject.eu/wordpress/wp-content/uploads/2019/02/SURE-Farm-D4.2-Report-on-Resilience-Assessment-Case-Studies.pdf

Figueiredo, J., Pereira, H.M., 2011. Regime shifts in a socio-ecological model of farmland abandonment. Landsc. Ecol. 26, 737–749. https://doi.org/10.1007/s10980-011-9605-3

- Folke, C., Carpenter, S.R., Walker, B., Scheffer, M., Chapin, T., Rockström, J., 2010. Resilience thinking: integrating resilience, adaptability and transformability. Ecol. Soc. 15(4), 20.
- Ford Denison, R., McGuire, A.M., 2015. What should agriculture copy from natural ecosystems? Glob. Food Sec. 4, 30–36. https://doi.org/https://doi.org/10.1016/j.gfs.2014.12.002
- Fuchs, R., Brown, C., Rounsevell, M., 2020. Europe's Green Deal offshores environmental damage to other nations. Nature 586, 671–673. https://doi.org/10.1038/d41586-020-02991-1
- García-Martínez, A., Olaizola, A., Bernués, A., 2009. Trajectories of evolution and drivers of change in European mountain cattle farming systems. Animal 3, 152–165. https://doi.org/10.1017/S1751731108003297
- Gaupp, F., Hall, J., Mitchell, D., Dadson, S., 2019. Increasing risks of multiple breadbasket failure under 1.5 and 2 °C global warming. Agric. Syst. 175, 34–45. https://doi.org/10.1016/J.AGSY.2019.05.010
- Ge, L., Anten, N.P.R., van Dixhoorn, I.D.E., Feindt, P.H., Kramer, K., Leemans, R., Meuwissen, M.P.M., Spoolder, H., Sukkel, W., 2016. Why we need resilience thinking to meet societal challenges in biobased production systems. Curr. Opin. Environ. Sustain. 23, 17–27. https://doi.org/10.1016/j.cosust.2016.11.009
- Geels, F.W., 2011. The multi-level perspective on sustainability transitions: Responses to seven criticisms. Environ. Innov. Soc. Transitions 1, 24–40. https://doi.org/10.1016/J.EIST.2011.02.002
- Geoghegan, J., Wainger, L.A., Bockstael, N.E., 1997. Spatial landscape indices in a hedonic framework: An ecological economics analysis using GIS. Ecol. Econ. 23, 251–264. https://doi.org/10.1016/S0921-8009(97)00583-1
- Gil, J.D.B., Cohn, A.S., Duncan, J., Newton, P., Vermeulen, S., 2017. The resilience of integrated agricultural systems to climate change. Wiley Interdiscip. Rev. Clim. Chang. 8, e461. https://doi.org/10.1002/wcc.461
- Giller, K.E., 2013. Guest editorial: Can we define the term "farming systems"? A question of scale. Outlook Agric. 42, 149–153. https://doi.org/10.5367/oa.2013.0139
- Giller, K.E., Delaune, T., Silva, J.V., Descheemaeker, K., van de Ven, G., Schut, A.G.T., van Wijk, M., Hammond, J., Hochman, Z., Taulya, G., Chikowo, R., Narayanan, S., Kishore, A., Bresciani, F., Teixeira, H.M., Andersson, J.A., van Ittersum, M.K., 2021. The future of farming: Who will produce our food? Food Secur. 13, 1073–1099. https://doi.org/10.1007/s12571-021-01184-6
- Giller, K.E., Leeuwis, C., Andersson, J.A., Andriesse, W., Brouwer, A., Frost, P., Hebinck, P., Heitkönig, I., Van Ittersum, M.K., Koning, N., Ruben, R., Slingerland, M., Udo, H., Veldkamp, T., Van de Vijver, C., Van Wijk, M.T., Windmeijer, P., 2008. Competing claims on natural resources: What role for science? Ecol. Soc. 13(2), 34.
- Gliessman, S.R., 2015. Agroecology: The ecology of sustainable food systems, Third Edit. ed. CRC Press, Taylor & Francis Group, Boca Raton, USA.
- Gobierno de Aragón, 2019. Datos estadísticos sobre ganadería en Aragón: efectivos ganaderos, distribución de ganadería, movimiento comercial pecuario, producciones ganaderas.
- Gobierno de Aragón, 2016. El ovino y el caprino en Aragón. Evolución en los últimos 20 años (1996-2016). Godfray, H.C.J., 2015. The debate over sustainable intensification. Food Secur. 7, 199-208.
- https://doi.org/10.1007/s12571-015-0424-2
- Goodwin, D., Holman, I., Pardthaisong, L., Visessri, S., Ekkawatpanit, C., Rey Vicario, D., 2022. What is the evidence linking financial assistance for drought-affected agriculture and resilience in tropical Asia? A systematic review. Reg. Environ. Chang. 22, 12. https://doi.org/10.1007/s10113-021-01867-y
- Groffman, P.M., Baron, J.S., Blett, T., Gold, A.J., Goodman, I., Gunderson, L.H., Levinson, B.M., Palmer, M.A., Paerl, H.W., Peterson, G.D., Poff, N.L., Rejeski, D.W., Reynolds, J.F., Turner, M.G., Weathers, K.C., Wiens, J., 2006. Ecological thresholds: The key to successful environmental management or an important concept with no practical application? Ecosystems 9, 1–13.
- Guarín, A., Rivera, M., Pinto-Correia, T., Guiomar, N., Šūmane, S., Moreno-Pérez, O.M., 2020. A new typology of small farms in Europe. Glob. Food Sec. 26, 100389. https://doi.org/10.1016/J.GFS.2020.100389
- Guerra, A.L., Gidel, T., Vezzetti, E., 2016. Toward a common procedure using likert and likert-type scales in small groups comparative design observations, in: Dorian, M., Mario, S., Neven, P., Nenad, B., Stanko, S. (Eds.), Proceedings of the DESIGN 2016 14th International Design Conference. pp. 23– 32.
- Gunderson, L.H., Holling, C.S., 2002. Panarchy: Understanding transformations in human and natural systems. Island Press, Washington D.C., USA.
- Hall, D.C., Norgaard, R.B., 1973. On the timing and application of pesticides. Am. J. Agric. Econ. 55, 198– 201. https://doi.org/10.3386/w16455
- Hardeman, E., Jochemsen, H., 2012. Are There Ideological Aspects to the Modernization of Agriculture? 25, 657–674.
- Helfenstein, J., Diogo, V., Bürgi, M., Verburg, P.H., Schüpbach, B., Szerencsits, E., Mohr, F., Siegrist, M., Swart, R., Herzog, F., 2022. An approach for comparing agricultural development to societal visions.

Agron. Sustain. Dev. 42, 5. https://doi.org/10.1007/s13593-021-00739-3

- Hermans, C.M.L., Geijzendorffer, I.R., Ewert, F., Metzger, M.J., Vereijken, P.H., Woltjer, G.B., Verhagen, A., 2010. Exploring the future of European crop production in a liberalised market, with specific consideration of climate change and the regional competitiveness. Ecol. Modell. 221, 2177–2187.
- Hernández-Mora, N., Gil, M., Garrido, A., Rodríguez-Casado, R., 2012. La sequía 2005-2008 en la cuenca del Ebro: vulnerabilidad, impactos y medidas de gestión. UPM-CEIGRAM-Madrid.
- Herrera, H., 2017a. Resilience for Whom? The Problem Structuring Process of the Resilience Analysis. Sustainability 9, 1196.
- Herrera, H., 2017b. From Metaphor to Practice: Operationalizing the Analysis of Resilience Using System Dynamics Modelling. Syst. Res. Behav. Sci. 34, 444–462. https://doi.org/10.1002/sres.2468
- Herrera, H., Kopainsky, B., 2020. Using system dynamics to support a participatory assessment of resilience. Environ. Syst. Decis. 40, 342–355. https://doi.org/10.1007/s10669-020-09760-5
- Herrera, H., Schütz, L., Paas, W., Reidsma, P., Kopainsky, B., 2022. Understanding resilience of farming systems: Insights from system dynamics modelling for an arable farming system in the Netherlands. Ecol. Modell. 464, 109848. https://doi.org/10.1016/J.ECOLMODEL.2021.109848
- Hijbeek, R., Cormont, A., Hazeu, G., Bechini, L., Zavattaro, L., Janssen, B., Werner, M., Schlatter, N., Guzmán, G., Bijttebier, J., Pronk, A.A., van Eupen, M., van Ittersum, M.K., 2017. Do farmers perceive a deficiency of soil organic matter? A European and farm level analysis. Ecol. Indic. 83, 390–403. https://doi.org/10.1016/J.ECOLIND.2017.08.023
- Hoekstra, A.Y., Bredenhoff-Bijlsma, R., Krol, M.S., 2018. The control versus resilience rationale for managing systems under uncertainty. Environ. Res. Lett. 13, 103002. https://doi.org/10.1088/1748-9326/aadf95
- Homer-Dixon, T., Walker, B., Biggs, R., Crépin, A.-S., Folke, C., Lambin, E.F., Peterson, G.D., Rockström, J., Scheffer, M., Steffen, W., Troell, M., 2015. Synchronous failure: the emerging causal architecture of global crisis. Ecol. Soc. 20. https://doi.org/10.5751/ES-07681-200306
- Huang, J., Tichit, M., Poulot, M., Darly, S., Li, S., Petit, C., Aubry, C., 2015. Comparative review of multifunctionality and ecosystem services in sustainable agriculture. J. Environ. Manage. 149, 138– 147. https://doi.org/10.1016/j.jenvman.2014.10.020
- Jones, L., 2019. Resilience isn't the same for all: Comparing subjective and objective approaches to resilience measurement. Wiley Interdiscip. Rev. Clim. Chang. 10, e552. https://doi.org/10.1002/wcc.552
- Jones, L., d'Errico, M., 2019. Whose resilience matters? Like-for-like comparison of objective and subjective evaluations of resilience. World Dev. 124, 104632. https://doi.org/10.1016/j.worlddev.2019.104632
- JRC of the European Commission, 2021. Gridded Agro-Meteorological Data in Europe [Dataset]. Just. R.E., Pope, R.D., 1978. Stochastic specification of production functions and economic implications. J.
- Econom. 7, 67–86. https://doi.org/10.1016/0304-4076(78)90006-4
- Kadaster, 2021. Agrarische grond [WWW Document]. URL https://www.kadaster.nl/zakelijk/vastgoedinformatie/vastgoedcijfers/vastgoeddashboard/agrarisch e-grond (accessed 2.17.21).
- Kanter, D.R., Musumba, M., Wood, S.L.R., Palm, C., Antle, J., Balvanera, P., Dale, V.H., Havlik, P., Kline, K.L., Scholes, R.J., Thornton, P., Tittonell, P., Andelman, S., 2018. Evaluating agricultural trade-offs in the age of sustainable development. Agric. Syst. 163, 73–88. https://doi.org/10.1016/j.agsy.2016.09.010
- Kharrazi, A., Savaget, P., Kudo, S., 2019. Resilience and Sustainability, in: Leal Filho, W. (Ed.), Encyclopedia of Sustainability in Higher Education. Springer, Cham. https://doi.org/10.1007/978-3-319-63951-2_92-1
- Kinnunen, P., Guillaume, J.H.A., Taka, M., D'Odorico, P., Siebert, S., Puma, M.J., Jalava, M., Kummu, M., 2020. Local food crop production can fulfil demand for less than one-third of the population. Nat. Food 1, 229–237. https://doi.org/10.1038/s43016-020-0060-7
- Kinzig, A.P., Ryan, P., Etienne, M., Allison, H., Elmqvist, T., Walker, B.H., 2006. Resilience and regime shifts: Assessing cascading effects. Ecol. Soc. 11, 20. https://doi.org/10.5751/ES-01678-110120
- Klapwijk, C.J., van Wijk, M.T., Rosenstock, T.S., van Asten, P.J.A., Thornton, P.K., Giller, K.E., 2014. Analysis of trade-offs in agricultural systems: Current status and way forward. Curr. Opin. Environ. Sustain. 6, 110–115.
- König, H.J., Uthes, S., Schuler, J., Zhen, L., Purushothaman, S., Suarma, U., Sghaier, M., Makokha, S., Helming, K., Sieber, S., Chen, L., Brouwer, F., Morris, J., Wiggering, H., 2013. Regional impact assessment of land use scenarios in developing countries using the FoPIA approach: Findings from five case studies. J. Environ. Manage. 127, S56–S64. https://doi.org/10.1016/j.jenvman.2012.10.021
- Koning, N., van Ittersum, M.K., 2009. Will the world have enough to eat? Curr. Opin. Environ. Sustain. 1, 77–82.
- Lamkowsky, M., Oenema, O., Meuwissen, M.P.M., Ang, F., 2021. Closing productivity gaps among Dutch dairy farms can boost profit and reduce nitrogen pollution. Environ. Res. Lett. 16, 124003. https://doi.org/10.1088/1748-9326/ac3286
- Lasanta-Martínez, T., Vicente-Serrano, S.M., Cuadrat-Prats, J.M., 2005. Mountain Mediterranean landscape evolution caused by the abandonment of traditional primary activities: A study of the Spanish Central Pyrenees. Appl. Geogr. 25, 47–65. https://doi.org/10.1016/j.apgeog.2004.11.001

- Lê Cao, K.A., Rossouw, D., Robert-Granié, C., P., B., 2008. A sparse PLS for variable selection when integrating omics data. Stat. Appl. Genet. Mol. Biol. 7, 35. https://doi.org/10.2202/1544-6115.1390
- Le Goff, U., Sander, A., Lagana, M.H., Barjolle, D., Phillips, S., Six, J., 2022. Raising up to the climate challenge - Understanding and assessing farmers' strategies to build their resilience. A comparative analysis between Ugandan and Swiss farmers. J. Rural Stud. 89, 1–12. https://doi.org/10.1016/J.JRURSTUD.2021.10.020

Liberti, S., 2019. Il gusto amaro delle nocciole. Internazionale 1312, 39-47.

- Mandryk, M., Reidsma, P., van Ittersum, M.K., 2017. Crop and farm level adaptation under future climate challenges: An exploratory study considering multiple objectives for Flevoland, the Netherlands. Agric. Syst. 152, 154–164. https://doi.org/10.1016/j.agsy.2016.12.016
- Mandryk, M., Reidsma, P., van Ittersum, M.K., 2012. Scenarios of long-term farm structural change for application in climate change impact assessment. Landsc. Ecol. 27, 509–527.
- Mann, C.C., 2019. The Wizard and the Prophet. Pan Macmillan, Basingstoke, UK.
- MAPA, 2020. El sector ovino y caprino en cifras. Principales indicadores económicos [WWW Document]. URL https://www.mapa.gob.es/es/ganaderia/temas/produccion-y-mercadosganaderos/indicadoreseconomicosdelsectorovinoycaprino_carne_tcm30-5114962019_tcm30-511496.pdf
- MAPA, 2019a. Encuestas de efectivos de ganado ovino y caprino. [WWW Document]. URL https://www.mapa.gob.es/es/estadistica/temas/estadisticas
 - agrarias/resultados_definitivos_nov2019_ovino-caprino_tcm30-526727.pdf
- MAPA, 2019b. Anuario de estadística [WWW Document]. URL https://www.mapa.gob.es/estadistica/pags/anuario/2019/CAPITULOSPDF/CAPITULO03/pdfc03_3.p df
- MAPA, 2019c. Red contable agraria nacional (RECAN) [WWW Document]. URL https://www.mapa.gob.es/es/estadistica/temas/estadisticas-agrarias/economia/red-contablerecan/ (accessed 6.5.19).
- MAPA, 2017. Estudios de costes y rentas de explotaciones ganaderas (ECREA) [WWW Document]. URL https://www.mapa.gob.es/es/ministerio/servicios/analisis-y-prospectiva/ganadoovinocarne_tcm30-508573.pdf
- Marchese, D., Reynolds, E., Bates, M.E., Morgan, H., Clark, S.S., Linkov, I., 2018. Resilience and sustainability: Similarities and differences in environmental management applications. Sci. Total Environ. 613–614, 1275–1283. https://doi.org/10.1016/J.SCITOTENV.2017.09.086
- Marshall, N.A., Park, S.E., Adger, A.N., Brown, K., Howden, S.M., 2012. Transformational capacity and the influence of place and identity. Environ. Res. Lett. 7, 034022.
- Martin-Collado, D., Díaz Martín, C., Serrano, M., Carabaño, M., Ramón, M., Zanoli, R., 2019. Sheep dairy and meat products: from urban consumers' perspective to industry innovations. Options Méditerranéennes 123, 277–281.
- Martin, G., Magne, M.-A., Cristobal, M.S., 2017. An Integrated Method to Analyze Farm Vulnerability to Climatic and Economic Variability According to Farm Configurations and Farmers' Adaptations. Front. Plant Sci. 8, 1483. https://doi.org/10.3389/fpls.2017.01483
- Mathijs, E., Wauters, E., 2020. Making farming systems truly resilient. Eurochoices 19, 72–76. https://doi.org/10.1111/1746-692X.12287
- Mayer, A.L., 2008. Strengths and weaknesses of common sustainability indices for multidimensional systems. Environ. Int. 34, 277–91. https://doi.org/10.1016/j.envint.2007.09.004
- Mazzucato, M., 2018. The value of everything: Making and taking in the global economy. Hachette UK.
- McKeown, B., Thomas, D., 2013. Q Methodology. https://doi.org/10.4135/9781483384412
- Meadows, D.H., 2008. Thinking in systems: a Primer. Chelsea Green Publishing, UK.
- Metzger, M.J., Rounsevell, M.D.A., Acosta-Michlik, L., Leemans, R., Schroter, D., 2006. The vulnerability of ecosystem services to land use change. Agric. Ecosyst. Environ. 114, 69–85.
- Meul, M., Nevens, F., Reheul, D., Hofman, G., 2007. Energy use efficiency of specialised dairy, arable and pig farms in Flanders. Agric. Ecosyst. Environ. 119, 135–144. https://doi.org/10.1016/j.agee.2006.07.002
- Meuwissen, M.P.M., Feindt, P.H., Midmore, P., Wauters, E., Finger, R., Appel, F., Spiegel, A., Mathijs, E., Termeer, C.J.A.M., Balmann, A., De Mey, Y., Reidsma, P., 2020. The struggle of farming systems in Europe: looking for explanations through the lens of resilience. Eurochoices 19, 4–11. https://doi.org/10.1111/1746-692X.12278.
- Meuwissen, M.P.M., Feindt, P.H., Slijper, T., Spiegel, A., Finger, R., de Mey, Y., Paas, W., Termeer, K.J.A.M., Poortvliet, P.M., Peneva, M., Urquhart, J., Vigani, M., Black, J.E., Nicholas-Davies, P., Maye, D., Appel, F., Heinrich, F., Balmann, A., Bijttebier, J., Coopmans, I., Wauters, E., Mathijs, E., Hansson, H., Lagerkvist, C.J., Rommel, J., Manevska-Tasevska, G., Accatino, F., Pineau, C., Soriano, B., Bardaji, I., Severini, S., Senni, S., Zinnanti, C., Gavrilescu, C., Bruma, I.S., Dobay, K.M., Matei, D., Tanasa, L., Voicilas, D.M., Zawalińska, K., Gradziuk, P., Krupin, V., Martikainen, A., Herrera, H., Reidsma, P., 2021. Impact of Covid-19 on farming systems in Europe through the lens of resilience thinking. Agric. Syst. 191, 103152. https://doi.org/10.1016/j.agsy.2021.103152
- Meuwissen, M.P.M., Feindt, P.H., Spiegel, A., Termeer, C.J.A.M., Mathijs, E., de Mey, Y., Finger, R., Balmann, A., Wauters, E., Urquhart, J., Vigani, M., Zawalinska, K., Herrera, H., Nicholas-Davies, P., Hansson, H., Paas, W., Slijper, T., Coopmans, I., Vroege, W., Ciechomska, A., Accatino, F.,

Kopainsky, B., Poortvliet, P.M., Candel, J., Maye, D., Severini, S., Senni, S., Soriano, B., Lagerkvist, C.J., Peneva, M., Gavrilescu, C., Reidsma, P., 2019. A framework to assess the resilience of farming systems. Agric. Syst. 176, 102656.

- Milán, M.J., Arnalte, E., Caja, G., 2003. Economic profitability and typology of Ripollesa breed sheep farms in Spain. Small Rumin. Res. 49, 97–105. https://doi.org/10.1016/S0921-4488(03)00058-0
- Mitter, H., Techen, A.-K., Sinabell, F., Helming, K., Kok, K., Priess, J., Bodirsky, B., Holman, I., Lehtonen, H., Leip, A., Le Mouël, C., Mathijs, E., Mehdi, B., Michetti, M., Mittenzwei, K., Mora, O., Øygarden, L., Reidsma, P., Schaldach, R., Schmid, E., Schönhart, M., 2020. Shared Socio-economic Pathways for European agriculture and food systems: The Eur-Agri-SSPs. Glob. Environ. Chang. 65, 102159.
- Mitter, H., Techen, A., Sinabell, F., Helming, K., Kok., K., Priess, J., Schmid, E., Bodirsky, B., Holman, I., Lehtonen, H., Leip, A., Le Mouël, C., Mathijs, E., Mehdi, B., Michetti, M., Mittenzwei, K., Mora, O., Øygarden, K., Øygarden, L., Reidsma, P., Schaldach, R., Schönhart, M., 2019. A protocol to develop Shared Socio-economic Pathways for European agriculture. J. Environ. Manage. 252, 109701. https://doi.org/10.1016/j.jenvman.2019.109701
- Morris, J.B., Tassone, V., de Groot, R., Camilleri, M., Moncada, S., 2011. A framework for participatory impact assessment: Involving stakeholders in European policy making, A case study of land use change in Malta. Ecol. Soc. 16, 12.
- Mosse, D., 1994. Authority, Gender and Knowledge: Theoretical Reflections on the Practice of Participatory Rural Appraisal. Dev. Change 25, 497–526. https://doi.org/10.1111/j.1467-7660.1994.tb00524.x
- Mulders, P.J.A.M., van den Heuvel, E.R., van den Borne, J., van de Molengraft, R., Heemels, W.P.M.H., Reidsma, P., 2021. Data science at farm level: Explaining and predicting within-farm variability in potato growth and yield. Eur. J. Agron. 123, 126220. https://doi.org/10.1016/J.EJA.2020.126220
- Nemec, K.T., Chan, J., Hoffman, C., Spanbauer, T.L., Hamm, J.A., Allen, C.R., Hefley, T., Pan, D., Shrestha, P., 2014. Assessing resilience in stressed watersheds. Ecol. Soc. 19, 34. https://doi.org/10.5751/ES-06156-190134
- Nera, E., Paas, W., Reidsma, P., Paolini, G., Antonioli, F., Severini, S., 2020. Assessing the Resilience and Sustainability of a Hazelnut Farming System in Central Italy with a Participatory Approach. Sustainability 12, 343. https://doi.org/10.3390/su12010343
- Norman, G., 2010. Likert scales, levels of measurement and the "laws" of statistics. Adv. Heal. Sci. Educ. 15, 625–632. https://doi.org/10.1007/s10459-010-9222-y
- Nyström, M., Jouffray, J.B., Norström, A. V., Crona, B., Søgaard Jørgensen, P., Carpenter, S.R., Bodin, Galaz, V., Folke, C., 2019. Anatomy and resilience of the global production ecosystem. Nature 575, 98–108. https://doi.org/10.1038/s41586-019-1712-3
- OECD, 2021. Long-term interest rates [WWW Document]. URL https://data.oecd.org/interest/long-terminterest-rates.htm (accessed 2.17.21).
- Ornai, A., Ne'eman, G., Keasar, T., 2020. Management of forest fire buffer zones: Implications for flowering plants and bees. For. Ecol. Manage. 473, 118310. https://doi.org/10.1016/j.foreco.2020.118310
- Ostrom, E., 1990. Governing the Commons: The Evolution of Institutions for collective Action. Cambridge University Press. .
- Paas, W., Accatino, F., Antonioli, F., Appel, F., Bardaji, I., Coopmans, I., Courtney, P., Gavrilescu, C., Heinrich, F., Krupin, V., Manevska-Tasevska, G., Neumeister, D., Peneva, M., Rommel, J., Severini, S., Soriano, B., Tudor, M., Urquhart, J., Wauters, E., Zawalinska, K., Meuwissen, M., Reidsma, P., 2019. D5.2 Participatory impact assessment of sustainability and resilience of EU farming systems. Sustainable and resilient EU farming systems (SureFarm) project report, EU Horizon 2020 Grant Agreement No. 727520. https://doi.org/10.13140/RG.2.2.25104.25601
- Paas, W., Accatino, F., Appel, F., Bijttebier, J., Black, J., Gavrilescu, C., Krupin, V., Manevska-Tasevska, G., Ollendorf, F., Peneva, M., Rommel, J., San Martín, C., Severini, S., Soriano, B., Valchovska, S., Vigani, M., Wauters, E., Zawalińska, K., Zinnanti, C., Meuwissen, M., Reidsma, P., 2020. FoPIA-SURE-Farm 2, in: Accatino, F., Paas, W., Herrera, H., Appel, F., Pinsard, C., Yong, S., Schutz, L., Kopainsky, B., Bańkowska, K., Bijttebier, J., Black, J., Gavrilescu, C., Krupin, V., Manevska-Tasevska, G., Ollendorf, F., Peneva, M., Rommel, J., San Martín, C., Severini, S., Soriano, B., Valchovska, S., Vigani, M., Wauters, E., Zawalińska, K., Zinnanti, C., Meuwissen, M., Reidsma, P. (Eds.), D5.5 Impacts of Future Scenarios on the Resilience of Farming Systems across the EU Assessed with Quantitative and Qualitative Methods. Sustainable and resilient EU farming systems (SureFarm) project report, EU Horizon 2020 Grant Agreement No. 727520.
- Paas, W., Reidsma, P., 2020. Guidelines for FoPIA-SURE-Farm 2, in: Accatino, F., et al. (Eds.), D5.5 Impacts of Future Scenarios on the Resilience of Farming Systems across the EU Assessed with Quantitative and Qualitative Methods. Sustainable and resilient EU farming systems (SureFarm) project report, EU Horizon 2020 Grant Agreement No. 727520.
- Pardos, L., Maza, M.T., Fantova, E., 2007. Influencia de la prima fija en los resultados técnicos y económicos de explotaciones ovinas de carne en Aragón. ITEA 28, 291–293.
- Pardos, L., Maza, M.T., Fantova, E., Sepúlveda, W., 2008. The diversity of sheep production systems in Aragón (Spain): Characterisation and typification of meat sheep farms. Spanish J. Agric. Res. 6, 497–507. https://doi.org/10.5424/sjar/2008064-344
- Peco, B., Navarro, E., Carmona, C.P., Medina, N.G., Marques, M.J., 2017. Effects of grazing abandonment on soil multifunctionality: The role of plant functional traits. Agric. Ecosyst. Environ. 249, 215–225. https://doi.org/10.1016/j.agee.2017.08.013

Peterson, C.A., Eviner, V.T., Gaudin, A.C.M., 2018. Ways forward for resilience research in agroecosystems. Agric. Syst. 162, 19–27. https://doi.org/10.1016/J.AGSY.2018.01.011

Pielke, R.A., 2007. The honest broker: Making sense of science in policy and politics. Cambridge University Press, UK. https://doi.org/10.1017/CB09780511818110

Pimm, S.L., 1984. The complexity and stability of ecosystems. Nature 307, 321–326. https://doi.org/10.1038/307321a0

Pinsard, C., Martin, S., Léger, F., Accatino, F., 2021. Robustness to import declines of three types of European farming systems assessed with a dynamic nitrogen flow model. Agric. Syst. 193, 103215. https://doi.org/https://doi.org/10.1016/j.agsy.2021.103215

Plumecocq, G., Debril, T., Duru, M., Magrini, M.B., Sarthou, J.P., Therond, O., 2018. The plurality of values in sustainable agriculture models: Diverse lock-in and coevolution patterns. Ecol. Soc. 23, 21. https://doi.org/10.5751/ES-09881-230121

Poppe, K., 2004. Het Bedrijven-Informatienet van A tot Z. LEI, The Hague.

Pretty, J., 2008. Agricultural sustainability: concepts, principles and evidence. Philos. Trans. R. Soc. B Biol. Sci. 363, 447–465. https://doi.org/10.1098/rstb.2007.2163

Pretty, J., Bharucha, Z.P., 2014. Sustainable intensification in agricultural systems. Ann. Bot. 114, 1571– 1596. https://doi.org/10.1093/aob/mcu205

Prosperi, P., Allen, T., Cogill, B., Padilla, M., Peri, I., 2016. Towards metrics of sustainable food systems: a review of the resilience and vulnerability literature. Environ. Syst. Decis. 36, 3–19. https://doi.org/10.1007/s10669-016-9584-7

Quinlan, A.E., Berbés-Blázquez, M., Haider, L.J., Peterson, G.D., 2016. Measuring and assessing resilience: broadening understanding through multiple disciplinary perspectives. J. Appl. Ecol. 53, 677–687. https://doi.org/10.1111/1365-2664.12550

Quist, J., Vergragt, P., 2006. Past and future of backcasting: The shift to stakeholder participation and a proposal for a methodological framework. Futures 38, 1027–1045. https://doi.org/10.1016/j.futures.2006.02.010

R Core Team, 2015. R: A Language and Environment for Statistical Computing.

Rabbinge, R., van Diepen, C.A., 2000. Changes in agriculture and land use in Europe. Eur. J. Agron. 13, 85–99.

Ravetz, J., 2002. The challenge beyond orthodox science. Futures 34, 200–203. https://doi.org/10.1016/S0016-3287(01)00060-X

Raworth, K., 2017. Doughnut economics : seven ways to think like a 21st-century economist. London : Random House.

Reidsma, P., Bakker, M.M., Kanellopoulos, A., Alam, S.J., Paas, W., Kros, J., de Vries, W., 2015. Sustainable agricultural development in a rural area in the Netherlands? Assessing impacts of climate and socio-economic change at farm and landscape level. Agric. Syst. 141, 160–173. https://doi.org/10.1016/j.agsy.2015.10.009

Reidsma, P., Ewert, F., 2008. Regional farm diversity can reduce vulnerability of food production to climate change. Ecol. Soc. 13, 38. https://doi.org/Artn 38

Reidsma, P., Ewert, F., Oude Lansink, A., 2007. Analysis of farm performance in Europe under different climatic and management conditions to improve understanding of adaptive capacity. Clim. Change 84, 403–422. https://doi.org/10.1007/s10584-007-9242-7

Reidsma, P., Ewert, F., Oude Lansink, A., Leemans, R., 2009a. Vulnerability and adaptation of European farmers: A multi-level analysis of yield and income responses to climate variability. Reg. Environ. Chang. 9, 25–40. https://doi.org/10.1007/s10113-008-0059-3

Reidsma, P., Oude Lansink, A., Ewert, F., 2009b. Economic impacts of climate variability and subsidies on European agriculture and observed adaptation strategies. Mitig. Adapt. Strateg. Glob. Chang. 14, 35–59.

Reidsma, P., Paas, W., Accatino, F., Appel, F., Black, J.E., Bijttebier, J., Gavrilescu, C., Kopainsky, B., Krupin, V., Manevska-Tasevska, G., Meuwissen, M.P.M., Ollendorf, F., Peneva, M., Senni, S., Severini, S., Soriano, B., Urquhart, J., Vigani, M., Zawalinska, K., Zinnanti, C., Herrera, H., 2021. Resilience of farming systems in the EU under current conditions and future scenarios: D5.7 Policy Brief.

Reidsma, P., Paas, W., Spiegel, A., Meuwissen, M., 2019a. Guidelines for the Framework of Participatory Impact Assessment of SUstainable and REsilient EU FARMing systems (FoPIA-Surefarm), in: Paas, W., Accatino, F., Antonioli, F., Appel, F., Bardaji, I., Coopmans, I., Courtney, P., Gavrilescu, C., Heinrich, F., Krupin, V., Manevska-Tasevska, G., Neumeister, D., Peneva, M., Rommel, J., Severini, S., Soriano, B., Tudor, M., Urquhart, J., Wauters, E., Zawalinska, K., Meuwissen, M., Reidsma, P. (Eds.), D5.2 Participatory Impact Assessment of Sustainability and Resilience of EU Farming Systems. Sustainable and resilient EU farming systems (SURE-Farm) project report, EU Horizon 2020 Grant Agreement No. 727520.

Reidsma, P., Spiegel, A., Paas, W., Accatino, F., Antonioli, F., Appel, F., Bardají, I., Berry, R., Bertolozzi, D., Bijttebier, J., Black, J., Buitenhuis, Y., Coopmans, I., Courtney, P., Feindt, P., Gavrilescu, C., Hansson, H., Jendrzejewski, B., Khafagy, A., Krupin, V., Lagerkvist, C.-J., Larson, S., Lievens, E., Mathijs, E., Manevska-Tasevska, G., Maye, D., Ollendorf, F., Peneva, M., Pettit, A., Pinsard, C., Rommel, J., Senni, S., Severini, S., Slijper, T., Soriano, B., Urquhart, J., Valchovska, S., Vigani, M., Wauters, E., Zawalińska, K., Meuwissen, M., 2019b. D5.3 Resilience assessment of current farming systems across the European Union. Sustainable and resilient EU farming systems (SURE-Farm)

project report, EU Horizon 2020 Grant Agreement No. 727520.

- Resilience Alliance, 2010. Assessing resilience in social-ecological systems: Workbook for practitioners. Version 2.0.
- Ridder, D., Pahl-Wostl, C., 2005. Participatory Integrated Assessment in local level planning. Reg. Environ. Chang. 5, 188–196.
- Riedel, J.L., Casasús, I., Bernués, A., 2007. Sheep farming intensification and utilization of natural resources in a Mediterranean pastoral agro-ecosystem. Livest. Sci. 111, 153–163. https://doi.org/10.1016/j.livsci.2006.12.013
- Rist, L., Felton, A., Nyström, M., Troell, M., Sponseller, R.A., Bengtsson, J., Österblom, H., Lindborg, R., Tidåker, P., Angeler, D.G., Milestad, R., Moen, J., 2014. Applying resilience thinking to production ecosystems. Ecosphere 5, 1–11. https://doi.org/10.1890/ES13-00330.1
- Robards, M.D., Schoon, M.L., Meek, C.L., Engle, N.L., 2011. The importance of social drivers in the resilient provision of ecosystem services. Glob. Environ. Chang. 21, 522–529. https://doi.org/10.1016/j.gloenvcha.2010.12.004
- Rocha, J.C., Peterson, G.D., Biggs, R., 2015. Regime shifts in the anthropocene: Drivers, risks, and resilience. PLoS One 10, e0134639. https://doi.org/10.1371/journal.pone.0134639
- Rockström, J., Steffen, W., Noone, K., Persson, Ä., Chapin, F.S., Lambin, E.F., Lenton, T.M., Scheffer, M., Folke, C., Schellnhuber, H.J., Nykvist, B., De Wit, C.A., Hughes, T., Van Der Leeuw, S., Rodhe, H., Sörlin, S., Snyder, P.K., Costanza, R., Svedin, U., Falkenmark, M., Karlberg, L., Corell, R.W., Fabry, V.J., Hansen, J., Walker, B., Liverman, D., Richardson, K., Crutzen, P., Foley, J.A., 2009a. A safe operating space for humanity. Nature 461, 472–475.
- Rockström, J., Steffen, W., Noone, K., Persson, A., Chapin Iii, F.S., Lambin, E., Lenton, T.M., Scheffer, M., Folke, C., Schellnhuber, H.J., Nykvist, B., de Wit, C.A., Hughes, T., van der Leeuw, S., Rodhe, H., Sörlin, S., Snyder, P.K., Costanza, R., Svedin, U., Falkenmark, M., Karlberg, L., Corell, R.W., Fabry, V.J., Hansen, J., Walker, B., Liverman, D., Richardson, K., Crutzen, P., Foley, J., 2009b. Planetary boundaries: Exploring the safe operating space for humanity. Ecol. Soc. 14, 32.
- Rodríguez-Ortega, T., Oteros-Rozas, E., Ripoll-Bosch, R., Tichit, M., Martín-López, B., Bernués, A., 2014. Applying the ecosystem services framework to pasture-based livestock farming systems in Europe. Animal 8, 1361–1372. https://doi.org/10.1017/S1751731114000421
- Ruiz-Mirazo, J., Robles, A.B., 2012. Impact of targeted sheep grazing on herbage and holm oak saplings in a silvopastoral wildfire prevention system in south-eastern Spain. Agrofor. Syst. 86, 477–491. https://doi.org/10.1007/s10457-012-9510-z
- Schaap, B., Blom-Zandstra, M., Hermans, C.M.L., Meerburg, B.G., Verhagen, J., 2011. Impact changes of climatic extremes on arable farming in the north of the Netherlands. Reg. Environ. Chang. 11, 731– 741.
- Schaap, B.F., Reidsma, P., Verhagen, J., Wolf, J., van Ittersum, M.K., 2013. Participatory design of farm level adaptation to climate risks in an arable region in The Netherlands. Eur. J. Agron. 48, 30–42.
- Schlüter, M., Biggs, R., Schoon, M.L., Robards, M.D., Anderies, J.M., 2015. Reflections on building resilience – interactions among principles and implications for governance, in: Schlüter, M., Schoon, M.L., Biggs, R. (Eds.), Principles for Building Resilience: Sustaining Ecosystem Services in Social-Ecological Systems. Cambridge University Press, Cambridge, pp. 251–282. https://doi.org/DOI: 10.1017/CB09781316014240.011
- Schulte-Uebbing, L., de Vries, W., 2021. Reconciling food production and environmental boundaries for nitrogen in the European Union. Sci. Total Environ. 786, 147427. https://doi.org/10.1016/J.SCITOTENV.2021.147427
- Schulte, R.P.O., O'Sullivan, L., Vrebos, D., Bampa, F., Jones, A., Staes, J., 2019. Demands on land: Mapping competing societal expectations for the functionality of agricultural soils in Europe. Environ. Sci. Policy 100, 113–125. https://doi.org/https://doi.org/10.1016/j.envsci.2019.06.011
- Sellberg, M.M., Cockburn, J., Holden, P.B., Lam, D.P.M., 2021. Towards a caring transdisciplinary research practice: navigating science, society and self. Ecosyst. People 17, 292–305. https://doi.org/10.1080/26395916.2021.1931452
- Shi, Y., Pinsard, C., Accatino, F., 2021. Land sharing strategies for addressing the trade-off between carbon storage and crop production in France. Reg. Environ. Chang. 21, 92. https://doi.org/10.1007/s10113-021-01818-7
 Shudemith M. Carata and C. Shudemith M. Carata and Carata a
- Shucksmith, M., Cameron, S., Merridew, T., Pichler, F., 2009. Urban–Rural Differences in Quality of Life across the European Union. Reg. Stud. 43, 1275–1289. https://doi.org/10.1080/00343400802378750
- Sieber, S., Amjath-Babu, T.S., Reidsma, P., Koenig, H., Piorr, A., Bezlepkina, I., Mueller, K., 2018. Sustainability impact assessment tools for land use policy advice: A comparative analysis of five research approaches. Land use policy 71, 75–85. https://doi.org/10.1016/j.landusepol.2017.11.042
- Silva, J.V., Reidsma, P., Baudron, F., Laborte, A.G., Giller, K.E., van Ittersum, M.K., 2021a. How sustainable is sustainable intensification? Assessing yield gaps at field and farm level across the globe. Glob. Food Sec. 30, 100552. https://doi.org/10.1016/j.gfs.2021.100552
- Silva, J.V., Reidsma, P., van Ittersum, M.K., 2017. Yield gaps in Dutch arable farming systems: Analysis at crop and crop rotation level. Agric. Syst. 158, 78–92. https://doi.org/10.1016/j.agsy.2017.06.005
- Silva, J.V., Tenreiro, T.R., Spätjens, L., Anten, N.P.R., van Ittersum, M.K., Reidsma, P., 2020. Can big data explain yield variability and water productivity in intensive cropping systems? F. Crop. Res. 255,

107828. https://doi.org/10.1016/J.FCR.2020.107828

- Silva, J.V., van Ittersum, M.K., ten Berge, H.F.M., Spätjens, L., Tenreiro, T.R., Anten, N.P.R., Reidsma, P., 2021b. Agronomic analysis of nitrogen performance indicators in intensive arable cropping systems: An appraisal of big data from commercial farms. F. Crop. Res. 269, 108176. https://doi.org/10.1016/J.FCR.2021.108176
- Silva, V., Catry, F.X., Fernandes, P.M., Rego, F.C., Paes, P., Nunes, L., Caperta, A.D., Sergio, C., Bugalho, M.N., 2019. Effects of grazing on plant composition, conservation status and ecosystem services of Natura 2000 shrub-grassland habitat types. Biodivers. Conserv. 28, 1205–1224. https://doi.org/10.1007/s10531-019-01718-7
- Slijper, T., de Mey, Y., Poortvliet, P.M., Meuwissen, M.P., 2021. Quantifying the resilience of European farms using FADN. Eur. Rev. Agric. Econ. 49, 121–150.
- Slijper, Thomas, de Mey, Y., Poortvliet, P.M., Meuwissen, M.P.M., 2021. Quantifying the resilience of European farms using FADN. Eur. Rev. Agric. Econ. 49, 121–150. https://doi.org/10.1093/erae/jbab042
- Slijper, T., de Mey, Y., Poortvliet, P.M., Meuwissen, M.P.M., 2020. From risk behavior to perceived farm resilience: a Dutch case study. Ecol. Soc. 25, 10. https://doi.org/10.5751/ES-11893-250410
- Slijper, T., Urquhart, J., Poortvliet, P.M., Soriano, B., Meuwissen, M.P.M., 2022. Exploring how social capital and learning are related to the resilience of Dutch arable farmers. Agric. Syst. 198, 103385. https://doi.org/10.1016/J.AGSY.2022.103385
- Sneessens, I., Sauvée, L., Randrianasolo-Rakotobe, H., Ingrand, S., 2019. A framework to assess the economic vulnerability of farming systems: Application to mixed crop-livestock systems. Agric. Syst. 176, 102658. https://doi.org/10.1016/J.AGSY.2019.102658
- Soriano, B., Bardají, I., Bertolozzi, D., San Martín, C., Spiegel, A., Slipjer, T., Meuwissen, M.P.M., Rommen, J., Hansson, H., Severini, S., Antonioli, F., Berry, R., Kahfagy, A., Urquhart, J., Harizanova-Bartos, H., Stoyonova, Z., Coopmans, I., Wauters, E., Bijttebier, J., Neumister, D., Accatino, F., Pinsard, C., Tudor, M.M., Gavrilescu, C., Luca, L., Izvoranu, A.M., Zawalińska, K., Jendrzejewski, B., Gradziuk, P., BAŃKOWSKA, K., Krupin, V., Ollendorf, F., Appel, F., Garrido, A., 2020. D2.6 Report on state and outlook for risk management in EU agriculture. Sustainable and resilient EU farming systems (SureFarm) project report. EU Horizon 2020 Grant Agreement No. 727520.
- Spiegel, A., Slijper, T., de Mey, Y., Meuwissen, M.P.M., Poortvliet, P.M., Rommel, J., Hansson, H., Vigani, M., Soriano, B., Wauters, E., Appel, F., Antonioli, F., Gavrilescu, C., Gradziuk, P., Finger, R., Feindt, P.H., 2021. Resilience capacities as perceived by European farmers. Agric. Syst. 193, 103224. https://doi.org/10.1016/J.AGSY.2021.103224
- Spiegel, A., Slijper, T., De Mey, Y., Poortvliet, M., Rommel, J., Hansson, H., Vigani, M., Soriano, B., Wauters, E., Appel, F., Antonioli, F., Harizanova, H., Gavrilescu, C., Gradziuk, P., Neumeister, D., Meuwissen, M., 2019. D2.1 Report on farmers' perceptions of risk, adaptive capacity and resilience. Sustainable and resilient EU farming systems (SureFarm) project report, EU Horizon 2020 Grant Agreement No. 727520.
- Spiegel, A., Soriano, B., de Mey, Y., Slijper, T., Urquhart, J., Bardají, I., Vigani, M., Severini, S., Meuwissen, M., 2020. Risk Management and its Role in Enhancing Perceived Resilience Capacities of Farms and Farming Systems in Europe. EuroChoices 19, 45–53. https://doi.org/https://doi.org/10.1111/1746-692X.12284
- Springmann, M., Clark, M., Mason-D'Croz, D., Wiebe, K., Bodirsky, B.L., Lassaletta, L., de Vries, W., Vermeulen, S.J., Herrero, M., Carlson, K.M., Jonell, M., Troell, M., DeClerck, F., Gordon, L.J., Zurayk, R., Scarborough, P., Rayner, M., Loken, B., Fanzo, J., Godfray, H.C.J., Tilman, D., Rockström, J., Willett, W., 2018. Options for keeping the food system within environmental limits. Nature 562, 519– 525. https://doi.org/10.1038/s41586-018-0594-0
- Steffen, W., Richardson, K., Rockström, J., Cornell, S.E., Fetzer, I., Bennett, E.M., Biggs, R., Carpenter, S.R., de Vries, W., de Wit, C.A., Folke, C., Gerten, D., Heinke, J., Mace, G.M., Persson, L.M., Ramanathan, V., Reyers, B., Sörlin, S., 2015. Planetary boundaries: Guiding human development on a changing planet. Science (80-.). 347, 1259855. https://doi.org/10.1126/science.1259855
- Sterman, J., 2000. Business dynamics: systems thinking and modeling for a complex world. Boston: Irwin/McGraw-Hill.
- Stockholm Resilience Centre, 2020. Regime Shifts DataBase: large persistent changes in ecosystem services. [WWW Document]. URL https://www.regimeshifts.org/what-is-a-regime-shift (accessed 12.16.20).
- Strijker, D., 2005. Marginal lands in Europe Causes of decline. Basic Appl. Ecol. 6, 99–106. https://doi.org/10.1016/j.baae.2005.01.001
- Suess-Reyes, J., Fuetsch, E., 2016. The future of family farming: A literature review on innovative, sustainable and succession-oriented strategies. J. Rural Stud. 47, 117–140. https://doi.org/10.1016/J.JRURSTUD.2016.07.008
- Sundkvist, Å., Milestad, R., Jansson, A.M., 2005. On the importance of tightening feedback loops for sustainable development of food systems. Food Policy 30, 224–239. https://doi.org/10.1016/j.foodpol.2005.02.003
- Sutton, M.A., Oenema, O., Erisman, J.W., Leip, A., van Grinsven, H., Winiwarter, W., 2011. Too much of a good thing. Nature 472, 159–161. https://doi.org/10.1038/472159a
- Ten Berge, H.F.M., Van Ittersum, M.K., Rossing, W.A.H., Van De Ven, G.W.J., Schans, J., 2000. Farming

options for The Netherlands explored by multi-objective modelling. Eur. J. Agron. 13, 263–277. https://doi.org/10.1016/S1161-0301(00)00078-2

- Tendall, D.M., Joerin, J., Kopainsky, B., Edwards, P., Shreck, A., Le, Q.B., Kruetli, P., Grant, M., Six, J., 2015. Food system resilience: Defining the concept. Glob. Food Sec. 6, 17–23. https://doi.org/10.1016/j.gfs.2015.08.001
- ter Braak, C.J.F., Šmilauer, P., 2015. Topics in constrained and unconstrained ordination. Plant Ecol. https://doi.org/10.1007/s11258-014-0356-5
- Termeer, C.J.A.M., Dewulf, A., 2019. A small wins framework to overcome the evaluation paradox of governing wicked problems. Policy Soc. 38, 298–314. https://doi.org/10.1080/14494035.2018.1497933
- Termeer, C.J.A.M., Dewulf, A., Biesbroek, G.R., 2017. Transformational change: governance interventions for climate change adaptation from a continuous change perspective. J. Environ. Plan. Manag. 60, 558–576. https://doi.org/10.1080/09640568.2016.1168288
- Termeer, C.J.A.M., Feindt, P.H., Karpouzoglou, T., Poppe, K.J., Hofstede, G.J., Kramer, K., Ge, L., Mathijs, E., Meuwissen, M.P.M., 2019. Institutions and the resilience of biobased production systems: the historical case of livestock intensification in the Netherlands. Ecol. Soc. 24, 15. https://doi.org/10.5751/ES-11206-240415
- Therond, O., Duru, M., Roger-Estrade, J., Richard, G., 2017. A new analytical framework of farming system and agriculture model diversities. A review. Agron. Sustain. Dev. 37, 21. https://doi.org/10.1007/s13593-017-0429-7
- Tilman, D., Cassman, K.G., Matson, P.A., Naylor, R., Polasky, S., 2002. Agricultural sustainability and intensive production practices. Nature 418, 671–677. https://doi.org/10.1038/nature01014
- Tittonell, P., 2020. Assessing resilience and adaptability in agroecological transitions. Agric. Syst. 184, 102862. https://doi.org/10.1016/j.agsy.2020.102862
- Toth, F.L., 2001. Participatory integrated assessment methods: An assessment of their usefulness to the European Environmental Agency. Technical report 64, European Environment Agency.
- Turner, N., 2005. Sustainable production of crops and pastures under drought in a Mediterranean environment. Ann. Appl. Biol. 144, 139–147.
- United Nations, 1987. Our Common Future Brundtland Report. Oxford University Press.
- Urquhart, J., Accatino, F., Appel, F., Antonioli, F., Berry, R., Bertolozzi, D., Bijttebier, J., Black, J., Bohateret, V.M., Bruma, I.S., Coopmans, I., Courtney, P., Daskiewicz, H., Dinculsecu, C., Dinu-Vasiliu, C., Dobay, K.-M., Gavrilescu, C., Gradziuk, P., Heinrich, F., Izvoranu, A.M., Johansson, S., Klein, K., Matei, D., Maye, D., Peneva, M., Petit, A., Pitson, C., Severini, S., Slijper, T., Soriano, B., Tanasa, L., Toma, C., Tudor, M.-M., Vigani, M., Zawalińska, K., 2019. D2.3. Report on farmers? learning capacity and networks of influence in 11 European case studies. Sustainable and resilient EU farming systems (SURE-Farm) project report, EU Horizon 2020 Grant Agreement No. 727520.
- Urruty, N., Tailliez-Lefebvre, D., Huyghe, C., 2016. Stability, robustness, vulnerability and resilience of agricultural systems. A review. Agron. Sustain. Dev. 36, 1–15. https://doi.org/10.1007/s13593-015-0347-5
- Vaidya, A., Mayer, A.L., 2014. Use of the participatory approach to develop sustainability assessments for natural resource management. Int. J. Sustain. Dev. World Ecol. 21, 369–379. https://doi.org/10.1080/13504509.2013.868376
- Van Apeldoorn, D.F., Kempen, B., Sonneveld, M.P.W., Kok, K., 2013. Co-evolution of landscape patterns and agricultural intensification: An example of dairy farming in a traditional Dutch landscape. Agric. Ecosyst. Environ. 172, 16–23. https://doi.org/10.1016/j.agee.2013.04.002
- van Apeldoorn, D.F., Kok, K., Sonneveld, M.P.W., Veldkamp, T.A., 2011. Panarchy rules: Rethinking resilience of agroecosystems, evidence from Dutch dairy-farming. Ecol. Soc. 16, 39. https://doi.org/10.5751/ES-03949-160139
- van Calker, K.J., Berentsen, P.B.M., Giesen, G.W.J., Huirne, R.B.M., 2005. Identifying and ranking attributes that determine sustainability in Dutch dairy farming. Agric. Human Values 22, 53–63.
- van de Ven, G.W.J., de Valença, A., Marinus, W., de Jager, I., Descheemaeker, K.K.E., Hekman, W., Mellisse, B.T., Baijukya, F., Omari, M., Giller, K.E., 2021. Living income benchmarking of rural households in low-income countries. Food Secur. 13, 729–749. https://doi.org/10.1007/s12571-020-01099-8
- van der Bolt, B., van Nes, E.H., Bathiany, S., Vollebregt, M.E., Scheffer, M., 2018. Climate reddening increases the chance of critical transitions. Nat. Clim. Chang. 8, 478–484. https://doi.org/10.1038/s41558-018-0160-7
- van der Veen, H.B., Ge, L., van der Meer, R.W., Vrolijk, H.C.J., 2014. Sample of Dutch FADN 2012 : design principles and quality of the sample of agricultural and horticultural holdings, LEI report / LEI Wageningen UR : 2014-027. LEI Wageningen UR, 906, LEI Performance en Impact Agrosectoren,.
- van der Werf, W., Bianchi, F., 2022. Options for diversifying agricultural systems to reduce pesticide use: Can we learn from nature? Outlook Agric. 51, 105–113. https://doi.org/10.1177/00307270221077442
- van Dijk, M., Morley, T., Jongeneel, R., van Ittersum, M., Reidsma, P., Ruben, R., 2017. Disentangling agronomic and economic yield gaps: An integrated framework and application. Agric. Syst. 154, 90–99. https://doi.org/10.1016/J.AGSY.2017.03.004
- van Grinsven, H.J.M., van Eerdt, M.M., Westhoek, H., Kruitwagen, S., 2019. Benchmarking Eco-Efficiency

and Footprints of Dutch Agriculture in European Context and Implications for Policies for Climate and Environment. Front. Sustain. Food Syst. 3. https://doi.org/10.3389/fsufs.2019.00013

Van Ittersum, M.K., Rabbinge, R., 1997. Concepts in production ecology for analysis and quantification of agricultural input-output combinations. F. Crop. Res. 52, 197–208.

- Van Ittersum, M.K., Rabbinge, R., Van Latesteijn, H.C., 1998. Exploratory land use studies and their role in strategic policy making. Agric. Syst. 58, 309–330.
- Van Passel, S., Massetti, E., Mendelsohn, R., 2017. A Ricardian Analysis of the Impact of Climate Change on European Agriculture. Environ. Resour. Econ. 67, 725–760. https://doi.org/10.1007/s10640-016-0001-y
- Van Passel, S., Meul, M., 2012. Multilevel and multi-user sustainability assessment of farming systems. Environ. Impact Assess. Rev. 32, 170–180. https://doi.org/10.1016/j.eiar.2011.08.005
- van Vliet, J.A., Schut, A.G.T., Reidsma, P., Descheemaeker, K., Slingerland, M., van de Ven, G.W.J., Giller, K.E., 2015. De-mystifying family farming: Features, diversity and trends across the globe. Glob. Food Sec. 5, 11–18. https://doi.org/10.1016/j.gfs.2015.03.001
- van Vliet, J.A., Slingerland, M.A., Waarts, Y.R., Giller, K.E., 2021. A Living Income for Cocoa Producers in Côte d'Ivoire and Ghana? Front. Sustain. Food Syst. 5, 732831. https://doi.org/10.3389/fsufs.2021.732831
- Vicente-Serrano, S., Lasanta-Martínez, Teodoro Cuadrat-Prats, J., 2000. Influencia de la ganadería en la evolución del riesgo de incendio en función de la vegetación en un área de montaña: el ejemplo del valle de Borau (Pirineo aragonés). Geographicalia 33–58. https://doi.org/10.26754/ojs_geoph/geoph.2000381382
- von Braun, J., Afsana, K., Fresco, L.O., Hassan, M., Torero, M., 2021. Food Systems Definition, Concept and Application for the UN Food Systems Summit.
- Walker, B., Carpenter, S., Anderies, J., Abel, N., Cumming, G., Janssen, M., Lebel, L., Norberg, J., Peterson, G.D., Pritchard, R., 2002. Resilience management in social-ecological systems: A working hypothesis for a participatory approach. Ecol. Soc. 6, 14.
- Walker, B., Carpenter, S.R., Folke, C., Gunderson, L., Peterson, G.D., Scheffer, M., Schoon, M., Westley, F.R., 2020. Navigating the chaos of an unfolding global cycle. Ecol. Soc. 25, 23. https://doi.org/10.5751/ES-12072-250423
- Walker, B., Holling, C.S., Carpenter, S.R., Kinzig, A.P., 2004. Resilience, Adaptability and Transformability in Social–ecological Systems. Ecol. Soc. 9 (2), 5.
- Walker, B., Salt, D., 2012. Resilience practice: Building capacity to absorb disturbance and maintain function. Island Press, Washington D.C., USA. https://doi.org/10.5822/978-1-61091-231-0
- WECR, 2020. Agrimatie informatie over de agrosector [WWW Document]. URL https://www.agrimatie.nl/Binternet.aspx?ID=2&Bedrijfstype=1%4022&SelectedJaren=2018%4020 17%402016%402015&GroteKlassen=Alle+bedrijven (accessed 3.2.20).
- Westley, F., Laban, S., Rose, C., McGowan, K., Robinson, K., Tjornbo, O., Tovey, M., 2015. Social Innovation Lab Guide.
- Westley, F., Olsson, P., Folke, C., Homer-Dixon, T., Vredenburg, H., Loorbach, D., Thompson, J., Nilsson, M., Lambin, E., Sendzimir, J., Banerjee, B., Galaz, V., Van Der Leeuw, S., 2011. Tipping toward sustainability: Emerging pathways of transformation. Ambio 40, 762. https://doi.org/10.1007/s13280-011-0186-9
- Westley, F.R., Tjornbo, O., Schultz, L., Olsson, P., Folke, C., Crona, B., Bodin, Ö., 2013. A Theory of Transformative Agency in Linked Social-Ecological Systems. Ecol. Soc. 18. https://doi.org/10.5751/ES-05072-180327
- Winkler, K.J., Scown, M.W., Nicholas, K.A., 2018. A classification to align social-ecological land systems research with policy in Europe. Land use policy 79, 137–145. https://doi.org/10.1016/J.LANDUSEPOL.2018.06.034
- Wittmayer, J.M., Schäpke, N., 2014. Action, research and participation: roles of researchers in sustainability transitions. Sustain. Sci. 9, 483–496. https://doi.org/10.1007/s11625-014-0258-4
- Zeileis, A., Leisch, F., Hornik, K., Kleiber, C., 2002. strucchange: An R Package for Testing for Structural Change in Linear Regression Models. J. Stat. Softw. 7, 1–38. https://doi.org/10.18637/jss.v007.i02

Summary

An increasing variety of stresses and shocks provides challenges for farming systems in Europe. As a consequence, the sustainability and resilience of Europe's diverse farming systems is at stake. In particular the possible presence of economic, social or environmental thresholds in farming systems is worrying, as beyond those thresholds permanent and undesired system change may happen.

Sustainability of a system is in this thesis defined as an adequate performance of all system functions across the environmental, economic and social domains. Sustainability of agricultural systems has been studied extensively, but existing frameworks and tools are not designed to study resilience which is much more about the different capacities of systems to deal with disturbances. Moreover, in agricultural sustainability assessments, the social aspects are often least integrated, compared to the economic and environmental. Of particular importance are participatory, integrated assessments that focus on improving sustainability *and* resilience as they are designed to come up with adaptation options and action-oriented approaches together with relevant stakeholders.

Resilience can be defined as the capacity to resist change without changing its feedback system and functionality, i.e. robustness, while acknowledging the possibility of alternative stable states. The acknowledgement of possible stable alternative states of a system has led scholars to argue that ecological resilience thinking should also encompass, besides robustness, the system's capacity to adapt or to organize structural and functional change. Resilience of agricultural systems has been studied at a conceptual level and resilience indicators have been proposed. The actual operationalization has taken place only to a limited extent. In particular, quantitative assessments of agricultural systems are lacking in literature. Lack of good data may be one of the reasons for this. Much resilience work on agricultural systems is therefore qualitative in nature, for instance involving assessments based on system parameters that supposedly bring resilience to the system. The examples in which the resilience concept of agricultural systems is operationalized, are not guided by an integrated resilience framework that was applied to many different agricultural systems.

The work presented in this thesis was part of the EU Horizon 2020 project SURE-Farm to assess the sustainability and resilience of European farming systems. Within SURE-Farm, a resilience framework was developed that considers the need for combining approaches, having a local focus and stimulating participation of relevant (local) actors. The SURE-Farm framework proposed five steps to assess farming system sustainability and resilience (Meuwissen et al., 2019; Figure 1.1). From Step 1 to 5 specific resilience is addressed, i.e. resilience to a specific type disturbance. Step 1 to 3 relate, therefore, to the questions resilience "of what?", "to what?" and "for what purpose?". Step 4 addresses the farming system specific resilience capacities that need to be developed. Based on the previous steps, system characteristics are identified that convey general resilience to the system, regardless the type of disturbance (Step 5).

The aim of this thesis is to operationalize the above introduced resilience framework with new and integrated methods and to assess the sustainability and resilience of current and future European farming systems. The following research questions are central in this thesis: 1) Is there a balance between social, economic and environmental functions in European farming systems in terms of importance and performance? 2) Are European farming systems approaching critical thresholds? 3) What resilience capacities do and should European farming systems have? 4) What strategies enhance sustainability and resilience of European farming systems? The methods applied in this thesis follow the five steps of the resilience framework as much as possible. From a methodological point of view, the methods aim to operationalise the framework by using locally adapted indicators and different sources and types of data.

Chapter 2 presents the Framework of Participatory Impact Assessment for Sustainable and Resilient FARMing systems (FoPIA-SURE-Farm I). FoPIA-SURE-Farm I investigates current farming system functioning, dynamics of main indicators, and specifies resilience for the different resilience capacities, i.e., robustness, adaptability, and transformability. Three case studies with specialized farming systems serve as an example for the used methodology: starch potato production in Veenkoloniën, The Netherlands; dairy production in Flanders, Belgium; and hazelnut production in Lazio, Italy. Chapter 3 presents the

synthesis of the application of FoPIA-SURE-Farm I to 11 European farming systems. Results from Chapters 2 & 3 overlap. In most farming systems, functions that related to food production, economic viability, and maintaining natural resources were perceived as most important. Perceived overall performance of system functions suggests moderate sustainability of the studied farming systems. Overall, the resilience of the studied farming systems was perceived as low to moderate, with robustness and adaptability often dominant over transformability. This indicates that finding pathways to more sustainability, which requires adaptability and transformability, will be a challenging process. General characteristics of farming systems that supposedly convey general resilience, the so-called resilience attributes, were indeed perceived to contribute positively to resilience. Profitability, having production coupled with local and natural resources, heterogeneity of farm types, social self-organization, and infrastructure for innovation were assessed as being important resilience attributes. To allow for transformability, being reasonably profitable and having access to infrastructure for innovation were viewed as essential. Past strategies were often geared towards making the system more profitable, and to a lesser extent towards the other important resilience attributes. To improve sustainability and resilience of European farming systems, responses to short-term processes should better consider long-term processes. Technological innovation is required, but it must be accompanied with structural, social, agro-ecological and institutional changes. The relative importance of some resilience attributes in the studied systems differed from case to case. This indicates that the local context in general, and stakeholder perspectives in particular, are important when evaluating general resilience and policy options based on resilience attributes. Overall, FoPIA-SURE-Farm I results seem a good starting point for raising awareness, further assessments, and eventually for developing a shared vision and action plan for improving sustainability and resilience of farming systems.

While Chapters 2 & 3 presented the development and applications of a participatory semi-quantitative method, Chapter 4 aimed to complement the results regarding sustainability and resilience of current farming systems with a quantitative method. A statistical method using longitudinal data was applied to study farm performance (food production, profitability, nitrogen surplus) under different conditions regarding weather, market and farm structure. A case study for potato production in three regions in the Netherlands was employed, using data from 2006-2019. Statistical model performance was at best moderate. Model results were easily influenced by the selection of response variables. Food production, economic and environmental performance levels and gradual dynamics were primarily determined by input intensity levels. How these levels are determined by intensity of cropping, i.e. positively or negatively, differed per case study. Year-to-year variability was determined by average yearly weather conditions and weather extremes. We did not find evidence of moderating effects of farm structure on the impact of weather conditions and weather extremes. Overall, the conclusion is that results do only provide insights that can confirm existing knowledge at case study level. In the context of resilience of farms, while using a relatively small dataset, the application seems limited to a rather homogeneous farm population in a stable economic environment. Researchers intending to apply this method in (arable) farming systems should be well aware of the influence they can have over the results through the selection of response variables.

Following up on the sustainability and resilience assessment of current systems, Chapter 5 aims to find pathways to more sustainable and resilient farming systems, while identifying and avoiding critical thresholds. To serve this purpose, a participatory, integrated and indicator-based methodology is presented that leads researchers and farming system actors in six steps to a multi-dimensional understanding of sustainability and resilience of farming systems in the future (FoPIA-SURE-Farm II). The method is presented for the case study of extensive sheep production farming system in Huesca, Spain. Participants in the participatory workshop indicated that their farming system is very close to a decline or even a collapse. Approaching and exceeding critical thresholds in the social, economic and environmental domain is currently causing a vicious circle that includes low economic returns, low attractiveness of the farming system and abandonment of pasture lands. More sustainable and resilient alternative systems to counteract the current negative system dynamics were proposed by participants: a semi-intensive system primarily aimed at improving production and a high-tech extensive system primarily aimed at providing public goods. Both alternatives place a strong emphasis on the role of technology, but differ in their approach towards grazing, which is reflected in the different strategies that are foreseen to realize those

alternatives. Although the high-tech extensive system seems most compatible with a future in which sustainable food production is very important, the semi-intensive system seems a less risky bet as it has on average the best compatibility with multiple future scenarios. Overall, the methodology can be regarded as relatively quick, interactive and transdisciplinary, providing ample information on critical thresholds, current system dynamics and future possibilities. As such, the method enables stakeholders to think and talk about the future of their system, paving the way for improved sustainability and resilience.

In Chapter 6, FoPIA-SURE-Farm II is applied to assess the presence of critical thresholds in 11 European farming systems. All studied systems were perceived to be close, at or beyond at least one identified critical threshold, i.e. these systems are (very likely) exceeding thresholds within the next 10 years. Stakeholders were particularly worried about economic viability and food production levels. Moreover, critical thresholds were perceived to interact across system levels (field, farm, farming system) and domains (social, economic, environmental), with low economic viability leading to lower attractiveness of the farming system, and in some farming systems making it hard to maintain natural resources and biodiversity. Overall, a decline in performance of all key system variables was expected by workshop participants in case critical thresholds would be exceeded. For instance, a decline in the attractiveness of the area and a lower maintenance of natural resources and biodiversity. Chapter 6 shows that concern for exceeding critical thresholds is justified and that thresholds need to be studied while considering system variables at field, farm and farming system level across the social, economic and environmental domains. For instance, economic variables at farm level (e.g. income) seem important to detect whether a system is approaching critical thresholds of social variables at farming system level (e.g. attractiveness of the area), while in multiple case studies there are also indications that approaching thresholds of social variables (e.g. labour availability) are indicative for approaching economic thresholds (e.g. farm income). Based on the results, reflections follow on the importance of considering system resources, such as knowledge levels, when monitoring and evaluating the sustainability and resilience of Europe's farming systems.

Chapter 7 provides a synthesis of the research presented in this thesis. Based on the synthesis, policy recommendations are made for improved sustainability and resilience. Subsequently, the framework and methods used in this thesis are evaluated. After that, the relevance of the work is discussed. Based on that discussion, reflections follow on improving the evaluation and monitoring of sustainability and resilience of farming systems in Europe. I conclude that sustainability and resilience of farming systems remains a challenging subject due to its complexity in terms of detail (different domains, many concepts and variables) and dynamics (non-linearity, thresholds, interactions). The research presented in this thesis confirmed the usefulness of the resilience framework in reducing this complexity through a step-wise approach tailored to farming systems. The participatory approaches presented in this thesis contributed mainly to describing and explaining sustainability and resilience of current farming systems. These methods provide, therefore, a good basis for exploring future farming systems. The quantitative approach (presented in Chapter 4) confirmed the impact of weather extremes on economic and environmental farm performance, but was limited in explaining resilience, and raised awareness about the influence researchers have on the results through the selection of response variables. Based on the work and reflections presented in this thesis I see scope for better understanding and assessing farming system sustainability and resilience through system thinking theory and the use of participatory integrated assessment

Samenvatting

Een toenemende diversiteit aan stressfactoren en schokken treft de landbouw in Europa. Als gevolg daarvan komen de duurzaamheid en veerkracht van Europa's diverse agrarische systemen in gevaar. De mogelijke aanwezigheid van economische, sociale of milieukundige kantelpunten zijn in het bijzonder reden tot zorg, omdat voorbij deze kantelpunten permanente en ongewenste systeemveranderingen plaats kunnen vinden.

De duurzaamheid van een systeem is in dit proefschrift gedefinieerd als een adequate prestatie van alle systeemfuncties in de milieukundige, economische en sociale domeinen. Duurzaamheid van agrarische systemen is uitgebreid bestudeerd, maar de bestaande raamwerken en methoden zijn niet ontworpen om veerkracht te bestuderen, wat veel meer gaat over de verschillende capaciteiten van een systeem om om te gaan met verstoringen. Bovendien zijn sociale aspecten in duurzaamheidsbeoordelingen in de landbouw vaak in geringe mate meegenomen ten opzichte van economische en milieukundige aspecten. Om duurzaamheid en veerkracht te beoordelen zijn participatieve en integrale beoordelingen van bijzonder belang omdat ze ontworpen zijn om ook met adaptatie-opties en actie-georiënteerde benaderingen te komen in samenspraak met relevante belanghebbenden.

Veerkracht kan worden gedefinieerd als de capaciteit van een systeem om verandering te weerstaan zonder dat de terugkoppelingsmechanismen en functionaliteit van het systeem veranderen, d.w.z. robuustheid, terwijl de mogelijkheid van alternatieve stabiele toestanden van het systeem ook wordt erkend. Het erkennen van mogelijke alternatieve stabiele toestanden heeft wetenschappers er toe gebracht om te stellen dat ecologische veerkracht, naast robuustheid, ook aanpassingsvermogen en het vermogen om structureel en functioneel te veranderen zou moeten omvatten. Veerkracht van agrarische systemen is bestudeerd op een conceptueel niveau en veerkracht indicatoren zijn geopperd. De toepassing op concrete agrarische systemen is echter in beperkte mate doorgevoerd. In het bijzonder kwantitatieve benaderingen ontbreken in de literatuur. Gebrek aan goede data zou hier een van de redenen voor kunnen zijn. Veel studies naar veerkracht in agrarische systemen zijn daardoor kwalitatief van aard, bijvoorbeeld door inschattingen te maken op basis van systeemkarakteristieken die zogenaamd veerkracht aan het systeem zouden moeten geven. De voorbeelden die er zijn waarin het veerkrachtconcept voor agrarische systemen is geoperationaliseerd volgen bovendien geen integraal raamwerk voor veerkracht dat toegepast is in veel verschillende agrarische systemen.

Het onderzoek dat gepresenteerd wordt in dit proefschrift was onderdeel van het EU Horizon 2020 project SURE-Farm om de duurzaamheid en veerkracht van Europese boerderijsystemen te bepalen. Binnen SURE-Farm werd een veerkrachtraamwerk ontwikkeld dat oog heeft voor het combineren van verschillende benaderingen, een lokale focus en het stimuleren van participatie van relevante (lokale) actoren. Het SURE-Farm veerkrachtraamwerk bestaat uit vijf stappen om de duurzaamheid en veerkracht van Europese boerderijsystemen te bepalen. Van Stap 1 tot 5 wordt specifieke veerkracht bepaald, d.w.z. veerkracht ten opzichte van een specifieke verstoring. Stap 1 tot 3 verhouden zich daarom tot de vragen "de veerkracht waarvan?", "ten opzichte van wat?" en "met welk doel?". Stap 4 gaat over de specifieke capaciteiten van veerkracht van het boerderijsysteem. Gebaseerd op de voorgenoemde stappen, systeemkarakteristieken kunnen worden geïdentificeerd die generieke veerkracht geven aan het systeem, ongeacht het type van verstoring (Stap 5).

Het doel van dit proefschrift is om het hierboven genoemde veerkracht raamwerk te operationaliseren met nieuwe en integrale methoden om de duurzaamheid en veerkracht van huidige en toekomstige Europese boerderijsystemen te kunnen bepalen. De volgende onderzoeksvragen staan centraal in dit proefschrift: 1) Is er een balans tussen sociale, economische en milieukundige functies van Europese boerderijsystemen? 2) Naderen Europese boerderijsystemen kritische drempelwaarden? 3) Welke capaciteiten van veerkracht hebben Europese boerderijsystemen, en welke zouden ze moeten hebben? 4) Welke strategieën verbeteren de duurzaamheid en veerkracht van Europese boerderijsystemen? De methoden die toegepast zijn in dit proefschrift volgen de vijf stappen van het veerkracht raamwerk zo goed als mogelijk. Vanuit een methodologisch perspectief operationaliseren de methoden het raamwerk door lokaal aangepaste indicatoren en verschillende bronnen en soorten van data te gebruiken. Hoofdstuk 2 presenteert het Framework of "Participatory Impact Assessment for Sustainable and Resilient FARMing systems" (FoPIA-SURE-Farm I), FoPIA-SURE-Farm I onderzoekt huidige boerderijsystemen m.b.t. het functioneren, de dynamiek van belangrijke indicatoren, en de verschillende capaciteiten van veerkracht, d.w.z. robuustheid, adaptatievermogen en het vermogen tot transformatie. Drie gevalstudies met gespecialiseerde boerderijsvstemen worden gebruikt als voorbeeld voor de gebruikte methodologie: de productie van zetmeelaardappelen in de Veenkoloniën, Nederland; zuivelproductie in Vlaanderen, België: en de productie van hazelnoten in Lazio, Italië, Hoofdstuk 3 presenteert de synthese van de toepassing van FoPIA-SURE-Farm I in 11 Europese boerderijsvstemen. De resultaten van Hoofdstukken 2 & 3 overlappen. In de meeste boerderijsvstemen worden functies relaterend aan voedselproductie. economische levensvatbaarheid en het onderhouden van natuurlijke hulpbronnen gezien als het meest belangrijk. De door de deelnemers waargenomen prestaties van systeemfuncties suggereert een middelmatige duurzaamheid van de bestudeerde boerderijsvstemen. In het algemeen werd de veerkracht ingeschat als laag tot middelmatig, waar robuustheid en aanpassingsvermogen sterker ingeschat werden dan het vermogen om te transformeren. Dit geeft aan dat het vinden van wegen naar meer duurzaamheid een uitdagend proces is, want daar is immers aanpassingsvermogen en vermogen om te transformeren voor nodig. De karakteristieken van boerderijsystemen die zogezegd generieke veerkracht geven, de zogenaamde attributen van veerkracht, werden inderdaad waargenomen als positief voor de veerkracht. Winstgevendheid, een productie gekoppeld aan lokale en natuurlijke hulpbronnen, heterogeniteit van boerderijtypes, sociale zelforganisatie, en infrastructuur voor innovatie werden ingeschat als belangrijke attributen van veerkracht. Voor het vermogen tot transformatie werden met name winstgevendheid en toegang tot infrastructuur voor innovatie essentieel geacht. Strategieën die toegepast werden in het verleden waren er vooral op gericht om de winstgevendheid te vergroten en minder gericht op de andere belangrijke attributen van veerkracht. Om de duurzaamheid en veerkracht van Europese landbouwsystemen te vergroten zullen reacties op korte termijn processen beter rekening moeten houden met lange termijn processen. Technologische innovatie is nodig, maar het moet gepaard gaan met structurele, sociale, agro-ecologische en institutionele veranderingen. De relatieve belangrijkheid van sommige attributen van veerkracht verschilde tussen boerderijsystemen. Dit geeft aan dat de lokale context in het algemeen, en perspectieven van belanghebbenden in het bijzonder, belangrijk zijn voor het evalueren van generieke veerkracht en beleidsopties op basis van attributen van veerkracht. Alles bij elkaar opgeteld, lijkt FoPIA-SURE-Farm I een goed startpunt voor het creëren van bewustzijn, verder onderzoek en uiteindelijk het ontwikkelen van een gedeelde visie en actieplan voor het verbeteren van duurzaamheid en veerkracht van boerderijsystemen.

Hoofdstuk 4 heeft als doel de kwalitatieve resultaten over huidige duurzaamheid en veerkracht uit Hoofdstukken 2 & 3 aan te vullen met een kwantitatieve benadering. Een statistische methode voor meerjarige datareeksen was toegepast om boerderijprestaties te bestuderen mb.t. voedselproductie, winstgevendheid en stikstofoverschot onder verschillende omstandigheden wat betreft weer, markt en boerderijstructuur. Een gevalstudie voor drie aardappel producerende regio's werd gebruikt met data van 2006 t/m 2019. De prestaties van de statistische modellen waren hoogstens middelmatig. Model resultaten waren gemakkelijk te beïnvloeden door de selectie van responsvariabelen. Het niveau en de graduele verandering van voedselproductie, economische prestaties en milieuprestaties werden voornamelijk beïnvloed door input-intensiteit. Hoe die beïnvloeding door intensiteit werkte, d.w.z. positief of negatief, verschilde per gevalstudie. Jaarlijkse variatie werd hoofdzakelijk bepaald door gemiddelde jaarlijkse weersomstandigheden en weersextremen. Er werd geen bewijs gevonden voor impactreductie van weersomstandigheden en weersextremen door boerderijstructuur. Alles bij elkaar genomen kan de conclusie getrokken worden dat de resultaten in dit hoofdstuk slechts bestaande kennis bevestigen op het niveau van de gevalstudies. In de context van veerkracht van boerderijen in combinatie met een relatieve kleine dataset lijkt de methode zich te beperken tot een vrij homogene boerderijpopulatie in een stabiele economische omgeving. Onderzoekers die de gebruikte methoden willen gebruiken in (akkerbouw) boerderijsystemen zullen goed moeten beseffen dat ze veel invloed kunnen hebben op de resultaten door de selectie van responsvariabelen.

Hoofdstuk 5 heeft als doel om paden naar meer duurzame en veerkrachtige boerderijsystemen te vinden, terwijl kritische drempelwaarden worden geïdentificeerd en ontweken. Voor dit doel werd een

participatieve, integrale en indicator-gebaseerde methode gepresenteerd. Deze methode leidt onderzoekers en belanghebbenden in het boerderijsysteem in zes stappen naar een multidimensionaal begrip van duurzaamheid en veerkracht van toekomstige boerderijsystemen (FoPIA-SURE-Farm II). De methode wordt gepresenteerd aan de hand van een gevalstudie met extensieve schapenhouderij in Huesca, Spanje. Deelnemers in de participatieve workshop gaven aan dat hun boerderijsysteem heel dicht bij een neergang of zelfs ineenstorting staat. Het naderen en overschrijden van kritische drempelwaarden in het sociale, economische en milieu domein zorgt op dit moment voor een vicieuze cirkel met onder meer lage winstgevendheid, lage aantrekkingskracht van het boerderijsvsteem en de verwaarlozing van weidegronden. Deelnemers stelden twee alternatieve, meer duurzame en veerkrachtige systemen voor: een semi-intensief systeem dat allereerst gericht is op het verhogen van de productie en een hightech extensief systeem dat allereerst gericht is op het aanbieden van publieke diensten. Beide alternatieven leggen een sterke nadruk op de rol van technologie, maar verschillen in hun benadering tot weidegang. Dit verschil wordt gereflecteerd in de verschillende strategieën die voorzien worden om de alternatieve systemen te bewerkstelligen. Alhoewel het hightech extensieve systeem het meest verenigbaar lijkt met een toekomstscenario waarin de productie van duurzaam voedsel erg belangrijk is, lijkt het semi-intensieve systeem minder risicovol omdat het, gemiddeld genomen, het meest verenigbaar is met meerdere andere toekomstscenario's. Alles bij elkaar genomen, kan de gebruikte methode gezien worden als relatief snel, interactief en trans-disciplinair, die veel informatie aanlevert over kritische drempelwaarden, huidige systeemdynamieken en toekomstige mogelijkheden. Zo doende helpt de methode belanghebbenden om na te denken en te praten over de toekomst van hun systeem, en helpt zo in de voorbereiding naar meer duurzaamheid en veerkracht.

In hoofdstuk 6 wordt FoPIA-SURE-Farm II toegepast om de aanwezigheid van kritische drempelwaarden te bepalen in 11 Europese boerderijsvstemen. Alle bestudeerde systemen bevinden zich, aldus deelnemers, dichtbij, op of over ten minste één geïdentificeerde kritische drempelwaarde, d.w.z. dat deze systemen (erg waarschijnlijk) kritische drempelwaarden gaan overschrijden binnen de komende 10 jaren. Deelnemers waren in het bijzonder bezorgd over economische levensvatbaarheid en niveaus van voedselproductie. Kritische drempelwaarden werden bovendien waargenomen als interacterend tussen systeemniveaus (veld, boerderii, boerderiisysteem) en domeinen (milieu, economisch, sociaal), waar lage economische levensvatbaarheid leidt tot een lagere aantrekkingskracht van het boerderijsysteem, en, voor sommige boerderijsvstemen, een hoge moeilijkheidsgraad om natuurlijke hulpbronnen en biodiversiteit te onderhouden. In het algemeen werd een neergang in prestaties van alle systeemvariabelen verwacht door workshop deelnemers in het geval kritische drempelwaarden zouden worden overschreden. Bijvoorbeeld een neergang in de aantrekkingskracht van een gebied en minder onderhoud aan natuurlijke hulpbronnen en biodiversiteit. Hoofdstuk 6 laat zien dat bezorgdheid over het overschrijden van kritische drempelwaarden terecht is en dat de drempelwaarden van systeemvariabelen onderzocht moeten worden op veld-, boerderij- en boerderijsvsteem niveau en in het sociale, economische en milieu domein. Economische variabelen op boerderijniveau (bijvoorbeeld inkomen), lijken belangrijk om te detecteren of een systeem richting kritische waarden van sociale variabelen gaat op het niveau van het boerderijsysteem (bijvoorbeeld aantrekkingskracht van het gebied). Tegelijkertijd zijn er in meerdere gevalstudies indicaties dat het naderen van drempelwaarden van sociale variabelen (bijvoorbeeld beschikbaarheid van arbeid) een aanwijzing zijn voor het naderen van economische drempelwaarden (bijvoorbeeld bedrijfsinkomen). Op basis van de resultaten volgen enkele reflecties op het belang van het in het oog houden van hulpbronnen van het systeem, zoals kennis niveaus, bij het monitoren en evalueren van de duurzaamheid en veerkracht van Europese boerderijsystemen.

Hoofdstuk 7 geeft een synthese van het onderzoek dat weergegeven wordt in dit proefschrift. Gebaseerd op deze synthese worden aanbevelingen gedaan voor het verbeteren van duurzaamheid en veerkracht. Vervolgens worden het theoretische raamwerk en de methodes geëvalueerd die gebruikt zijn in dit proefschrift. Daarna volgt een discussie over de relevantie van het onderzoek. Op basis van de voorgaande discussies worden reflecties gegeven over het verbeteren van het monitoren en evalueren van duurzaamheid en veerkracht van boerderijsystemen in Europa. Ik concludeer dat de duurzaamheid en veerkracht van boerderijsystemen een uitdagend onderwerp blijft door de complexiteit in termen van detail (verschillende domeinen, veel concepten en variabelen) en dynamieken (non-lineariteit, drempelwaarden,

interacties). Het onderzoek dat gepresenteerd wordt in dit proefschrift bevestigt het nut van het gebruikte veerkracht raamwerk: de complexiteit rondom veerkracht is gereduceerd middels een stapsgewijze benadering aangepast aan boerderijsystemen. De participatieve benaderingen in dit proefschrift droegen vooral bij aan het beschrijven en uitleggen van duurzaamheid en veerkracht van huidige boerderijsystemen. De resultaten van deze methoden vormen daardoor een goede basis voor het exploreren van toekomstige boerderijsystemen. De kwantitatieve benadering (gepresenteerd in Hoofdstuk 4) bevestigde de impact van weersextremen op economische en milieukundige boerderijprestaties, maar was beperkt wat betreft het verstrekken van inzicht in veerkracht. De kwantitatieve benadering liet ook zien hoe onderzoekers invloed kunnen hebben op de resultaten door de selectie van responsvariabelen. Gebaseerd op resultaten en reflecties in dit proefschrift zie ik ruimte voor verbetering wat betreft het begrijpen en bepalen van duurzaamheid en veerkracht door meer gebruik te maken van systeemdenken en het gebruik van participatieve integrale assessments.

Acknowledgements

Herewith I would like to acknowledge the people that have been important while being in the PhD-process.

First of all, I'd like to thank my supervisors for their support in good and bad times. Pytrik, Miranda and Martin, you have been very accessible colleagues and nice people to spend time with. Pytrik, you are remarkably bright and quick in thinking; we've only worked out a small part of our ideas on defining and assessing resilience. I very much enjoyed the wide field of opportunities that you opened up for doing research within our work package of the SURE-Farm project. Miranda, thank you for your inclusive presence and thoughtful reflections on process and content. Having you as the SURE-Farm consortium leader created lively discussions and collaborations where all individual voices could be heard. Martin, thank you for monitoring and guiding the overall process; it helped a lot in finishing this thesis in time. I also appreciate the time we took for reflecting on conducting scientific research.

I would like to thank all my colleagues in the Plant Production Systems Group as well. There are some colleagues that I would like to thank in specific. Bert and Marcel, I'm very grateful for your invitation to your office in 2018. It was very nice to share the office, later also with Mink, and have a chat and a laugh from time to time. Bert and Eva H., thanks for being my paranymphs today, the honour is mine. Eva H. and Wytze, I enjoyed our regular talks while being in the process of finishing a PhD. Ken and Wytze, thanks for the occasions where we could discuss the importance of indigenous knowledge while felling trees. Linda and Karen, thanks a lot for your continuous support, in listening, guiding and taking swift action when necessary. Bert and Inge, thank you for assisting in the workshops in the Veenkoloniën. Also a big thank you to my MSc-Students Letizia, Elena, Jorrit, William, Yvonne, Lilli, Abel, Roos and Clemence. It was a pleasure working with you!

I would like to thank all partners and participants in the SURE-Farm project too. Some persons I would like to thank in specific. First of all, I would like to thank the farming system stakeholders in the Veenkoloniën and other case study areas for their participation in workshops. Francesco and Hugo, thank you for the smooth collaborations within the context of our work package. Simone and Isabeau, thank you for our collaboration on Chapter 2. To all case study partners a big 'thank you' for our collaboration on Chapter 3 and 6 and multiple project reports. Carolina and Barbara, thank you for our collaboration on Chapter 5. I would also like to thank Ruud for supporting the access to data for Chapter 4.

Friends and family have been important throughout the process of writing this dissertation. As I'm not sure how to express that in words, right here, I would like to suggest that we just keep creating good times together!

About the author

Wim Paas was born and raised in Genemuiden, the Netherlands, where he enjoyed the water-rich environment and a hands-on work culture. He completed a BSc study International Land and Water Management at Wageningen University in 2011. During an internship in Ethiopia on small-scale irrigation schemes he became interested in agronomy. He completed a MSc study Plant Sciences with a specialization in Natural Resource Management at Wageningen University in 2013. After graduation, he worked as a junior researcher on tropical farming systems with the Farming Systems Ecology group (2013-2014) and the Plant Production Systems group of Wageningen University (2015-2016). In 2017 he started a PhD on the sustainability and resilience of European farming systems with the Plant Production Systems group and Business Economics group of Wageningen University. The PhD and additional activities were conducted within the context of SURE-Farm, an international consortium funded by the Horizon 2020 program of the European Commission. In his research, Wim is keen on applying quantitative and qualitative research methods in complementary ways to capture the various dimensions of social-ecological systems. Besides doing research and reflecting on existence, he likes to move around by foot and bicycle, take care of trees and gardens, tinker with tools and machines and build constructions that improve everyday life in and around the homestead. In the near future, he envisions to settle with his family on a farm in rural France where he will put his body and mind to work.

Publications

Peer reviewed articles

- Herrera, H., Schütz, L., Paas, W., Reidsma, P., Kopainsky, B., 2022. Understanding resilience of farming systems: Insights from system dynamics modelling for an arable farming system in the Netherlands. Ecol. Modell. 464, 109848. https://doi.org/10.1016/J.ECOLMODEL.2021.109848
- Paas, W., Accatino, F., Bijttebier, J., Black, J.E., Gavrilescu, C., Krupin, V., Manevska-Tasevska, G., Ollendorf, F., Peneva, M., San Martin, C., Zinnanti, C., Appel, F., Courtney, P., Severini, S., Soriano, B., Vigani, M., Zawalińska, K., van Ittersum, M.K., Meuwissen, M.P.M., Reidsma, P., 2021. Participatory assessment of critical thresholds for resilient and sustainable European farming systems. J. Rural Stud. 88, 214–226. https://doi.org/10.1016/J.JRURSTUD.2021.10.016
- Paas, W., San Martin, C., Soriano, B., van Ittersum, M.K., Meuwissen, M.P.M., Reidsma, P., 2021. Assessing future sustainability and resilience of farming systems with a participatory method: A case study on extensive sheep farming in Huesca, Spain. Ecol. Indic. 132, 108236. https://doi.org/https://doi.org/10.1016/i.ecolind.2021.108236
- Meuwissen, M.P.M., Feindt, P.H., Slijper, T., Spiegel, A., Finger, R., de Mey, Y., Paas, W., Termeer, K.J.A.M., Poortvliet, P.M., Peneva, M., Urquhart, J., Vigani, M., Black, J.E., Nicholas-Davies, P., Maye, D., Appel, F., Heinrich, F., Balmann, A., Bijttebier, J., Coopmans, I., Wauters, E., Mathijs, E., Hansson, H., Lagerkvist, C.J., Rommel, J., Manevska-Tasevska, G., Accatino, F., Pineau, C., Soriano, B., Bardaji, I., Severini, S., Senni, S., Zinnanti, C., Gavrilescu, C., Bruma, I.S., Dobay, K.M., Matei, D., Tanasa, L., Voicilas, D.M., Zawalińska, K., Gradziuk, P., Krupin, V., Martikainen, A., Herrera, H., Reidsma, P., 2021. Impact of Covid-19 on farming systems in Europe through the lens of resilience thinking. Agric. Syst. 191, 103152. https://doi.org/10.1016/j.agsy.2021.103152
- Paas, W., Coopmans, I., Severini, S., van Ittersum, M., Meuwissen, M., Reidsma, P., 2021. Participatory assessment of sustainability and resilience of three specialized farming systems. Ecol. Soc. 26, 2. https://doi.org/https://doi.org/10.5751/ES-12200-260202
- Reidsma, P., Meuwissen, M., Accatino, F., Appel, F., Bardaji, I., Coopmans, I., Gavrilescu, C., Heinrich, F., Krupin, V., Manevska-Tasevska, G., Peneva, M., Rommel, J., Severini, S., Soriano, B., Urquhart, J., Zawalinska, K., Paas, W., 2020. How do stakeholders perceive the sustainability and resilience of EU farming systems? Eurochoices 19, 18–27. https://doi.org/10.1111/1746-692X.12280
- Nera, E., Paas, W., Reidsma, P., Paolini, G., Antonioli, F., Severini, S., 2020. Assessing the Resilience and Sustainability of a Hazelnut Farming System in Central Italy with a Participatory Approach. Sustainability 12, 343. https://doi.org/10.3390/su12010343
- Meuwissen, M.P.M., Feindt, P.H., Spiegel, A., Termeer, C.J.A.M., Mathijs, E., de Mey, Y., Finger, R., Balmann, A., Wauters, E., Urquhart, J., Vigani, M., Zawalinska, K., Herrera, H., Nicholas-Davies, P., Hansson, H., Paas, W., Slijper, T., Coopmans, I., Vroege, W., Ciechomska, A., Accatino, F., Kopainsky, B., Poortvliet, P.M., Candel, J., Maye, D., Severini, S., Senni, S., Soriano, B., Lagerkvist, C.J., Peneva, M., Gavrilescu, C., Reidsma, P., 2019. A framework to assess the resilience of farming systems. Aqric. Syst. 176, 102656.
- Alvarez, S., Timler, C.J., Michalscheck, M., Paas, W., Descheemaeker, K., Tittonell, P., Andersson, J.A., Groot, J.C.J., 2018. Capturing farm diversity with hypothesis-based typologies: An innovative methodological framework for farming system typology development. PLoS One 13, e0194757. https://doi.org/10.1371/journal.pone.0194757
- Paas, W., Groot, J.C.J., 2017. Creating adaptive farm typologies using Naive Bayesian classification. Inf. Process. Agric. 4, 220–227. https://doi.org/10.1016/j.inpa.2017.05.005
- Paas, W., Kanellopoulos, A., van de Ven, G., Reidsma, P., 2016. Integrated impact assessment of climate and socio-economic change on dairy farms in a watershed in the Netherlands. NJAS - Wageningen J. Life Sci. 78, 35–45. <u>https://doi.org/10.1016/j.njas.2015.12.004</u>
- Reidsma, P., Bakker, M.M., Kanellopoulos, A., Alam, S.J., Paas, W., Kros, J., de Vries, W., 2015. Sustainable agricultural development in a rural area in the Netherlands? Assessing impacts of climate and socioeconomic change at farm and landscape level. Agric. Syst. 141, 160–173. https://doi.org/10.1016/j.agsy.2015.10.009

Book chapters

- Meuwissen, M.P.M., Feindt, P.H., Spiegel, A., Paas, W., Soriano, B., Mathijs, E., Balmann, A., Urquhart, J., Kopainsky, B., Garrido, A., Reidsma, P., 2022. SURE-Farm Approach to Assess the Resilience of European Farming Systems, in: Garrido, A., Spiegel, A., Soriano, B., Mathijs, E., Urquhart, J., Meuwissen, M.M.P., Feindt, P.H. (Eds.), Resilient and Sustainable Farming Systems in Europe: Exploring Diversity and Pathways. Cambridge University Press, Cambridge, pp. 1–17. https://doi.org/DOI: 10.1017/9781009093569.002
- Black, J.E., Courtney, P., Maye, D., Urquhart, J., Vigani, M., Paas, W., Senni, S., Bertolozzi-Caredio, D., Reidsma, P., 2022. Constrained Sustainability and Resilience of Agricultural Practices from Multiple Lock-In Factors and Possible Pathways to Tackle Them: An Assessment of Three European Farming Systems, in: Garrido, A., Spiegel, A., Soriano, B., Mathijs, E., Urquhart, J., Meuwissen, M.M.P., Feindt, P.H. (Eds.), Resilient and Sustainable Farming Systems in Europe: Exploring Diversity and Pathways. Cambridge University Press, Cambridge, pp. 88–111. https://doi.org/DOI: 10.1017/9781009093569.006
- Spiegel, A., Reidsma, P., Buitenhuis, Y., Slijper, T., Paas, W., de Mey, Y., Feindt, P.H., Candel, J., Poortvliet, P.M., Meuwissen, M.P.M., 2022. Realising Transformation in Response to Future Challenges: The Case of an Intensive Arable Farming System in the Veenkoloniën, the Netherlands, in: Garrido, A., Spiegel, A., Soriano, B., Mathijs, E., Urquhart, J., Meuwissen, M.M.P., Feindt, P.H. (Eds.), Resilient and Sustainable Farming Systems in Europe: Exploring Diversity and Pathways. Cambridge University Press, Cambridge, pp. 201–214. https://doi.org/DOI: 10.1017/9781009093569.013
- Accatino, F., Paas, W., Herrera, H., Pinsard, C., Severini, S., Appel, F., Kopainsky, B., Bańkowska, K., Bijttebier, J., Gavrilescu, C., Khafagy, A., Krupin, V., Manevska-Tasevska, G., Ollendorf, F., Peneva, M., Martín, C.S., Zinnanti, C., Reidsma, P., 2022. Integrated Assessment of the Sustainability and Resilience of Farming Systems: Lessons from the Past and Ways Forward for the Future, in: Garrido, A., Spiegel, A., Soriano, B., Mathijs, E., Urquhart, J., Meuwissen, M.M.P., Feindt, P.H. (Eds.), Resilient and Sustainable Farming Systems in Europe: Exploring Diversity and Pathways. Cambridge University Press, Cambridge, pp. 279–301. https://doi.org/DOI: 10.1017/9781009093569.018
- Soriano, B., Bardají, I., Buitenhuis, Y., Bertolozzi-Caredio, D., Candel, J., Feindt, P.H., Meuwissen, M.P.M., Paas, W., Reidsma, P., Martín, C.S., Slijper, T., Spiegel, A., Garrido, A., 2022. Lessons Learned on Resilience from a Multi-scale Co-creation Methodology: From Regional to European Scale, in: Garrido, A., Spiegel, A., Soriano, B., Mathijs, E., Urquhart, J., Meuwissen, M.M.P., Feindt, P.H. (Eds.), Resilient and Sustainable Farming Systems in Europe: Exploring Diversity and Pathways. Cambridge University Press, Cambridge, pp. 321–341. https://doi.org/DOI: 10.1017/9781009093569.020
- Feindt, P.H., Meuwissen, M.P.M., Balmann, A., Finger, R., Mathijs, E., Paas, W., Soriano, B., Spiegel, A., Urquhart, J., Reidsma, P., 2022. Understanding and Addressing the Resilience Crisis of Europe's Farming Systems: A Synthesis of the Findings from the SURE-Farm Project, in: Garrido, A., Spiegel, A., Soriano, B., Mathijs, E., Urquhart, J., Meuwissen, M.M.P., Feindt, P.H. (Eds.), Resilient and Sustainable Farming Systems in Europe: Exploring Diversity and Pathways. Cambridge University Press, Cambridge, pp. 342–374. https://doi.org/DOI: 10.1017/9781009093569.021

Scientific reports

- Reidsma, P., Paas, W., Accatino, F., Appel, F., Black, J.E., Bijttebier, J., Gavrilescu, C., Kopainsky, B., Krupin, V., Manevska-Tasevska, G., Meuwissen, M.P.M., Ollendorf, F., Peneva, M., Senni, S., Severini, S., Soriano, B., Urquhart, J., Vigani, M., Zawalinska, K., Zinnanti, C., Herrera, H., 2021. Resilience of farming systems in the EU under current conditions and future scenarios: D5.7 Policy Brief.
- Reidsma, P., Paas, W., Accatino, F., Appel, F., Black, J., Bijttebier, J., Gavrilescu, C., Kopainsky, B., Krupin, V., Manevska-Tasevska, G., Meuwissen, M., Ollendorf, F., Peneva, M., Senni, S., Severini, S., Soriano, B., Urquhart, J., Vigani, M., Zawalińska, K., Zinnanti, C., Herrera, H., 2020. D5.6 Impacts of improved strategies and policy options on the resilience of farming systems across the EU.
- Accatino, F., Paas, W., Herrera, H., Appel, F., Pinsard, C., Shi, Y., Schutz, L., Kopainsky, B., Bańkowska, K., Bijttebier, J., Black, J., Gavrilescu, C., Krupin, V., Manevska-Tasevska, G., Ollendorf, F., Peneva, M., Rommel, J., San Martín, C., Severini, S., Soriano, B., Valchovska, S., Vigani, M., Wauters, E., Zawalińska, K., Zinnanti, C., Meuwissen, M., Reidsma, P., 2020. D5.5 Impacts of future scenarios on the resilience of farming systems across the EU assessed with quantitative and qualitative methods.
- Reidsma, P., Spiegel, A., Paas, W., Accatino, F., Antonioli, F., Appel, F., Bardají, I., Berry, R., Bertolozzi, D., Bijttebier, J., Black, J., Buitenhuis, Y., Coopmans, I., Courtney, P., Feindt, P., Gavrilescu, C., Hansson, H., Jendrzejewski, B., Khafagy, A., Krupin, V., Lagerkvist, C.-J., Larson, S., Lievens, E., Mathijs, E., Manevska-Tasevska, G., Maye, D., Ollendorf, F., Peneva, M., Pettit, A., Pinsard, C., Rommel, J., Senni, S., Severini, S., Slijper, T., Soriano, B., Urquhart, J., Valchovska, S., Vigani, M., Wauters, E., Zawalińska, K., Meuwissen, M., 2019. D5.3 Resilience assessment of current farming systems across the European Union. Sustainable and resilient EU farming systems (SURE-Farm) project report, EU Horizon 2020 Grant Agreement No. 727520.
- Paas, W., Accatino, F., Antonioli, F., Appel, F., Bardaji, I., Coopmans, I., Courtney, P., Gavrilescu, C., Heinrich, F., Krupin, V., Manevska-Tasevska, G., Neumeister, D., Peneva, M., Rommel, J., Severini, S., Soriano, B., Tudor, M., Urquhart, J., Wauters, E., Zawalinska, K., Meuwissen, M., Reidsma, P., 2019. D5.2 Participatory impact assessment of sustainability and resilience of EU farming systems. Sustainable and resilient EU farming systems (SureFarm) project report, EU Horizon 2020 Grant Agreement No. 727520. https://doi.org/10.13140/RG.2.2.25104.25601
- Meuwissen, M., Paas, W., Slijper, T., Coopmans, I., Ciechomska, A., Lievens, E., Deckers, J., Vroege, W., Mathijs, E., Kopainsky, B., Herrera, H., Nitzko, S., Finger, R., De Mey, Y., Poortvliet, M., Nicholas-Davies, P., Midmore, P., Vigani, M., Maye, D., Urquhart, J., Balmann, A., Appel, F., Termeer, K., Feindt, P., Candel, J., Tichit, M., Accatino, F., Severini, S., Wauters, E., Bardaji, I., Soriano, B., Zawalinska, K., Lagerkvist, C.J., Manevska-Tasevska, G., Hansson, H., Peneva, M., Gavrilescu, C., Reidsma, P., 2018. D1.1 Report on resilience framework for EU agriculture. Sustainable and resilient EU farming systems (SureFarm) project report.
- Herrera, H., Kopainsky, B., Appel, F., Balmann, A., Accatino, F., Tichit, M., Antonioli, F., Severini, S., Paas, W., Reidsma, P., 2018. D5.1 Impact assessment tool to assess the resilience of farming systems and their delivery of private and public goods. Sustainable and resilient EU farming systems (SureFarm) project report, EU Horizon 2020 Grant Agreement No. 727520.

PE&RC Training and Education Statement

With the training and education activities listed below the PhD candidate has complied with the requirements set by the C.T. de Wit Graduate School for Production Ecology and Resource Conservation (PE&RC) which comprises of a minimum total of 32 ECTS (= 22 weeks of activities)

Review of literature (4.5 ECTS)

- Methods (and indicators) to study the sustainability and resilience of farming systems

Writing of project proposal (4.5 ECTS)

- Integrated assessment of the sustainability and resilience of European farming systems

Post-graduate courses (7.6 ECTS)

- Companion modelling: WUR/CIRAD (2014)
- Participatory trials design; ICRAF (2014)
- Bayesian statistics; PE&RC; SENSE (2017)
- Multivariate analysis; PE&RC; SENSE (2019)
- GIS in practice; PE&RC; WIMEK (2019)

Invited review of (unpublished) journal manuscript (8 ECTS)

 11 verified reviews for 8 individual manuscripts in 7 different journals (<u>https://www.webofscience.com/wos/author/record/1863521</u>)

Competence strengthening / skills courses (2.2 ECTS)

- Effective behaviour in the professional surroundings; WUR (2014)
- Project and time management; WUR (2019)
- Competence assessment; WGS (2020)
- Career assessment; WGS (2022)

Scientific Integrity/Ethics in science activities (0.3 ECTS)

- Ethics in plant and environmental sciences; WGS (2019)

PE&RC Annual meetings, seminars and the PE&RC weekend (1.5 ECTS)

- PE&RC Midterm weekend (2019)
- PE&RC Day (2019)
- PE&RC Last year weekend; (2021)

Discussion groups / local seminars or scientific meetings (11.1 ECTS)

- Sustainable intensification of agricultural systems (SIAS) (2014-2015)
- Tuesdays with resilience (2017)
- SURE-Farm consortium meetings (2017-2021)
- 7th International Symposium for Farming System Design (2021)

International symposia, workshops and conferences (3.5 ECTS)

- 173rd symposium of the European Association of Agricultural Economists; Bucharest (2019)
- XVI European Society for Agronomy Congress; online (2020)



Societally relevant exposure (0.5 ECTS)

- News article in WUR's magazine Resource (2021)

Lecturing/supervision of practicals/tutorials (6.9 ECTS)

- Quantitative analysis of land use systems (2020-2021)
- Academic advisor for academic consultancy training (2022)

BSc/MSc thesis supervision (26 ECTS)

- The effects of change in extreme climate events on agriculture in the Netherlands
- Assessing the resilience and sustainability of a hazelnut farming system in Central Italy
- An assessment of the sustainability and resilience of two livestock farming systems in Europe using a participatory approach
- Participatory assessment of the functions, sustainability and resilience of large-scale arable farming in the East of England
- Assessing resilience and sustainability in German large-scale corporate arable farms
- Analyzing the resilience of an arable farming system in the Veenkoloniën, NL, using system dynamics modelling
- Exploring the impact of nematode dynamics on starch potato yield in the Veenkoloniën
- The impact and assessment of extreme drought events on arable crop production in the Netherlands
- Synergies and trade-offs between perceptions of sustainability and resilience in two small-scale European farming systems

This research described in this thesis was financially supported by the SURE-Farm Horizon 2020 project (Grant Agreement No 727520).

Financial support from Wageningen University for printing this thesis is gratefully acknowledged.

Cover design by Wim Paas in PowerPoint Printed by ProefschriftMaken.nl

